UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) of the SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): February 10, 2025

Super League Enterprise, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

001-38819

(Commission File Number)

47-1990734

(IRS Employer Identification Number)

2856 Colorado Avenue Santa Monica, California 90404

(Address of principal executive offices)

(213) 421-1920

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Chec	ck the appropriate box below if the Form 8-K filing is intend	ded to simultaneously satisfy the filing o	bligation of the registrant under any of the following provisions:								
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)										
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a -12)										
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d -2(b))										
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e -4(c))										
	Securiti	ies registered pursuant to Section 12(b	o) of the Act:								
	Title of each class Common Stock, par value \$0.001 per share	Trading Symbol(s) SLE	Name of each exchange on which registered Nasdaq Capital Market								
	cate by check mark whether the Registrant is an emerging g the Securities Exchange Act of 1934 (§240.12b-2 of this ch		of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b- Emerging growth company								
	n emerging growth company, indicate by check mark if th ncial accounting standards provided pursuant to Section 13(a	C	extended transition period for complying with any new or revised								

Item 1.01. Entry Into a Material Definitive Agreement

Entry into Loan Agreement

On February 10, 2025 (the "Effective Date"), the Company and its subsidiary, InPVP, LLC ("Subsidiary"), entered into a Business Loan and Security Agreement (the "Agile Loan Agreement"), with Agile Capital Funding, LLC as collateral agent ("Collateral Agent"), and Agile Lending, LLC ("Agile"), pursuant to which the Company issued to Agile a Confessed Judgment Secured Promissory Note for an aggregate value of \$2.5 million (the "Agile Note"). Pursuant to the Agile Loan Agreement: (i) the Agile Note matures 32 weeks from the Effective Date; (ii) carries an aggregate total interest payment of approximately \$1.05 million (the "Applicable Rate"), and (iii) immediately upon the occurrence and during the continuance of an Event of Default (as defined in the Agile Loan Agreement), interest shall accrue at a fixed per annum rate equal to the Applicable Rate plus five percent. The Company is required to repay all the obligations due under the Agile Loan Agreement and the Agile Note in 32 equal payments of \$110,937.50, with the first payment being made to Agile on February 17, 2025, and every seven days thereafter until the Maturity Date. The proceeds received from the Agile Note will be used to fund general working capital needs.

Pursuant to the Agile Loan Agreement, upon the occurrence of certain events, including (a) a change in the Company's business other than the business engaged in by the Company on the Effective Date, (b) cause or permit, voluntarily or involuntarily, any Key Person to cease being actively engagement in the management of the Company without prior notice to Agile, (c) a change in control of the Company (expressly excluding the pending transactions with Infinite Reality, Inc.) or otherwise approve the liquidation or dissolution of the Company or its Subsidiary (collectively, a "Change in Business, Management, or Ownership"), or (d) the Term Loan is accelerated upon the occurrence of an Event of Default, the Company shall be required to immediately pay to Agile an amount equal to the sum of: (i) all outstanding principal of the Agile Note plus accrued and unpaid interest thereon through the prepayment date, (ii) a fee equal to the aggregate and actual amount of interest (at the contract rate of interest) that would be paid through the Maturity Date (the "Prepayment Fee"), plus (iii) all other obligations that are due and payable, including, without limitation, interest at the Default Rate with respect to any past due amounts. The Company is allowed to make a full prepayment or partial prepayment (and receive a discount thereon if repaid in full within sixty days of note issuance) of any or all of the Obligations arising under the Agile Loan Agreement and the Agile Note, provided, the Company shall be obligated to pay the Prepayment Fee.

The Agile Loan Agreement imposes various restrictions on the activities of the Company, including, subject to certain exceptions set forth in the Agile Loan Agreement (including, without limitation, the pending transactions with Infinite Reality, Inc.), a prohibition on: (i) creating, incurring, assuming, or being liable for any indebtedness, or allow the Subsidiary to do so (expressly excluding up to \$3,000,000 of unsecured loans through one or more third parties); (ii) any Change in Business, Management, or Ownership; (iii) fundamental changes to the Company or its subsidiaries (including certain consolidations, mergers and sales/transfers of assets outside the ordinary course of business, and limitations on the ability of the Company and its Subsidiary to grant liens upon their property or assets); (iv) pay any dividends (other than dividends payable solely in capital stock) or make any distribution or payment in respect of or redeem, retire or purchase any capital stock; (v) enter into certain transactions with the Company's affiliates; (vi) make or permit any payment on any debt that is subordinate to the obligations under the Agile Loan Agreement and the Agile Note (expressly excluding up to \$3,000,000 of unsecured loans through one or more third parties); and (vi) other than in the ordinary course of business, entering into any material agreement, or terminating or materially amending a material agreement.

As security for the full and prompt payment and performance of any obligations arising under the Agile Loan Agreement and the Agile Note, the Company and its Subsidiary granted to Agile a continuing first priority security interest in all the assets of the Company and its Subsidiary; provided, however, the filing of a financing statement and/or the taking of any action required to perfect Agile's security interest in the collateral may only occur upon an event of default. The Agile Loan Agreement also provides for standard Events of Default, customary provisions, including representations, warranties and covenants, indemnification, waiver of jury trial, arbitration, and the exercise of remedies upon a breach or default.

In connection with entering into the Agile Loan Agreement, the Company was required to pay an administrative fee of \$125,000 to the Collateral Agent, which was paid at the closing out of proceeds of the issuance of the Agile Note.

The Agile Note and the Agile Loan Agreement are filed as Exhibits 4.1 and 10.1, respectively, to this Current Report on Form 10-K, and are incorporated herein by reference. The description of the terms of the Agile Note and the Agile Loan Agreement set forth above are qualified in their entirety by reference to such exhibit.

Entry into Equity Purchase Agreement

On February 14, 2025, we entered into an equity purchase agreement (the "Purchase Agreement") with Hudson Global Ventures, LLC, a Nevada limited liability company ("Hudson"). Pursuant to the Purchase Agreement, the Company has the right, but not the obligation, to sell to Hudson, and Hudson is obligated to purchase, up to \$2.9 million of newly issued shares (the "Total Commitment") of the Company's common stock, from time to time during the term of the Purchase Agreement, subject to certain limitations and conditions (the "Hudson Offering"). As consideration for Hudson's commitment to purchase shares of common stock under the Purchase Agreement, we issued to Hudson 300,000 shares of common stock, valued at \$165,000, following the execution of the Purchase Agreement (the "Commitment Shares").

The Purchase Agreement initially precludes us from issuing and selling more than 3,319,323 shares of our common stock, including the Commitment Shares, which number equals 19.99% of our common stock issued and outstanding as of February 14, 2025, unless we obtain stockholder approval to issue additional shares, or unless certain exceptions apply. In addition, a beneficial ownership limitation in the agreement initially limits us from directing Hudson to purchase shares of common stock if such purchases would result in Hudson beneficially owning more than 4.99% of the then-outstanding shares of our common stock.

From and after the initial satisfaction of the conditions to our right to commence sales to Hudson under the Purchase Agreement (such event, the "Commencement," and the date of initial satisfaction of all such conditions, the "Commencement Date"), we may direct Hudson to purchase shares of common stock at a purchase price per share equal to the lesser of (i) 92% of the closing price of the Company's Common Stock, as listed on Nasdaq, on the trading day immediately preceding the respective Put Date (the "Initial Purchase Price"), or (ii) 92% of the lowest closing price of the Company's Common Stock on the Nasdaq Capital Market on any trading day during the period beginning on the Put Date (as defined in the Purchase Agreement) and continuing through the date that is three trading days immediately following the Clearing Date (as defined in the Purchase Agreement) associated with the applicable Put Notice (such three trading day period is the "Valuation Period", and the price is the "Market Price"), on such date on which the Purchase Price is calculated in accordance with the terms of the Purchase Agreement. We will control the timing and amount of any such sales of common stock to Hudson. Actual sales of shares of common stock to Hudson will depend on a variety of factors to be determined by us from time to time, including, among other things, market conditions, the trading price of our common stock, and determinations by us as to the appropriate sources of funding for the Company and our operations.

Unless earlier terminated, the Purchase Agreement will automatically terminate upon the earliest of (i) the expiration of the Commitment Period (as defined in the Purchase Agreement), (ii) Hudson's purchase or receipt of the Total Commitment worth of common stock, or (iii) the occurrence of certain other events set forth in the Purchase Agreement. We have the right to terminate the Purchase Agreement at any time after Commencement, at no cost or penalty, upon prior written notice to Hudson.

We intend to use the net proceeds, if any, from the Hudson Offering for working capital and general corporate purposes, including sales and marketing activities, product development and capital expenditures. We may also use a portion of the net proceeds to acquire or invest in complementary businesses, products and technologies. The Purchase Agreement contains customary representations, warranties and agreements by us, as well as customary indemnification obligations of the Company. In connection with the Purchase Agreement, the Company entered into a Registration Rights Agreement, pursuant to which the Company agreed to register the Commitment Shares and the shares issuable pursuant to the Purchase Agreement.

The securities in the Hudson Offering are being offered pursuant to our effective S-3 shelf registration statement (File No. 333-283812), which was filed with the SEC on December 13, 2024 and declared effective on December 20, 2024. The Hudson Offering will be made only by means of a prospectus supplement and the accompanying base prospectus that form part of the S-3. A prospectus supplement relating to the Hudson Offering will be filed with the SEC pursuant to Rule 424(b) under the Securities Act on or around February 14, 2025.

The Purchase Agreement and Registration Rights Agreement are filed as Exhibit 10.2 and 10.3, respectively, to this Current Report on Form 8-K and are incorporated herein by reference. The description of the terms of the Purchase Agreement and Registration Rights Agreement set forth above are qualified in their entirety by reference to such exhibit. In addition, the legal opinion letter and consent of Disclosure Law Group, a Professional Corporation, counsel to the Company, regarding the validity of the shares of common stock to be issued from time to time in connection with the Hudson Offering is filed as Exhibit 5.1 and 23.1, respectively, to this Current Report on Form 8-K. The legal opinion letter is also filed with reference to, and is hereby incorporated by reference into, the S-3 shelf registration statement.

Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

Financial Statements and Exhibits.

Item 9.01

The information set forth above under Item 1.01 of this Current Report on Form 8-K under the title "Debt Financing" is hereby incorporated by reference in this Item 2.03.

The offering and sale of the Note has not been registered under the Securities Act of 1933, as amended (the 'Securities Act'), nor any state securities laws, and such Notes may not be offered or sold in the United States absent registration with the Securities and Exchange Commission or an applicable exemption from the registration requirements. The Company offered and sold the Note in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act. The Purchaser represented that they were acquiring the Note for its own account and not with a view to distribution or resale in violation of the Securities Act or any applicable state securities law.

Neither this Current Report on Form 8-K, nor any exhibit attached hereto, is an offer to sell or the solicitation of an offer to buy any securities of the Company.

(d) Exhibits Index	
Exhibit No.	Description
4.1	Confessed Judgment Secured Promissory Note, dated February 10, 2025, issued to Agile Capital Funding, LLC, by Super League Enterprise, Inc.
5.1	Opinion of Disclosure Law Group, a Professional Corporation.
10.1*	Business Loan and Security Agreement, dated February 10, 2025, between Super League Enterprise, Inc. and Agile Capital Funding, LLC.
10.2	Equity Purchase Agreement, dated February 14, 2025, between Super League Enterprise, Inc. and Hudson Global Ventures, LLC.
10.3	Registration Rights Agreement, dated February 14, 2025, between Super League Enterprise, Inc., and Hudson Global Ventures, LLC.
23.1	Consent of Disclosure Law Group, a Professional Corporation (included in Exhibit 5.1).
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

^{*}Certain schedules and exhibits to this agreement have been omitted pursuant to Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished to the SEC upon request.

Signatures

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Super League Enterprise, Inc.

Date: February 14, 2025 By: /s/ Clayton Haynes

Clayton Haynes Chief Financial Officer

CONFESSED JUDGMENT SECURED PROMISSORY NOTE

IMPORTANT NOTICE: THIS INSTRUMENT CONTAINS A CONFESSION OF JUDGMENT PROVISION WHICH CONSTITUTES A WAIVER OF IMPORTANT RIGHTS YOU MAY HAVE AS A DEBTOR AND ALLOWS THE CREDITOR TO OBTAIN A JUDGMENT AGAINST YOU WITHOUT ANY FURTHER NOTICE.

CONFESSED JUDGMENT SECURED PROMISSORY NOTE

\$2,500,000.00 Dated: February 10, 2025

FOR VALUE RECEIVED, the undersigned SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation ("Parent" or "Borrower") and its subsidiaries, INPVP, LLC, A Domestic Delaware Corporation, individually and collectively, jointly and severally, "Guarantors"), HEREBY JOINTLY AND SEVERALLY PROMISE TO PAY to the order of Agile Lending, LLC, or its designees or assigns ("Lead Lender") the principal amount of TWO MILLION FIVE HUNDRED THOUSAND DOLLARS (\$2,500,000.00) or such lesser amount as shall equal the outstanding principal balance of the Term Loan made to Borrower by Lender, plus interest on the aggregate unpaid principal amount of such Term Loan, at the rates and in accordance with the terms of the Business Loan and Security Agreement dated February 10, 2025, by and among Borrower, Lender, Collateral Agent, and the other Lenders from time to time party thereto (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"). If not sooner paid, the entire principal amount and all accrued and unpaid interest hereunder shall be due and payable on the Maturity Date as set forth in the Loan Agreement. Any capitalized term not otherwise defined herein shall have the meaning attributed to such term in the Loan Agreement.

Principal, interest and all other amounts due with respect to the Term Loan, are payable in lawful money of the United States of America to Lender as set forth in the Loan Agreement and this Confessed Judgment Secured Promissory Note (this "Note").

The Loan Agreement, among other things, (a) provides for the making of a secured Term Loan by Lender to Borrower, and (b) contains provisions for acceleration of the maturity hereof upon the happening of certain stated events.

This Note may not be prepaid except as set forth in Section 2.2 (c) and Section 2.2(d) of the Loan Agreement.

This Note and the obligation of Borrower to repay the unpaid principal amount of the Term Loan, interest on the Term Loan and all other amounts due Lender under the Loan Agreement is secured as provided under the Loan Agreement.

Presentment for payment, demand, notice of protest and all other demands and notices of any kind in connection with the execution, delivery, performance and enforcement of this Note are hereby waived.

Borrower shall pay all reasonable fees and expenses, including, without limitation, reasonable attorneys' fees and costs, incurred by Lender in the enforcement or attempt to enforce any of Borrower's obligations hereunder not performed when due.

This Note shall be governed by, and construed and interpreted in accordance with, the internal laws of the Commonwealth of Virginia.

The ownership of an interest in this Note shall be registered on a record of ownership maintained by Lender or its agent. Notwithstanding anything else in this Note to the contrary, the right to the principal of, and stated interest on, this Note may be transferred only if the transfer is registered on such record of ownership and the transferee is identified as the owner of an interest in the obligation. Borrower shall be entitled to treat the registered holder of this Note (as recorded on such record of ownership) as the owner in fact thereof for all purposes and shall not be bound to recognize any equitable or other claim to or interest in this Note on the part of any other person or entity.

BORROWER HEREBY KNOWINGLY, VOLUNTARILY AND INTELLIGENTLY WAIVES ANY AND ALL RIGHTS THAT EACH PARTY TO THIS NOTE MAY NOW OR HEREAFTER HAVE UNDER THE LAWS OF THE UNITED STATES OF AMERICA OR THE COMMONWEALTH OF VIRGINIA, TO A TRIAL BY JURY OF ANY AND ALL ISSUES ARISING DIRECTLY OR INDIRECTLY IN ANY ACTION OR PROCEEDING RELATING TO THIS NOTE, THE LOAN DOCUMENTS OR ANY TRANSACTIONS CONTEMPLATED THEREBY OR RELATED THERETO. IT IS INTENDED THAT THIS WAIVER SHALL APPLY TO ANY AND ALL DEFENSES, RIGHTS, CLAIMS AND/OR COUNTERCLAIMS IN ANY SUCH ACTION OR PROCEEDING. BORROWER EXPRESSLY ACKNOWLEDGES THAT BORROWER IS SUBJECT TO PERSONAL JURISDICTION IN THE COMMONWEALTH OF THE VIRGINIA, THAT BORROWER INTENTIONALLY ENTERED INTO THE TRANSACTIONS THAT ARE THE SUBJECT OF THIS CONFESSED JUDGMENT PROMISSORY NOTE WITH LENDER, WHO IS LOCATED IN THE COMMONWEALTH OF VIRGINIA, AND THAT BORROWER WAIVES ANY AND ALL OBJECTIONS TO THE EXERCISE OF PERSONAL JURISDICTION OVER BORROWER OF THE COMMONWEALTH OF VIRGINIA AND TO VENUE IN THE CIRCUIT COURT FOR ARLINGTON COUNTY, VIRGINIA AND ANY OTHER COURT WITHIN THE COMMONWEALTH OF VIRGINIA.

BORROWER UNDERSTANDS THAT THIS WAIVER IS A WAIVER OF A CONSTITUTIONAL SAFEGUARD, AND EACH PARTY INDIVIDUALLY BELIEVES THAT THERE ARE SUFFICIENT ALTERNATE PROCEDURAL AND SUBSTANTIVE SAFEGUARDS, INCLUDING, A TRIAL BY AN IMPARTIAL JUDGE, THAT ADEQUATELY OFFSET THE WAIVER CONTAINED HEREIN.

UPON THE OCCURRENCE OF AN EVENT OF DEFAULT HEREUNDER OR UNDER THE LOAN AGREEMENT, LEAD LENDER MAY CONFESS JUDGMENT AGAINST BORROWER AS PROVIDED HEREIN. UPON THE OCCURRENCE OF ANY EVENT OF DEFAULT HEREUNDER, BORROWER HEREBY AUTHORIZES AND EMPOWERS THE CLERK OF ANY COURT OF RECORD IN THE COMMONWEALTH OF VIRGINIA, INCLUDING BUT NOT LIMITED TO THE CLERK OF THE CIRCUIT COURT FOR THE COUNTY OF ARLINGTON TO ENTER JUDGMENT BY CONFESSION AGAINST BORROWER IN FAVOR OF LEAD LENDER FOR THE FULL AMOUNT DUE AND PAYABLE UNDER THE FINANCING AGREEMENTS AND SECURED BY THE LOAN AGREEMENT, TOGETHER WITH ALL PERMITTED FEES AND INTEREST, AS EVIDENCED BY AN AFFIDAVIT SIGNED BY AN OFFICER OF LEAD LENDER SETTING FORTH THE AMOUNT THEN DUE, TOGETHER WITH REASONABLE ATTORNEYS 'FEES AND COLLECTION COSTS INCURRED BY LEAD LENDER AS PROVIDED IN THIS INSTRUMENT, TO THE EXTENT PERMITTED BY LAW, EXPRESSLY WAIVING SUMMONS AND OTHER PROCESS, AND DOES HEREBY CONSENT TO THE IMMEDIATE EXECUTION OF SUCH JUDGMENT, EXPRESSLY WAIVING THE BENEFIT OF ALL EXEMPTION OR HOMESTEAD LAWS.

BORROWER HEREBY CONSTITUTES AND APPOINTS JODIE E. BUCHMAN, ESQ., PIERCE C. MURPHY, ESQ., OF SILVERMAN, THOMPSON, SLUTKIN & WHITE, 400 E PRATT ST, SUITE 900, BALTIMORE, MD, 21202, OR A DULY APPOINTED SUBSTITUTE AS THE TRUE AND LAWFUL ATTORNEY-IN-FACT FOR BORROWER AND ALL PERSONS CLAIMING THROUGH OR UNDER BORROWER TO SIGN AN AGREEMENT FOR ENTERING IN ANY COMPETENT COURT AN AMICABLE ACTION IN EJECTMENT FOR POSSESSION OF THE COLLATERAL AND/OR TO APPEAR IN THE CLERK'S OFFICE OF THE CIRCUIT COURT OF ARLINGTON COUNTY, VIRGINIA, OR ANY COURT OF COMPETENT JURISDICTION AND TO CONFESS JUDGMENT AGAINST BORROWER, AND ALL PERSONS CLAIMING UNDER OR THROUGH BORROWER IN FAVOR OF LEAD LENDER, FOR WHICH THIS NOTE, OR A COPY THEREOF VERIFIED BY AFFIDAVIT, SHALL BE SUFFICIENT WARRANT; WHEREUPON A WRIT OF POSSESSION MAY IMMEDIATELY ISSUE FOR POSSESSION OF THE COLLATERAL, WITHOUT ANY PRIOR WRIT OR PROCEEDING WHATSOEVER AND WITHOUT ANY STAY OF EXECUTION. LEAD LENDER MAY BRING AN AMICABLE ACTION IN EJECTMENT AND/OR CONFESS JUDGMENT THEREIN EITHER BEFORE OR AFTER THE INSTITUTION OF PROCEEDINGS TO ENFORCE THIS NOTE AND/OR AFTER ENTRY OF JUDGMENT ON THIS NOTE, OR AFTER A PUBLIC SALE OF THE COLLATERAL IN WHICH LEAD LENDER IS THE SUCCESSFUL BIDDER.

BORROWER HEREBY RATIFIES AND CONFIRMS ALL THAT SAID ATTORNEY OR ATTORNEYS MAY DO PURSUANT TO THE FOREGOING POWER. PURSUANT TO SECTION 8.01-435 OF THE CODE OF VIRGINIA OF 1950, AS AMENDED, BORROWER IS HEREBY NOTIFIED THAT A SUBSTITUTE ATTORNEY-IN-FACT UNDER THIS PARAGRAPH MAY BE APPOINTED BY THE LEAD LENDER, OBLIGEE, OR PERSON OTHERWISE ENTITLED TO PAYMENT UNDER THIS AGREEMENT BY RECORDING AN INSTRUMENT NAMING SUCH SUBSTITUTE ATTORNEY-IN-FACT IN THE CLERK'S OFFICE WHERE JUDGMENT IS TO BE CONFESSED.

THE FOREGOING AUTHORIZATION TO PURSUE PROCEEDINGS FOR CONFESSING JUDGMENT AND ANY AND ALL JUDGMENT ENFORCEMENT MEASURES THAT LEAD LENDER OPTS TO PURSUE, INCLUDING BUT NOT LIMITED TO OBTAINING POSSESSION OF THE COLLATERAL, AND IS AN ESSENTIAL PART OF LEAD LENDER'S REMEDIES FOR ENFORCEMENT OF THIS NOTE AND THE LOAN AGREEMENT AND SHALL SURVIVE ANY ENFORCEMENT ACTIONS OR FORECLOSURE SALE BY OR TO LEAD LENDER.

[Signature Page to Follow]

]	BORROWER:
Ī	s/ Clayton Haynes By: CLAYTON JAMES HAYNES Date: February 10, 2025
STA'	TE: NTY OF:
HAY Dom execu	by certify that on February 10, 2025, before me, the undersigned, Notary Public in and for the State of California, at large, personally appeare CLAYTON JAMES NES, individually and as the CFO of SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation ('Parent') and its subsidiaries, INPVP, LLC, A estic Delaware Corporation known to me or satisfactorily proven to be the person whose name is subscribed to the foregoing instrument and acknowledged that he ted the foregoing on behalf of himself individually, SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation ("Parent") and its subsidiaries, P, LLC, A Domestic Delaware Corporation for the purposes set forth therein.

IN WITNESS WHEREOF, Borrower caused this Note to be duly executed under seal by one of its officers thereunto duly authorized on the date hereof.

(Seal)
Notary Public _____
My Commission Expires: Registration Number:



February 14, 2024

Super League Gaming, Inc. 2912 Colorado Avenue, Suite 203 Santa Monica, California 90404

Ladies and Gentlemen:



You have requested our opinion, as counsel to Super League Enterprise, Inc., a Delaware corporation (the "Company"), with respect to certain matters in connection with the proposed offer and sale by the Company of up to \$3,100,000 in newly issued shares of its common stock, par value \$0.001 (the "Shares") to be issued pursuant to an "equity line of credit" pursuant to the terms of an Equity Purchase Agreement, dated February 14, 2024 (the "Purchase Agreement"), by and between the Company and Hudson Global Ventures, LLC (the "Hudson"). The Shares are being offered and sold pursuant to (i) a currently effective shelf registration statement on Form S-3 (File No. 333-283812) that was originally filed under the Securities Act of 1933, as amended (the "Securities Act"), with the Securities and Exchange Commission (the "Commission") on December 13, 2024 and declared effective by the Commission on December 20, 2024 (as amended, the "Registration Statement"), (ii) the base prospectus contained in the Registration Statement (the "Base Prospectus"), and (iii) the prospectus supplement relating to the offering filed with the Commission pursuant to Rule 424(b) under the Securities Act (the "Prospectus Supplement," and, together with the Base Prospectus, the "Prospectus"). This opinion letter is furnished to you at your request to enable you to fulfill the requirements of Item 601(b)(5) of Regulation S-K, 17 C.F.R. § 229.601(b)(5), in connection with the Registration Statement.

In connection with the preparation of this opinion, we have examined and relied upon the Registration Statement and the Prospectus, the Company's Second Amended and Restated Certificate of Incorporation and Amended and Restated Bylaws, as currently in effect, and the originals or copies certified to our satisfaction of such records, documents, certificates, memoranda and other instruments as we have deemed relevant in connection with this opinion. We have assumed the genuineness and authenticity of all documents submitted to us as originals, the conformity to originals of all documents submitted to us as copies, and the accuracy, completeness and authenticity of certificates of public officials.

The opinions set forth in this letter are limited to the Delaware General Corporation Law and the law of the State of California, in each case as in effect on the date hereof. We are not rendering any opinion as to compliance with any federal or state antifraud law, rule or regulation relating to securities or to the sale or issuance thereof. On the basis of the foregoing, and in reliance thereon, and subject to the qualifications herein stated, we are of the opinion that the Shares have been duly authorized by all necessary corporate action on the part of the Company and when issued and sold in the manner described in the Registration Statement, the Prospectus and the Purchase Agreement, will be validly issued, fully paid and non-assessable.

We consent to the reference to our firm under the caption "Legal Matters" in the Prospectus and to the filing of this opinion as an exhibit to a Current Report on Form 8-K to be filed with the Commission for incorporation by reference into the Registration Statement. In giving this consent, we do not admit that we are "experts" within the meaning of Section 11 of the Act or within the category of persons whose consent is required by Section 7 of the Securities Act.

This opinion is intended solely for use in connection with the offer and sale of the Shares pursuant to the Registration Statement and the Prospectus, and may only be relied upon by you and by persons entitled by law to rely upon it pursuant to the applicable provisions of the federal securities laws. This opinion is rendered as of the date hereof and based solely on our understanding of facts in existence as of such date after the examination described in this opinion. We assume no obligation to advise you of any fact, circumstance, event or change in the law or the facts that may hereafter be brought to our attention whether or not such occurrence would affect or modify the opinions expressed herein.

Very truly yours,

/s/ Disclosure Law Group
Disclosure Law Group, a Professional Corporation

600 West Broadway Suite 700 San Diego CA 92101 T 619.272.7050

www.disclosurelawgroup.com

BUSINESS LOAN AND SECURITY AGREEMENT

THIS BUSINESS LOAN AND SECURITY AGREEMENT (as the same may be amended, restated, modified, or supplemented from time to time, this 'Agreement') dated as of February 10, 2025 (the "Effective Date") among Agile Capital Funding, LLC as collateral agent (in such capacity, together with its successors and assigns in such capacity, "Collateral Agent"), and Agile Lending, LLC, a Virginia limited liability company ("Lead Lender") and each assignee that becomes a party to this Agreement pursuant to Section 12.1 (each individually with the Lead Lender, a "Lender" and collectively with the Lead Lender; the "Lenders"), and SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation ("Parent" or "Borrower") and its <u>subsidiaries</u>, INPVP, LLC, A Domestic Delaware Corporation, individually and collectively, jointly and severally, "Guarantors"), and provides the terms on which the Lenders shall lend to Borrower and Borrower shall repay the Lenders the loans described herein. The Collateral Agent, Lenders, and Borrower, each a "Party" and collectively the "Parties", intending to be legally bound, hereby agree as follows:

1. DEFINITIONS, ACCOUNTING AND OTHER TERMS

1.1 Capitalized terms used herein shall have the meanings set forth in Section 13 to the extent defined therein. All other capitalized terms used but not defined herein shall have the meaning given to such terms in the Code. Any accounting term used but not defined herein shall be construed in accordance with GAAP and all calculations shall be made in accordance with GAAP. The term "financial statements" shall include the accompanying notes and schedules thereto. Any section, subsection, schedule or exhibit references are to this Agreement unless otherwise specified.

2. LOANS AND TERMS OF PAYMENT

2.1 Promise to Pay. Borrower hereby unconditionally promises to pay each Lender the outstanding principal amount of the Term Loan advanced to Borrower by such Lender and accrued and unpaid interest thereon and any other amounts due hereunder as and when due in accordance with this Agreement.

2.2 Term Loans.

- (a) <u>Availability</u>. The Lenders, relying upon each of the representations and warranties set out in this Agreement, as well as each of the representations, covenants and warranties set out in the other Loan Documents, hereby severally and not jointly agree with the Borrower that, subject to and upon the terms and conditions of this Agreement, shall advance the Principal Loan to the Borrower on the Effective Date, but in any event no later than two (2) Business Days after the date hereof, by wiring the funds to the Borrower's Account.
- (b) Repayment. Borrower agrees to pay all amounts owing pursuant to the terms of this Agreement, including any financing charge, specified fees, interest and any other charges that may be assessed as provided in this Agreement or as documented in the Business Loan and Security Agreement Supplement (the "Supplement") or the Secured Promissory Note (as defined below). The Term Loan shall be repaid by Borrower on the dates specified on Exhibit B-4 of this Agreement (each a "Scheduled Repayment Date") by the amount set out opposite each Scheduled Repayment Date (each a "Scheduled Repayment Amount") and in accordance with the Term Loan Amortization Schedule. If any payment on the Secured Promissory Note is due on a day which is not a Business Day, such payment shall be due on the next succeeding Business Day, and such extension of time shall be taken into account in calculating the amount of interest payable under this Note. All unpaid principal and accrued and unpaid interest with respect to the Term Loan is due and payable in full on the Maturity Date. The Term Loan may only be prepaid in accordance with Sections 2.2(c) and 2.2(d). Once repaid, no portion of the Term Loan may be reborrowed.
- (c) <u>Mandatory Prepayments</u>. If an event described in Section 7.2 hereof occurs, or the Term Loan is accelerated following the occurrence of an Event of Default, Borrower shall immediately pay to Lenders, payable to each Lender in accordance with its respective Pro Rata Share, an amount equal to the sum of: (i) all outstanding principal of the Term Loans plus accrued and unpaid interest thereon through the prepayment date, (ii) the Prepayment Fee (as defined in Section 2.2(d) below), plus (iii) all other Obligations that are due and payable, including, without limitation, interest at the Default Rate with respect to any past due amounts.

(d) <u>Permissive Prepayments and Make-Whole Premium.</u> Borrower shall have the right to make a full prepayment or partial prepayment of any or all of the Obligations in accordance with the prepayment amendment in Exhibit E of this Agreement. The foregoing notwithstanding, upon the prepayment of any principal amount, Borrower shall be obligated to pay a make-whole premium payment on account of such principal so paid, which shall be equal to the aggregate and actual amount of interest (at the contract rate of interest) that would be paid through the Maturity Date ("**Prepayment Fee**").

2.3 Payment of Interest on the Term Loans.

- (a) Interest Rate. Borrower agrees to pay in full the interest as set forth in the Supplement found in Exhibit B-5 of this Agreement. Interest shall accrue on the Term Loan commencing on, and including, the Effective Date of such Term Loan, and shall accrue on the principal amount outstanding under the Term Loan through and including the day on which the Term Loan is paid in full.
- (b) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, Obligations shall accrue interest at a fixed per annum rate equal to the rate that is otherwise applicable thereto <u>plus</u> five percentage points (5.00%) (the '**Default Rate**'). Payment or acceptance of the increased interest rate provided in this Section 2.3(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Collateral Agent.
 - (c) 360 Day Year. Interest shall be computed on the basis of a three hundred sixty (360) day year and the actual number of days elapsed.
- (d) <u>Debit of Accounts; Payments</u>. All payments on the Secured Promissory Note shall be made via automated clearing house transfers of immediately available funds to be initiated by Lender in accordance with the authorization and direction of Borrower to Lead Lender provided in Exhibit B-6 of this Agreement.
- (e) <u>Usury Savings Clause</u>. This Agreement and the other Loan Documents are subject to the express condition that at no time shall Borrower be required to pay interest on the principal balance of the Term Loan at a rate which could subject Lenders to either civil or criminal liability as a result of being in excess of the Maximum Legal Rate. If by the terms of this Agreement or the other Loan Documents, Borrower is at any time required or obligated to pay interest on the principal balance due hereunder at a rate in excess of the Maximum Legal Rate, the Interest Rate or the Default Rate, as the case may be, shall be deemed to be immediately reduced to the Maximum Legal Rate and all previous payments in excess of the Maximum Legal Rate shall be deemed to have been payments in reduction of principal and not on account of the interest due hereunder. All sums paid or agreed to be paid to the Collateral Agent or Lenders for the use, forbearance, or detention of the sums due under the Loan, shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread throughout the full stated term of the Loan until payment in full.
 - **2.4 Fees**. Borrower shall pay to Collateral Agent and/or Lenders:
- (a) Administrative Agent Fee. The Administrative Agent Fee of ONE HUNDRED TWENTY FIVE THOUSAND DOLLARS (\$125,000.00, which shall be paid at closing out of proceeds of the Term Loan for the account of Collateral Agent.
- 2.5 Secured Promissory Notes. The Term Loan shall be evidenced by a Secured Promissory Note in the form attached as Exhibit D hereto (Secured Promissory Note") and shall be repayable as set forth in this Agreement.

3. CONDITIONS OF LOANS

3.1 Conditions Precedent to Term Loan. Each Lender's obligation to make the Term Loan is subject to the condition precedent that each Lender shall consent to or shall have received, in form and substance satisfactory to each Lender, such documents, and completion of such other matters, as each Lender may reasonably deem necessary or appropriate.

4. CREATION OF SECURITY INTEREST

- 4.1 Grant of Security Interest. Effective from and after the Effective Date of the Term Loan, Borrower hereby grants Collateral Agent, for the ratable benefit of the Lenders, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Collateral Agent, for the ratable benefit of the Lenders, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. If Borrower shall acquire a commercial tort claim (as defined in the Code), Borrower shall grant to Collateral Agent, for the ratable benefit of the Lenders, a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Collateral Agent. If this Agreement is terminated, Collateral Agent's Lien in the Collateral shall continue until the Obligations (other than inchoate indemnity obligations) and at such time as the Lenders' obligation to extend the Term Loan has terminated, Collateral Agent shall, at the sole cost and expense of Borrower, release its Liens in the Collateral and all rights therein shall revert to Borrower.
- 4.2 Authorization to File Financing Statements. Borrower hereby authorizes Collateral Agent to file such financing statements and/or take any other action required to perfect Collateral Agent's security interests in the Collateral, without notice to Borrower, with all appropriate jurisdictions to perfect or protect Collateral Agent's interest or rights in the Collateral and under the Loan Documents; provided, however. Collateral Agent may only file such financing statements and/or take any other action required to perfect Collateral Agent's security interests in the Collateral, upon the occurrence of an Event of Default
 - 4.3 Guaranty. (Intentionally omitted).

5. REPRESENTATIONS AND WARRANTIES

Each Borrower, jointly and severally, represents and warrants to Collateral Agent and the Lenders as follows:

- **5.1 Due Organization, Authorization: Power and Authority.** Each Borrower and each of its respective Subsidiaries is duly formed, validly existing and in good standing as under the laws of its jurisdiction of organization or formation and each Borrower and each of its respective Subsidiaries is qualified and licensed to do business and is in good standing in any jurisdiction in which the conduct of its businesses or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to result in a Material Adverse Change.
- 5.2 Collateral. Borrower and Subsidiaries have good title to, have rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien under the Loan Documents, free and clear of any and all Liens except Permitted Liens, and neither Borrower nor any of its Subsidiaries have any deposit accounts, securities accounts, commodity accounts or other investment accounts of the "Collateral Accounts"), if any, described in the Perfection Certificates delivered to Collateral Agent in connection herewith with respect to which Borrower has given Collateral Agent notice and taken, subject to Section 6.6 (a), such actions as are necessary to give Collateral Agent a perfected security interest therein. The security interests granted herein are and shall at all times continue to be a first priority perfected security interest in the Collateral, subject only to Permitted Liens that are permitted by the terms of this Agreement to have priority to Collateral Agent's Lien. All Inventory and Equipment that is part of the Collateral is in all material respects of good and marketable quality, free from material defects.
- 5.3 Litigation. Except as disclosed on the Perfection Certificate, there are no actions, suits, investigations, or proceedings pending or, to the knowledge of any of the Responsible Officers, threatened in writing by or against Borrower or any of its Subsidiaries involving more than Fifty Thousand Dollars (\$50,000.00).

- 5.4 No Material Adverse Change; Financial Statements All consolidated financial statements for Parent and its Subsidiaries, delivered to Collateral Agent fairly present, in conformity with GAAP, in all material respects the consolidated financial condition of Parent and its Subsidiaries, and the consolidated results of operations of Parent and its Subsidiaries. Since the date of the most recent financial statements submitted to any Lender, there has not been a Material Adverse Change.
 - 5.5 Solvency. As of September 30, 2024, Borrower and each of its Subsidiaries, when taken as a whole, is Solvent.
- 5.6 Regulatory Compliance. Neither Borrower nor any of its Subsidiaries has violated any laws, ordinances or rules, the violation of which could reasonably be expected to result in a Material Adverse Change. Borrower and each of its Subsidiaries has obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all governmental authorities that are necessary to continue their respective businesses as currently conducted.
- 5.7 Investments. Neither Borrower nor any of its Subsidiaries owns any stock, shares, partnership interests or other equity securities except for Permitted Investments.
- 5.8 Tax Returns and Payments; Pension Contributions. Each Borrower and each of its respective Subsidiaries has timely filed all required tax returns and reports, and, except as disclosed, each Borrower and each of its respective Subsidiaries, has timely paid all foreign, federal, state, and local taxes, assessments, deposits and contributions owed by such Borrower and such Subsidiaries, in all jurisdictions in which such Borrower or any such Subsidiary is subject to taxes, including the United States, unless such taxes are being contested in good faith.
- 5.9 Use of Proceeds. Borrower shall use the proceeds of the Term Loan to pay off existing balance of \$1,501,142.84 for MID:434125 funded by Agile Capital Funding, LLC, on November 13, 2024 and to fund its general business requirements in accordance with the provisions of this Agreement, and not for personal, family, household or agricultural purposes.
- 5.10 Full Disclosure. No written representation, warranty or other statement of any Borrower or any of its Subsidiaries in any certificate or written statement given to Collateral Agent or any Lender, as of the date such representation, warranty, or other statement was made, taken together with all such written certificates and written statements given to Collateral Agent or any Lender, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the certificates or statements not misleading (it being recognized that projections and forecasts provided by Borrower in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).
- 5.11 Shares. Each Borrower has full power and authority to create a first lien on its Shares and no disability or contractual obligation exists that would prohibit such Borrower from pledging the Shares pursuant to this Agreement. To Borrower's knowledge, there are no subscriptions, warrants, rights of first refusal or other restrictions on transfer relative to, or options exercisable with respect to the Shares. With respect to each Subsidiary which is a corporation, the Shares have been and will be duly authorized and validly issued, and are fully paid and non-assessable. To Borrower's knowledge, the Shares are not the subject of any present or threatened suit, action, arbitration, administrative or other proceeding, and Borrower knows of no reasonable grounds for the institution of any such proceedings.
 - 5.12 Guarantee. (Intentionally omitted)

6. AFFIRMATIVE COVENANTS

Borrower shall, and shall cause each of its Subsidiaries to, do all of the following:

6.1 Government Compliance. Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of organization and maintain qualification in each jurisdiction in which the failure to so qualify could reasonably be expected to have a Material Adverse Change.

6.2 Financial Statements, Reports, Certificates, Notices

- (a) Deliver to Collateral Agent and each Lender: (i) as soon as available, but no later than forty-five (45) days after the last day of each month, a company prepared consolidated and consolidating balance sheet, income statement and cash flow statement covering the consolidated operations of Parent and its Subsidiaries for such month certified by a Responsible Officer and in a form reasonably acceptable to Collateral Agent; (ii) prompt notice of any material amendments of or other changes to the capitalization table of Borrower (other than Parent) and to the Operating Documents of Borrower or any of its Subsidiaries, together with any copies reflecting such amendments or changes with respect thereto; (iii) as soon as available, but no later than forty-five (45) days after the last day of each month, copies of the month end account statements for each Collateral Account maintained by Borrower or its Subsidiaries, which statements may be provided to Collateral Agent and each Lender by Borrower or directly from the applicable institution(s); (iv) prompt notice of any event that (A) could reasonably be expected to materially and adversely affect the Borrower's Intellectual Property and (B) could reasonably be expected to result in a Material Adverse Change; (v) written notice at least (10) days' prior to Borrower's creation of a new Subsidiary in accordance with the terms of Section 6.10; (vi) written notice at least (30) days' prior to Borrower's (A) changing its jurisdiction of organization, (B) changing its organizational structure or type, (C) changing its legal name, (D) changing any organizational number (if any) assigned by its jurisdiction of organization, or (E) registering or filing any Intellectual Property; (vii) upon Borrower becoming aware of the existence of any Event of Default or event which, with the giving of notice or passage of time, or both, would constitute an Event of Default, prompt (and in any event within three (3) Business Days) written notice of such occurrence, which such notice shall include a reasonably detailed description of such Event of Default or event which, with the giving of notice or passage of time, or both, would constitute an Event of Default; (viii) notice of any commercial tort claim of Borrower or any Guarantor and of the general details thereof; (ix) other information as reasonably requested by Collateral Agent or any Lender. (x) written notice of any litigation or governmental proceedings pending or threatened (in writing) against Borrower or any of its Subsidiaries, which could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of more than Fifty Thousand Dollars (\$50,000.00); and (xi) written notice of all returns, recoveries, disputes and claims regarding Inventory that involve more than Fifty Thousand Dollars (\$50,000.00) individually or in the aggregate in any calendar year.
- (b) Keep proper, complete and true books of record and account in accordance with GAAP and in all material respects. Borrower shall, and shall cause each of its Subsidiaries to, allow, at the sole cost of Borrower, Collateral Agent or any Lender, during regular business hours upon reasonable prior notice (provided that no notice shall be required when an Event of Default has occurred and is continuing), to visit and inspect any of its properties, to examine and make abstracts or copies from any of its books and records, and to conduct a collateral audit and analysis of its operations and the Collateral. Such audits shall be conducted no more often than twice every year unless (and more frequently if) an Event of Default has occurred and is continuing. Notwithstanding the foregoing, upon request of any Lender, Borrower agrees to permit such Lender to communicate with Borrower's accounting firm, in the presence of a Responsible Officer of the Borrower or the Parent, with respect to the consolidated financial statements delivered pursuant to this Section 6.2.
- **6.3 Inventory and Returns**. Keep all Inventory in good and marketable condition, free from material defects. Returns and allowances between Borrower, or any of its Subsidiaries, and their respective account debtors shall follow Borrower's, or such Subsidiary's, customary practices as they exist at the Effective Date.
- **6.4** Taxes. Timely file and require each of its Subsidiaries to timely file, all required tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all foreign, federal, state, and local taxes, assessments, deposits and contributions owed by Borrower or its Subsidiaries, except as otherwise permitted pursuant to the terms of Section 5.8 hereof.
- 6.5 Insurance. Keep Borrower's and its Subsidiaries' business and the Collateral insured for risks and in amounts standard for companies in Borrower's and its Subsidiaries' industry and location and as Collateral Agent may reasonably request (including customary lender's loss payable endorsements and naming the Collateral Agent as an additional insured), and give the Collateral Agent thirty (30) days' prior written notice before any such policy or policies shall be materially altered or canceled (other than cancellation for non-payment of premiums, for which ten (10) days' prior written notice shall be required). At Collateral Agent's request, Borrower shall deliver certified copies of policies and evidence of all premium payments to Collateral Agent. If Borrower or any of its Subsidiaries fails to obtain insurance as required under this Section 6.5 or to pay any amount or furnish any required proof of payment to third persons, Collateral Agent and/or any Lender may make (but has no obligation to do so), at Borrower's expense, all or part of such payment or obtain such insurance policies required in this Section 6.5, and take any action under the policies Collateral Agent or such Lender deems prudent.

- **6.6 Operating Accounts.** Borrower shall provide Collateral Agent ten (10) days' prior written notice before Borrower or any of its Subsidiaries establishes any Collateral Account.
- 6.7 Litigation Cooperation. Commencing on the Effective Date and continuing through the termination of this Agreement, make available to Collateral Agent and the Lenders, without expense to Collateral Agent or the Lenders, Borrower and each of Borrower's officers, employees and agents and Borrower's books and records, to the extent that Collateral Agent or any Lender may reasonably deem them necessary to prosecute or defend any third party suit or proceeding instituted by or against Collateral Agent or any Lender with respect to any Collateral or relating to Borrower.
- **6.8 Landlord Waivers; Bailee Waivers.** In the event that Borrower, after the Effective Date, intends to add any new offices or business locations, including warehouses, or otherwise store any portion of the Collateral with, or deliver any portion of the Collateral to, a bailee, in each case pursuant to Section 7.2, then Borrower must first receive the written consent of Collateral Agent to do so.
- 6.9 Further Assurances. Execute any further instruments and take any and all further action as Collateral Agent or any Lender reasonably requests to perfect or continue Collateral Agent's Lien in the Collateral or toeffect the purposes of this Agreement, including without limitation, permit Collateral Agent or any Lender to discuss Borrower's financial condition with Borrower's accountants in the presence of a Responsible Officer of the Borrower or the Parent.
- 6.10 Lockbox Agreement. Upon the request of any Lender at any time after the Effective Date and for any reason in Lenders' sole and absolute discretion,
 Borrower shall enter into a lockbox arrangement with Lenders with respect to Borrower's accounts receivable at a financial institution of the Lenders' choosing in their sole
 and absolute discretion and shall execute a deposit control agreement in favor of Lenders in a form satisfactory to Lenders in their sole and absolute discretion; provided,
 however, a lockbox agreement will not be utilized unless an Event of Default occurs.

7. NEGATIVE COVENANTS

Borrower shall not, and shall not permit any of its Subsidiaries to, do any of the following without the prior written consent of the Required Lenders, provided, however, all pending transactions by and between Borrower and Infinite Reality, Inc., a Delaware corporation, shall be deemed permissible in all respects and not violative of any provision of this Section 7:

- 7.1 Dispositions. Convey, sell, lease, transfer, assign, dispose of (collectively, 'Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property (including Intellectual Property), except for Transfers (a) of (i) Inventory in the ordinary course of business and (ii) Inventory, that, prior to the Effective Date, has been written down or written off, together with related tangible assets and non-material Intellectual Property; (b) of worn out or obsolete Equipment; (c) in connection with Permitted Liens, Permitted Investments and Permitted Licenses; (d) of any non-material Intellectual Property; (e) from (i) Borrower to another Borrower Guarantor, (ii) a non-Borrower Subsidiary to a Borrower, and (iii) a non-Borrower Subsidiary to another non-Borrower; or (f) permitted under Section 7.3 below.
- 7.2 Changes in Business or Management, Ownership. (a) Engage in or permit any of its Subsidiaries to engage in any business other than the businesses engaged in by Borrower as of the Effective Date or reasonably related thereto; (b) liquidate or dissolve or permit any of its Subsidiaries to liquidate or dissolve; or (c) cause or permit, voluntarily or involuntarily, any Key Person to cease to be actively engaged in the management of Borrower unless written notice thereof is provided to Collateral Agent and each Lender within ten (10) days of such Key Person ceasing to be actively engaged in the management of Borrower.
- 7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock, shares or property of another Person

- 7.4 Indebtedness. Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to doso, other than Permitted Indebtedness. For the avoidance of doubt, Indebtedness includes Merchant Cash Advances.
- 7.5 Encumbrance. Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except for Permitted Liens, or permit any Collateral not to be subject to the first priority security interest granted herein (except for Permitted Liens), or enter into any agreement, document, instrument or other arrangement (except with or in favor of Collateral Agent, for the ratable benefit of the Lenders) with any Person which directly or indirectly prohibits or has the effect of prohibiting Borrower, or any of its Subsidiaries, from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Borrower's or such Subsidiary's Intellectual Property.
 - 7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.6 hereof.
- 7.7 Restricted Payments. Pay any dividends (other than dividends payable solely in capital stock) or make any distribution or payment in respect of or redeem, retire or purchase any capital stock.
 - 7.8 Investments. Directly or indirectly make any Investment other than Permitted Investments, or permit any of its Subsidiaries to do so.
- 7.9 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of Borrower or any of its Subsidiaries (other than among Borrower), except for (a) transactions that are in the ordinary course of Borrower's or such Subsidiary's business, upon fair and reasonable terms that are no less favorable to Borrower or such Subsidiary than would be obtained in an arm's length transaction with a non- affiliated Person, and (b) Subordinated Debt or equity investments by Borrower's investors in Borrower or its Subsidiaries.
 - 7.10 Subordinated Debt. Make or permit any payment on any Subordinated Debt or alternative financings that may encumber any assets of Borrower.
- 7.11 Material Agreements. Other than in the ordinary course of business, (a) enter into a Material Agreement or (b) terminate or materially amend a Material Agreement.
 - 7.12 Financial Covenants. Waived.

8. EVENTS OF DEFAULT

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

- **8.1** Payment Default. Borrower fails to (a) make any payment of principal or interest on the Term Loan on its due date, or (b) pay any other Obligation within three (3) Business Days after such Obligation is due and payable (which three (3) Business Day grace period shall not apply to payments due on the Maturity Date or the date of acceleration pursuant to Section 9.1 (a) hereof.
- **8.2** Covenant Default. Borrower or any of its Subsidiaries fails or neglects to perform any obligation in Sections 6.2 (Financial Statements, Reports, Certificates), 6.4 (Taxes), 6.5 (Insurance), 6.6 (Operating Accounts), or Borrower violates any provision in Section 7.
 - 8.3 Material Adverse Change. A Material Adverse Change has occurred.
 - 8.4 Attachment; Levy; Restraint on Business.
- (a) (i) The service of process seeking to attach, by trustee or similar process, any funds of Borrower or any of its Material Subsidiaries or of any entity under control of Borrower or its Material Subsidiaries on deposit with any institution at which Borrower or any of its Subsidiaries maintains a Collateral Account, or (ii) a notice of lien, levy, or assessment is filed against Borrower or any of its Material Subsidiaries or their respective assets by any government agency, and the same under subclauses (i) and (ii) hereof are not, within ten (10) days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); and

- (b) (i) any material portion of Borrower's or any of its Subsidiaries' assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents Borrower or any of its Subsidiaries from conducting any part of its business;
- **8.5** Insolvency. (a) Parent is or becomes Insolvent; (b) Parent and its Subsidiaries, taken as a whole, are or become Insolvent; (c) Borrower or any Material Subsidiary begins an Insolvency Proceeding; or (d) an Insolvency Proceeding is begun against Borrower or any Material Subsidiary and is not dismissed or stayed within forty five (45) days (but no Term Loan shall be extended while Parent or any Subsidiary is Insolvent and/or until any Insolvency Proceeding is dismissed);
- **8.6** Other Agreements. There is a default in any agreement between Borrower or any of its Subsidiaries and a third party or parties resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness.
- **8.7 Judgments.** (a) One or more judgments, orders, or decrees for the payment of money in an amount, individually or in the aggregate, of at least Two Hundred Fifty Thousand Dollars (\$250,000.00) (not covered by independent third party insurance) shall be rendered against Borrower or any of its Subsidiaries and shall remain unsatisfied, unvacated, or unstayed for a period of twenty (20) days after the entry thereof or (b) any judgments, orders or decrees rendered against Borrower that could reasonably be expected to result in a Material Adverse Change;
- 8.8 Misrepresentations. Borrower or any of its Subsidiaries or any Person acting for Borrower or any of its Subsidiaries makes any representation, warranty, or other statement now or later in this Agreement, any Loan Document or in any writing delivered to Collateral Agent and/or Lenders or to induce Collateral Agent and/or the Lenders to enter this Agreement or any Loan Document, and such representation, warranty, or other statement, when taken as a whole, is incorrect in any material respect when made.
- **8.9** Subordinated Debt. A default or breach occurs under any agreement between Borrower or any of its Subsidiaries and any creditor of Borrower or any of its Subsidiaries that signed a subordination, intercreditor, or other similar agreement with Collateral Agent or the Lenders, or any creditor that has signed such an agreement with Collateral Agent or the Lenders breaches any terms of such agreement;
 - 8.10 Guaranty. (Intentionally Omitted)
- **8.11** Lien Priority. Any Lien created hereunder or by any other Loan Document shall at any time fail to constitute a valid and perfected first Lien on any of the Collateral purported to be secured thereby, subject to no prior or equal Lien, other than Permitted Liens arising as a matter of applicable law.

9. RIGHTS AND REMEDIES

9.1 Rights and Remedies. Upon the occurrence of an Event of Default hereunder (unless all Events of Default have been cured by Borrower, as applicable, or waived by Lenders in writing), Lenders may, at their option: (i) by written notice to Borrower, declare the entire unpaid principal balance of the Term Loan, together with all accrued interest thereon and any other charges or fees payable hereunder, immediately due and payable regardless of any prior forbearance and (ii) exercise any and all rights and remedies available to it hereunder, under the Secured Promissory Note and/or under applicable law, including, without limitation, the right to collect from Borrower all sums due under this Agreement and the Secured Promissory Note and repossess any Collateral at Borrower's expense. Borrower shall pay all reasonable costs and expenses incurred by or on behalf of Lenders or Collateral Agent in connection with Lenders' exercise of any or all of its rights and remedies under this Agreement or the Secured Promissory Note, including, without limitation, reasonable attorneys' fees. Borrower waives the right to any stay of execution and the benefit of all exemption laws now or hereafter in effect.

- 9.2 Power of Attorney. Borrower hereby irrevocably appoints Collateral Agent as its lawful attorney in fact, exercisable upon the occurrence and during the continuance of an Event of Default, to: (a) endorse Borrower's or any of its Subsidiaries' name on any checks or other forms of payment or security; (b) sign Borrower's or any of its Subsidiaries' name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) settle and adjust disputes and claims about the Accounts directly with Account Debtors, for amounts and on terms Collateral Agent determines reasonable; (d) make, settle, and adjust all claims under Borrower's insurance policies; (e) pay, contest or settle any Lien, charge, encumbrance, security interest, and adverse claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of Collateral Agent or a third party as the Code or any applicable law permits. Borrower hereby appoints Collateral Agent as its lawful attorney in fact to sign Borrower's or any of its Subsidiaries' name on any documents necessary to perfect or continue the perfection of Collateral Agent's security interest in, and lien on, the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity obligations) have been satisfied in full and Collateral Agent and the Lenders are under no further obligation to extend the Term Loan hereunder. Collateral Agent's foregoing appointment as Borrower's or any of its Subsidiaries' attorney in fact, and all of Collateral Agent's rights and powers, coupled with an interest, are irrevocable until all Obligations (other than inchoate indemnity obligations) have been fully repaid and performed and Collateral Agent's and the Lenders' obligation to provide the Term Loan terminates.
- 9.3 No Waiver; Remedies Cumulative. Failure by Collateral Agent or any Lender, at any time or times, to require strict performance by Borrower of any provision of this Agreement or any other Loan Document shall not waive, affect, or diminish any right of Collateral Agent or any Lender thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by Collateral Agent and the Required Lenders and then is only effective for the specific instance and purpose for which it is given. The rights and remedies of Collateral Agent and the Lenders under this Agreement and the other Loan Documents are cumulative. Collateral Agent and the Lenders have all rights and remedies provided under the Code, any applicable law, by law, or in equity. The exercise by Collateral Agent or any Lender of one right or remedy is not an election, and Collateral Agent's or any Lender's waiver of any Event of Default is not a continuing waiver. Collateral Agent's or any Lender's delay in exercising any remedy is not a waiver, election, or acquiescence.
- 9.4 Demand Waiver. Borrower waives, to the fullest extent permitted by law, demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Collateral Agent or any Lender on which Borrower or any Subsidiary is liable.

10. NOTICES

All notices, consents, requests, approvals, demands, or other communication (collectively, "Communication") by any party to this Agreement or any other Loan Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by facsimile transmission or e-mail; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address, facsimile number, or email address indicated below. Any of Collateral Agent, any Lender or Borrower may change its mailing address or facsimile number by giving the other party written notice thereof in accordance with the terms of this Section 10.

If to Borrower: Super League Enterprise, Inc. 2856 Colorado Ave. Santa Monica, CA 90404 E-Mail Address: [***]

If to Collateral Agent:

Agile Capital Funding, LLC 244 Madison Ave, Suite 168 New York, NY 10016 E-Mail Address: [***]

11. CHOICE OF LAW, VENUE AND JURY TRIAL WAIVER

11.1 Waiver of Jury Trial. EACH OF BORROWER, COLLATERAL AGENT AND LENDERS UNCONDITIONALLY WAIVES ANY AND ALL RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT, ANY OF THE OTHER LOAN DOCUMENTS, ANY OF THE INDEBTEDNESS SECURED HEREBY, ANY DEALINGS AMONG BORROWER, COLLATERAL AGENT AND/OR LENDERS RELATING TO THE SUBJECT MATTER OF THIS TRANSACTION OR ANY RELATED TRANSACTIONS, AND/OR THE RELATIONSHIP THAT IS BEING ESTABLISHED AMONG BORROWER, COLLATERAL AGENT AND/OR LENDERS. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT. THIS WAIVER IS IRREVOCABLE. THIS WAIVER MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING. THE WAIVER ALSO SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT, ANY OTHER LOAN DOCUMENTS, OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THIS TRANSACTION OR ANY RELATED TRANSACTION. THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

11.2 Governing Law and Jurisdiction.

(a) THIS AGREEMENT, THE OTHER LOAN DOCUMENTS (EXCLUDING THOSE LOAN DOCUMENTS THAT BY THEIR OWN TERMS ARE EXPRESSLY GOVERNED BY THE LAWS OF ANOTHER JURISDICTION) AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL IN ALL RESPECTS BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE COMMONWEALTH OF VIRGINIA (WITHOUT REGARD TO THE CONFLICT OF LAWS PRINCIPLES THAT WOULD RESULT IN THE APPLICATION OF ANY LAWS OTHER THAN THE LAWS OF THE COMMONWEALTH OF VIRGINIA), INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, REGARDLESS OF THE LOCATION OF THE COLLATERAL, PROVIDED, HOWEVER, THAT IF THE LAWS OF ANY JURISDICTION OTHER THAN VIRGINIA SHALL GOVERN IN REGARD TO THE VALIDITY, PERFECTION OR EFFECT OF PERFECTION OF ANY LIEN OR IN REGARD TO PROCEDURAL MATTERS AFFECTING ENFORCEMENT OF ANY LIENS IN COLLATERAL, SUCH LAWS OF SUCH OTHER JURISDICTIONS SHALL CONTINUE TO APPLY TO THAT EXTENT.

- (b) Submission to Jurisdiction. Any legal action or proceeding with respect to the Loan Documents shall be brought exclusively in the courts of the Commonwealth of Virginia, including, without limitation the Circuit Court of Arlington County in the Commonwealth of Virginia and, by execution and delivery of this Agreement, Borrower hereby accepts for itself and in respect of its Property, generally and unconditionally, the jurisdiction of the aforesaid courts. Notwithstanding the foregoing, Collateral Agent and Lenders shall have the right to bring any action or proceeding against Borrower (or any property of Borrower) in the court of any other jurisdiction Collateral Agent or Lenders deem necessary or appropriate in order to realize on the Collateral or other security for the Obligations. The parties hereto hereby irrevocably waive any objection, including any objection to the laying of venue or based on the grounds of forum non conveniens, that any of them may now or hereafter have to the bringing of any such action or proceeding in such jurisdictions. Borrowers expressly acknowledge that they are subject to personal jurisdiction in the Commonwealth of the Virginia, that they intentionally entered into the transactions that are the subject of this Agreement with Collateral Agent and Lender, who are located in the Commonwealth of Virginia, and that Borrowers waive any and all objections to the exercise of personal jurisdiction over them of the Commonwealth of Virginia and to venue in the Circuit Court for Arlington County, Virginia and any other court within the Commonwealth of Virginia.
- (c) Service of Process. Borrower irrevocably waives personal service of any and all legal process, summons, notices and other documents and other service of process of any kind and consents to such service in any suit, action or proceeding brought in the United States of America with respect to or otherwise arising out of or in connection with any Loan Document by any means permitted by applicable requirements of law, including by the mailing thereof (by registered or certified mail, postage prepaid) to the address of Borrower specified herein (and shall be effective when such mailing shall be effective, as provided therein). Borrower agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.
- (d) Non-exclusive Jurisdiction. Nothing contained in this Section 11.2 shall affect the right of Collateral Agent or Lenders to serve process in any other manner permitted by applicable requirements of law or commence legal proceedings or otherwise proceed against Borrower in any other jurisdiction.

12. GENERAL PROVISIONS

- 12.1 Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each Party. Borrower may not transfer, pledge or assign this Agreement or any rights or obligations under it without Collateral Agent's prior written consent (which may be granted or withheld in Collateral Agent's discretion, subject to Section 12.5). The Lenders have the right, without the consent of or notice to Borrower, to sell, transfer, assign, pledge, negotiate, or grant participation in (any such sale, transfer, assignment, negotiation, or grant of a participation, a "Lender Transfer") all or any part of, or any interest in, any one or more Lenders' obligations, rights, and benefits under this Agreement and the other Loan Documents. In the event of such a Lender Transfer, Collateral Agent or Lead Lender shall have the right to, at its respective sole and absolute option, (a) notify Borrower of such Lender Transfer, in accordance with Section 10 hereof, and direct Borrower to make payments directly to such other Lender or Lenders, indicating such other Lenders' Pro Rata share of the Term Loan and the amount of the payment to be made in connection therewith, or (b) continue to collect payments hereunder and under the other Loan Documents and pay such other Lenders their Pro Rata Share of the Term Loan, in accordance with, and on such terms, as are determined by and between the Lenders.
- 12.2 Indemnification. Borrower, jointly and severally, agrees to indemnify, defend and hold Collateral Agent and the Lenders and their respective members, managers, directors, officers, employees, consultants, agents, attorneys, or any other Person affiliated with or representing Collateral Agent or the Lenders (each, an "Indemnified Person") harmless against: (a) all obligations, demands, claims, and liabilities (collectively, Claims") asserted by any other party in connection with; related to; following; or arising from, out of or under, the transactions contemplated by the Loan Documents; and (b) all losses or expenses incurred, or paid by Indemnified Person in connection with; related to; following; or arising from, out of or under, the transactions contemplated by the Loan Documents between Collateral Agent, and/or the Lenders and Borrower (including reasonable attorneys' fees and expenses), except for Claims and/or losses directly caused by such Indemnified Person's gross negligence or willful misconduct. Borrower hereby further, jointly and severally, indemnifies, defends and holds each Indemnified Person harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims, costs, expenses and disbursements of any kind or nature whatsoever (including the fees and disbursements of counsel for such Indemnified Person) in connection with any investigative, response, remedial, administrative or judicial matter or proceeding, whether or not such Indemnified Person shall be designated a party thereto and including any such proceeding initiated by or on behalf of Borrower, and the reasonable expenses of investigation by engineers, environmental consultants and similar technical personnel and any commission, fee or compensation claimed by any broker (other than any broker retained by Collateral Agent or Lenders) asserting any right to payment for the transactions contemplated hereby which may be imposed on, incurred by or asserted against such Indemnified

- 12.3 Severability of Provisions. Each provision of this Agreement is severable from every other provisionin determining the enforceability of any provision.
- 12.4 Correction of Loan Documents. Collateral Agent may correct patent errors and fill in any blanks in this Agreement and the other Loan Documents consistent with the agreement of the parties.
- 12.5 Amendments in Writing; Integration. (a) No amendment, modification, termination or waiver of any provision of this Agreement or any other Loan Document, no approval or consent thereunder, and no consent to any departure by Borrower or any of its Subsidiaries therefrom, shall in any event be effective unless the same shall be in writing and signed by Borrower, Collateral Agent and the Required Lenders provided that:
- (i) no such amendment, waiver or other modification that would have the effect of increasing or reducing a Lender's Term Loan Commitment or Commitment Percentage shall be effective as to such Lender without such Lender's written consent;
- (ii) no such amendment, waiver or modification that would affect the rights and duties of Collateral Agent shall be effective without Collateral Agent's written consent or signature; and
- (iii) no such amendment, waiver or other modification shall, unless signed by all the Lenders directly affected thereby, (A) reduce the principal of, rate of interest on or any fees with respect to the Term Loanor forgive any principal, interest (other than default interest) or fees (other than late charges) with respect to the Term Loan (B) postpone the date fixed for, or waive, any payment of principal of the Term Loan or of interest on the Term Loan (other than default interest) or any fees provided for hereunder (other than late charges or for any termination of any commitment); (C) change the definition of the term "Required Lenders" or the percentage of Lenders which shall be required for the Lenders to take any action hereunder; (D) release all or substantially all of any material portion of the Collateral, authorize Borrower to sell or otherwise dispose of all or substantially all or any material portion of the Collateral, except, in each case with respect to this clause (D), as otherwise may be expressly permitted under this Agreement or the other Loan Documents (including in connection with any disposition permitted hereunder); (E) amend, waive or otherwise modify this Section 12.5 or the definitions of the terms used in this Section 12.5 insofar as the definitions affect the substance of this Section 12.5; (F) consent to the assignment, delegation or other transfer by Borrower of any of its rights and obligations under any Loan Document or release Borrower of its payment obligations under any Loan Document, except, in each case with respect to this clause (F), pursuant to a merger or consolidation permitted pursuant to this Agreement; (G) amend any of the provisions of Section 9.4 or amend any of the definitions of Pro Rata Share, Term Loan Commitment, Commitment Percentage or that provide for the Lenders to receive their Pro Rata Shares of any fees, payments, setoffs or proceeds of Collateral hereunder; (H) subordinate the Liens granted in favor of Collateral Agent securing the Obligations. It is he
- (b) Other than as expressly provided for in Section 12.5(a)(i) (iii), Collateral Agent may, if requested by the Required Lenders, from time to time designate covenants in this Agreement less restrictive by notification to a representative of Borrower.
- (c) This Agreement and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements with respect to such subject matter. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Agreement and the Loan Documents merge into this Agreement and the Loan Documents.

- 12.6 Counterparts; Electronic Signatures. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement. Any and all electronic signatures, whether by scan, e-mail, PDF, Docusign or similar means, and any electronic delivery of signature pages hereto, shall be treated as originals.
- 12.7 Survival. All covenants, representations and warranties made in this Agreement continue in full force and effect until this Agreement has terminated pursuant to its terms and all Obligations (other than inchoate indemnity obligations and any other obligations which, by their terms, are to survive the termination of this Agreement) have been satisfied. The obligation of Borrower in Section 12.2 to indemnify each Lender and Collateral Agent, as well as the confidentiality provisions in Section 12.8 below, shall survive until the statute of limitations with respect to such claim or cause of action shall have run.
- 12.8 Confidentiality. In handling any confidential information of Borrower, the Lenders and Collateral Agent shall exercise the same degree of care that it exercises for their own proprietary information, but disclosure of information may be made: (a) subject to the terms and conditions of this Agreement, to the Lenders' and Collateral Agent's Subsidiaries or Affiliates; (b) to prospective transferees (other than those identified in (a) above) or purchasers of any interest in the Term Loan (provided, however, the Lenders and Collateral Agent shall obtain such prospective transferee's or purchaser's agreement to the terms of this provision or to similar confidentiality terms); (c) as required by law, regulation, subpoena, or other order; (d) to Lenders' or Collateral Agent's regulators or as otherwise required in connection with an examination or audit; (e) as Collateral Agent reasonably considers appropriate in exercising remedies under the Loan Documents; and (f) to third party service providers of the Lenders and/or Collateral Agent so long as such service providers have executed a confidentiality agreement or have agreed to similar confidentiality terms with the Lenders and Collateral Agent with terms no less restrictive than those contained herein. Confidential information does not include information that either: (i) is in the public domain or in the Lenders' and/or Collateral Agent's possession when disclosed to the Lenders and/or Collateral Agent of the public domain after disclosure to the Lenders and/or Collateral Agent at no fault of the Lenders or the Collateral Agent; or (ii) is disclosed to the Lenders and/or Collateral Agent by a third party, if the Lenders and/or Collateral Agent does not know that the third party is prohibited from disclosing the information. Collateral Agent and the Lenders may use confidential information for any purpose, including, without limitation, for the development of client databases, reporting purposes, and market analysis. The provisions of the immediately preceding
- 12.9 Right of Set Off. Borrower hereby grants to Collateral Agent and to each Lender, a lien, security interest and right of set off as security for all Obligations to Collateral Agent and each Lender hereunder, whether now existing or hereafter arising, upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Collateral Agent or the Lenders or any entity under the control of Collateral Agent or the Lenders (including a Collateral Agent affiliate) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Collateral Agent or the Lenders may set off the same or any part thereof and apply the same to any liability or obligation of Borrower even though unmatured and regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE COLLATERAL AGENT TO EXERCISE ITS RIGHTS OR REMEDIES WITHRESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED BY BORROWER.

- 12.10 Borrower Liability. Each Borrower may, acting singly, request credit extensions hereunder. Each Borrower hereby appoints the other as agent for the other for all purposes hereunder, including with respect to requesting credit extensions hereunder. Each Borrower hereunder shall be jointly and severally obligated to repay all credit extensions made hereunder, regardless of which Borrower actually receives said credit extension, as if each Borrower hereunder directly received all credit extensions. Each Borrower waives (a) any suretyship defenses available to it under the Code or any other applicable law, and (b) any right to require Collateral Agent or any Lender to: (i) proceed against any Borrower or any other person; (ii) proceed against or exhaust any security; or (iii) pursue any other remedy. Collateral Agent and/or any Lender may exercise or not exercise any right or remedy it has against any Borrower or any security it holds (including the right to foreclose by judicial or non-judicial sale) without affecting any Borrower's liability. Notwithstanding any other provision of this Agreement or other related document, each Borrower irrevocably waives all rights that it may have at law or in equity (including, without limitation, any law subrogating Borrower to the rights of Collateral Agent and the Lenders under this Agreement) to seek contribution, indemnification or any other form of reimbursement from any other Borrower, or any other Person now or hereafter primarily or secondarily liable for any of the Obligations, for any payment made by Borrower with respect to the Obligations in connection with this Agreement or otherwise and all rights that it might have to benefit from, or to participate in, any security for the Obligations as a result of any payment made by Borrower with respect to the Obligations in connection with this Agreement or otherwise. Any agreement providing for indemnification, reimbursement or any other arrangement prohibited under this Section 12.10 shall be null and void. If any
- 12.11. Change of Law. If, due to any change in applicable law or regulations, or the interpretation thereof by any court of law or other governing body having jurisdiction subsequent to the date of this Agreement, the performance of any provision of this Agreement, the loans granted pursuant hereto or any transaction contemplated hereby shall become unlawful, impracticable or impossible, the Lender shall have the right, with the consent of the Borrower not to be unreasonably withheld, conditioned or delayed, to amend the terms hereof in good faith so as to comply with the then current laws, rules and/or regulations in the way that, in its reasonable judgment, best and most closely reflects the terms and conditions negotiated herein and intended hereby.

13. **DEFINITIONS**

As used in this Agreement, the following terms have the following meanings:

- "Accounts" shall mean accounts receivable of Parent.
- "Affiliate" of any Person is a Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners if such Person is a partnership and, for any Person that is a limited liability company, that Person's managers and members.
- "Borrowing Base" shall mean, at any time, an amount equal to 100% of Eligible Accounts. Business Day" is any day that is not a Saturday, Sunday or a day on which banks are closed in the Commonwealth of Virginia.
- "Code" is the Uniform Commercial Code, as enacted in the Commonwealth of Virginia. "Collateral" is any and all properties, rights and assets of Borrower described on Exhibit A.
 - "Disbursement Instruction Form" is that certain form attached hereto as Exhibit B-2.
 - "Drawdown" means any principal amount borrowed or to be borrowed (by any means) under the provisions hereof.
- "Eligible Accounts" shall mean Accounts that are not excluded as ineligible by virtue of one or more of the criteria set forth below. None of the following shall be Eligible Accounts: (A) Accounts (i) with respect to which the scheduled due date is more than 60 days after the original invoice date, (ii) which are unpaid more than (A) 90 days after the date of the original invoice therefor; (B) Accounts which (i) do not arise from the sale of goods or performance of services in the ordinary course of business, (ii) are not evidenced by an invoice or other documentation reasonably satisfactory to the Collateral Agent, (iii) represent a progress billing, or (iv) are contingent upon any Borrower's completion of any further performance; (C) Accounts which are owed by an account debtor which (i) does not maintain its chief executive office in the United States or (ii) is not organized under any applicable law of the United States, any State of the United States or the District of Columbia; (D) Accounts which are owed in any currency other than dollars; or (E) Accounts which are owed by any Affiliate, employee, officer, director or stockholder of any Borrower or Guarantor.

- "Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.
- "Existing Indebtedness" is the indebtedness of Borrower listed in the Perfection Certificate. "Indebtedness" is (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations, (d) merchant cash advances; and (e) Contingent Obligations in respect of any of the foregoing.
- "Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions or proceedings seeking reorganization, arrangement, or other relief.
 - "Insolvent" means not Solvent.
- "Intellectual Property" shall mean, all (a) trademarks, trademark rights, trade names, trade name rights, service marks, service mark rights, logos, trade dress, domain names, web sites, and all other indicia of origin or quality, and goodwill associated therewith and arising therefrom; (b) patents and patent rights; and (c) works of authorship and copyrights therein, and all common law rights in all of the foregoing, and registration and applications for all of the foregoing issued by or filed with the US Patent and Trademark Office, any State of the US, the US Copyright Office, or any foreign equivalent thereof, and all of the foregoing (a)-(c) used in, at, or in connection with and/or necessary for the (i) conduct of any Borrower's business and/or (ii) use and/or operation of the Collateral.
- "Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made under the Code, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of any Person's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.
- "Investment" is any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.
 - "Key Person" is CLAYTON JAMES HAYNES.
- "Lien" is a mortgage, deed of trust, levy, charge, pledge, security interest, or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.
- "Loan Documents" are, collectively, this Agreement, each Secured Promissory Note, each Disbursement Instruction Form, any subordination agreements, any note, or notes or guaranties executed by Borrower or any other Person, and any other present or future document, certificate, form or agreement entered into by Borrower or any other Person for the benefit of the Lenders and Collateral Agent in connection with this Agreement; all as amended, restated, or otherwise modified or supplemented from time to time.
- "Material Adverse Change" is (a) a material adverse change in the business, operations or condition (financial or otherwise) of Parent, or Parent and each Subsidiary, taken as a whole; (b) a material impairment of the prospect of repayment of any portion of the Obligations, or (c) a material adverse effect on the Collateral.

"Material Agreement" is any license, agreement or other similar contractual arrangement with a Person or Governmental Authority whereby Borrower or any of its Subsidiaries is reasonably likely to be required to transfer, either in-kind or in cash, prior to the Maturity Date, assets or property valued (book or market) at more than Fifty Thousand Dollars (\$50,000.00) in the aggregate or any license, agreement or other similar contractual arrangement conveying rights in or to any material Intellectual Property.

"Maturity Date" is 32 weeks from the Effective Date.

"Maximum Legal Rate" shall mean the maximum nonusurious interest rate, if any, that at any time or from time to time may be contracted for, taken, reserved, charged or received on the indebtedness evidenced by the Note and as provided for herein or the other Loan Documents, under the laws of such state or states whose laws are held by any court of competent jurisdiction to govern the interest rate provisions of the Term Loan.

"Obligations" are all of Borrower's obligations to pay when due any debts, principal, interest, the Prepayment Fee, the Final Fee, and other amounts Borrower owes the Lenders now or later, in connection with, related to, following, or arising from, out of or under, this Agreement or, the other Loan Documents, or otherwise, and including interest accruing after Insolvency Proceedings begin (whether or not allowed) and debts, liabilities, or obligations of Borrower assigned to the Lenders and/or Collateral Agent, and the performance of Borrower's duties under the Loan Documents.

"Operating Documents" are, for any Person, such Person's formation documents, as certified by the Secretary of State (or equivalent agency) of such Person's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Effective Date, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.

"Perfection Certificate" is that certain form attached hereto as Exhibit B-1.

"Permitted Indebtedness" is: (a) Borrower's Indebtedness to the Lenders and Collateral Agent under this Agreement and the other Loan Documents; (b) Indebtedness existing on the Effective Date and disclosed on the Perfection Certificate(s); (c) unsecured Indebtedness to trade creditors and Indebtedness in connection with credit cards incurred in the ordinary course of business; (d) up to \$3,000,000 in the form of an unsecured loan through one or more third parties introduced by SternAegis based on ongoing discussions; (e) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (d) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose materially more burdensome terms upon Borrower, or its Subsidiary, as the case may be.

"Permitted Investments" are: (a) investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business; (b) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this paragraph (b) shall not apply to Investments of Borrower in any Subsidiary; and (c) shares of common stock of Infinite Reality, Inc., a Delaware corporation, received pursuant to the equity exchange agreement executed by and between Borrower and Infinite Reality, Inc. on September 30, 2024 and amended and restated on October 29, 2024.

"Permitted Licenses" are licenses of over-the-counter software that is commercially available to the public. 'Permitted Liens' are Liens existing on the Effective Date, including SLR Digital Finance, LLC, and disclosed on the Perfection Certificates or arising under this Agreement and the other Loan Documents;

"Person" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.

"Property" means any interest in any kind of property or asset, whether real, personal or mixed, and whether tangible or intangible.

"Pro Rata Share" is, as of any date of determination, with respect to each Lender, a percentage (expressed as a decimal, rounded to the ninth decimal place) determined by dividing the outstanding principal amount of the Term Loan held by such Lender by the aggregate outstanding principal amount of the Term Loan.

"Related Persons" means, with respect to any Person, each Affiliate of such Person and each director, officer, employee, agent, trustee, representative, attorney, accountant and each insurance, environmental, legal, financial and other advisor and other consultants and agents of or to such Person or any of its Affiliates.

"Required Lenders" means (i) for so long as the Lead Lender has not assigned or transferred any of its interests in the Term Loan, Lenders holding one hundred percent (100%) of the aggregate outstanding principal balance of the Term Loan, or (ii) at any time from and after the Lead Lender has assigned or transferred any interest in its Term Loan, Lenders holding at least fifty one percent (51%) of the aggregate outstanding principal balance of the Term Loan.

"Responsible Officer" is any of the President, Chief Executive Officer, or Chief Financial Officer of Borrower or Parent.

"Secured Promissory Note" is defined in Section 2.5.

"Shares" means one hundred percent (100.0%) of the stock, units or other evidence of equity ownership held by Borrower or its Subsidiaries of any Subsidiary which is organized under the laws of the United States.

"Solvent" is, with respect to any Person: the fair salable value of such Person's consolidated assets (including goodwill minus disposition costs) exceeds the fair value of such Person's liabilities; such Person is not left with unreasonably small capital after the transactions in this Agreement; and such Person is able to pay its debts (including trade debts) as they mature in the ordinary course (without taking into account any forbearance and extensions related thereto).

"Subordinated Debt" is indebtedness incurred by Borrower or any of its Subsidiaries subordinated to all Indebtedness of Borrower and/or its Subsidiaries to the Lenders (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Collateral Agent and the Lenders entered into between Collateral Agent, Borrower, and/or any of its Subsidiaries, and the other creditor), on terms acceptable to Collateral Agent and the Lenders.

"Subsidiary" is, with respect to any Person, any Person of which more than fifty percent (50%) of the voting stock or other equity interests (in the case of Persons other than corporations) is owned or controlled, directly or indirectly, by such Person or through one or more intermediaries. Unless otherwise specified, references herein to a Subsidiary means a Subsidiary of Borrower.

"Term Loan" is defined in Section 2.2(a) hereof.

"Term Loan Amortization Schedule" means the amortization schedule set forth in Exhibit B-4 of this Agreement.

[Balance of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by one of its officers thereunto duly authorized on the date hereof.

<u>Parties</u>	Name of Signatory and Title	-	Signature
B <u>orrowers</u> SUPER LEAGUE ENTERPRISE, INC	CLAYTON JAMES HAYNES, CFO		s/Clayton Haynes
INPVP, LLC	CLAYTON JAMES HAYNES, CFO	<u>s</u>	s/ Clayton Haynes
Guarantors			
SUPER LEAGUE ENTERPRISE, INC	CLAYTON JAMES HAYNES, CFO	_	s/ Clayton Haynes
INPVP, LLC	CLAYTON JAMES HAYNES, CFO	<u> </u>	s/ Clayton Haynes
LEAD LENDER:		COLLATERAL AGENT	
Agile Lending, LLC		Agile Capital Funding, LLC	
/s/ Aaron Greenblott		/s/ Aaron Greenblott	_
By: Aaron Greenblott		By: Aaron Greenblott	
Its: Member		Its: Member	

EXHIBITS TO FOLLOW

APPENDIX 1

BORROWER LIST

The Borrower is presently indebted to its account receivable factoring provider (SLR Digital Finance, LLC, or "SLR") in the approximate amount of \$140,000, which amount is paid down upon receipt of receivables by SLE that have been financed by SLR. SLE and SLR utilize a lockbox arrangement for effectuation. The Borrower and SLR commenced the relationship on December 5, 2023 upon the execution of a Financing and Security Agreement ("Agreement"). SLR holds a security interest in the assets of Borrower evidenced by the Agreement as well as a financing statement filed on Form UCC-1.

EXHIBIT A

DESCRIPTION OF COLLATERAL

The Collateral consists of all of Borrower's right, title and interest in and to the following property:

All of Borrower's goods, Accounts, Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles (including Intellectual Property), commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts and other Collateral Accounts, all certificates of deposit, fixtures, letters of credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and

All of Borrower's books and records relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing.

Notwithstanding the foregoing, the Collateral does not include (i) any license or contract, in each case if the granting of a Lien in such license or contract is prohibited by or would constitute a default under the agreement governing such license or contract (but (A) only to the extent such prohibition is enforceable under applicable law and (B) other than to the extent that any such term would be rendered ineffective pursuant to Sections 9-406, 9-408 or 9-409 (or any other Section) of Division 9 of the Code); provided that upon the termination, lapsing or expiration of any such prohibition, such license or contract, as applicable, shall automatically be subject to the security interest granted in favor of Collateral Agent hereunder and become part of the "Collateral."

Title	Case Number	Atty Ref.	Country	Application Number		Publication Number	Publication Date	Patent Number	Issue Date	Expiration Date	Status	Owner	Inventors	Chain of Title	Next Action Item
MULTI- PLAYE R GAMING ON PUBLIC DISPLAYS	094920- 598454	SLG-001P	US	62/075,357	05-Nov- 2014					05-Nov- 2015	Expired	Super League Gaming, Inc.	John Steigelf est, David	1. NVS to Nth Games, INC. 2. Nth Games, Inc. to Super League Game, Inc.	
GAM E SYST EM	094920- 598455	SLG- 001PCT	WO	PCT/US2015/ 0 29532	06-May- 2015	WO/2016/07 30 35	12-May- 2016				Expired		Miller, John Steigelf est, David		
MULTI- USER GAME SYSTEM WITH TRIGGER- BAS ED GENERATI ON OF PROJECTIO N VIEW	094920- 598456	SLG-001	US	15/179,868	10-Jun- 2016	US 2016- 0279511 A1	29-Sep- 2016	10,946,274	2021	10-Jun- 2036	Granted	Super League Gaming, Inc.	John Steigelf est, David	Gaming, Inc	1st MF due 3/16/2024
MULTI- USER GAME SYSTEM WITH CHARACTE R- BASED GENERATI ON OF PROJECTIO N VIEW (TRACK 1)		SLG- 001COA	US	15/179,878	10-Jun- 2016	US 2016- 0279509 A1	29-Sep- 2016	10,702,771	2020	10-Jun- 2036	Granted	Super League Gaming, Inc.	John Steigelf est, David	Gaming, Inc	1st MF due 1/7/2024
MULTI- USER GAME SYSTEM WITH CHARACTE R- BASED GENERATI ON OF PROJECTIO	094920- 659100	SLG- 001COB	US	16/922,877	07-Jul- 2020	US 2020- 0330862 A1	22-Oct- 2020	11,534,683	27-Dec- 2022	07-Jul- 2040	Granted	Super League Gaming, Inc.	John Steigelf est,	INVS to Super League Gaming, Inc	1st MF due 12/27/2025

N VIEW	I	I	I								I	I	David	I	
CLOUD-	094920-	SLG-002P	US	62/702,900	24-Jul-					24-Jul-	Expired	Super	Mueller,	INVS to	
BASE D	598458				2018					2019		League	Aaron	Super	
GAME												Gaming,		League	
STREAMI												Inc.	David	Gaming,	
NG CLOUD-	094920-	SLG-002	US	16/250,843	17-Jan-	US	30-Jan-	11,260,295	01 Man	30-Oct-	Granted	C	Mueller,	Inc INVS to	1st MF due
BASE D	602519	SLG-002	US	10/230,843		2020-	2020	11,260,293	2022	2039	Granted	Super League	Aaron		3/1/2025
GAME	002319					0030697	2020		2022	2039		Gaming,		League	3/1/2023
STREAMI						A1						Inc.	David	Gaming,	
NG												1110.	Davia	Inc	
CLOUD-	094920-	SLG-	WO	PCT/US2019/	22-Jul-	WO	30-Jan-				Expired		Mueller,	No	
BASE D	631747	002PCT		0	2019	2020/023403	2020				1		Aaron	assignment	
GAME				42859									Steigelfest,		
STREAMI													David		
NG															
CLOUD-	094920-	SLG-	US	17/169,110		US	03-Jun-	11,794,102		17-Jan-	Granted	Super	Mueller,	INVS to	1st MF
BASE D	680234	002DVA				2021-	2021		2023	2039		League	Aaron		due 10/24/2026
GAME						0162300						Gaming,	Steigelfest,		
STREAMI						A1						Inc.	David	Gaming,	
NG INTELLIGE	004020	SLG-003	US	17/188,913	01-Mar-	US 2022-	01.0	11,794,119	24-Oct-	01-Mar-	Granted	G	Steigelfest,	Inc INVS to	1st MF due
NT ELLIGE	675002	SLG-003	US	1//188,913		0S 2022- 0274026	01-Sep- 2022	11,/94,119	24-061-	2041	Granted	Super League	David	Super	10/24/2026
PRIORITIZ	073002				2021	A1	2022		2023	2041		Gaming,	Kmbikyan,		10/24/2020
ATI ON						All						Inc.	Arthur	Gaming,	
AND												me.	Gramo,	Inc	
MANIPULA													Brian		
TIO N OF															
STREAM															
VIEWS															
(SLG-004)	094920-	SLG-004	US	17/217,863		US 2022-	06-Oct-	11,601,698	07-Mar-	30-Mar-	Granted	Super	Gramo,	INVS to	1st MF due
INTELLIG	675004	1			2021	0321933	2022		2023	2041		League	Brian	Super	3/7/2026
ENT						A1						Gaming,	Kmbikyan,		
SYNCHRO												Inc.	Arthur	Gaming,	
NIZ ATION													Steigelfest,	Inc	
OF MEDIA													David		
STREAMS]		

Country/Jur is diction	Mark	Design/Device	App. No./Reg. No.	Registration Date	Owner/Company	Status
US Federal	ATLANTA BASHERS SLG and	9	App. No.:87856855 Reg. No.:5814502	July 23, 2019	Super League Gaming, Inc.	Registered
US Federal	BOSTON REVOLT SLG and Design		App. No.:87517749 Reg. No.:6646544	February 15, 2022	Super League Gaming, Inc.	Registered
US Federal	CHICAGO FORCE SLG and Design		App. No.:87190038 Reg. No.:5758074	May 21, 2019	Super League Gaming, Inc.	Registered
US Federal	D.C. CONSPIRAT O RS SLG and Design	TORSPIRATORS	App. No.:87693083 Reg. No.:5752306	May 14, 2019	Super League Gaming, Inc.	Registered
US Federal	DALLAS DYNAMITE SLG and Design		App. No.:87190016 Reg. No.:5906038	November 12, 2019	Super League Gaming, Inc.	Registered
US Federal	DENVER DRAKES SLG and Design		App. No.:87517764 Reg. No.:5751989	May 14, 2019	Super League Gaming, Inc.	Registered
US Federal	DETROIT VENGEANC E		App. No.:88830879 Reg. No.:6349538	May 11, 2021	Super League Gaming, Inc.	Registered
US Federal	DETROIT VENGEANC E SLG and	2	App. No.:88841767 Reg. No.:6349605	May 11, 2021	Super League Gaming, Inc.	Registered

US Federal	FRAMERAT E	App. No.:88498010 Reg. No.:5960768	January 14, 2020	Super League Gaming, Inc.	Registered
US Federal	FRAMERAT E and Design	App. No.:88498007 Reg. No.:5960766	January 14, 2020	Super League Gaming, Inc.	Registered
US Federal	HOUSTON BLAST SLG and Design	App. No.:87517777 Reg. No.:5764256	May 28, 2019	Super League Gaming, Inc.	Registered
US Federal	LA LOS ANGELES SHOCKWA VE S SLG and Design	App. No.:87189727 Reg. No.:6205951	November 24, 2020	Super League Gaming, Inc.	Registered

EXHIBIT B-1

PERFECTION CERTIFICATE

[INTENTIONALLY OMITTED]

EXHIBIT B-2

DISBURSEMENT INSTRUCTION FORM

The proceeds of the first advance of Term Loan shall be disbursed as follows:

Term Loan \$2,500,000.00

Less:

Administrative Agent Fee to be remitted to <u>Agile Capital Funding, LLC</u> (\$125,000.00)

SUPER LEAGUE ENTERPRISE, INC. - 434125 - Balance {\$1,501,142.84}

TOTAL TERM LOAN NET PROCEEDS TO BORROWER \$873,857.16

The aggregate net proceeds of the Term Loan shall be transferred to the Designated Deposit Account as follows:

BORROWER: SUPER LEAGUE ENTERPRISE, INC.,

Account Name: Super League Enterprise, Inc.

Bank Name: Wells Fargo Bank

ABA Number: 121000248 Account Number: 3360458826

The proceeds of the subsequent advances of the Term Loan shall be disbursed as follows:

EXHIBIT B-3

DRAWDOWN SCHEDULE

Within 2 Business Days of Closing Date.

EXHIBIT B-4 REPAYMENT AND AMORTIZATION SCHEDULE

Projected Payment Schedule
Weekly Payment
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$110,937.50
\$3,550,000.00

EXHIBIT B-5

Business Loan and Security Agreement Supplement

Principal Amount of Loan:	\$2,500,000.00, including the Administrative Agent Fee, available as set forth in the Drawdown Schedule found in
	Exhibit B-3 of this Agreement.
Total Repayment Amount:	The total repayment amount of the Term Loan, including all interest, lender fees, and third-party fees, assuming all
	payments are made on time is
	\$3,550,000.00.
Payment Schedule:	As set forth in the Repayment and Amortization Schedule found in Exhibit B-4 of the Agreement.
Payment Multiplier: (The per	1.42
dollar cost of the loan inclusive of all interest and	
fees).	
Interest Charge:	\$1,050,000.00, assuming all payments are made on time.
Fees payable to Collateral Agent and its	Administrative Agent Fee: \$125,000.00, payable at closing out of proceeds of the Term Loan
designees:	

EXHIBIT B-6

AUTHORIZATION AGREEMENT FOR AUTOMATED CLEARING HOUSE TRANSACTIONS

[INTENTIONALLY OMITTED]

EXHIBIT D

CONFESSED JUDGMENT SECURED PROMISSORY NOTE

IMPORTANT NOTICE: THIS INSTRUMENT CONTAINS A CONFESSION OF JUDGMENT PROVISION WHICH CONSTITUTES A WAIVER OF IMPORTANT RIGHTS YOU MAY HAVE AS A DEBTOR AND ALLOWS THE CREDITOR TO OBTAIN A JUDGMENT AGAINST YOU WITHOUT ANY FURTHER NOTICE.

CONFESSED JUDGMENT SECURED PROMISSORY NOTE

\$2,500,000.00 Dated: February 07, 2025

FOR VALUE RECEIVED, the undersigned SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation (**Parent**" or "**Borrower**") and its subsidiaries, INPVP, LLC, A Domestic Delaware Corporation, individually and collectively, jointly and severally, "**Guarantors**"), HEREBY JOINTLY AND SEVERALLY PROMISE TO PAY to the order of Agile Lending, LLC, or its designees or assigns ("**Lead Lender**") the principal amount of TWO MILLION FIVE HUNDRED THOUSAND DOLLARS (\$2,500,000.00) or such lesser amount as shall equal the outstanding principal balance of the Term Loan made to Borrower by Lender, plus interest on the aggregate unpaid principal amount of such Term Loan, at the rates and in accordance with the terms of the Business Loan and Security Agreement dated February 07, 2025, by and among Borrower, Lender, Collateral Agent, and the other Lenders from time to time party thereto (as amended, restated, supplemented or otherwise modified from time to time, the "**Loan Agreement**"). If not sooner paid, the entire principal amount and all accrued and unpaid interest hereunder shall be due and payable on the Maturity Date as set forth in the Loan Agreement. Any capitalized term not otherwise defined herein shall have the meaning attributed to such term in the Loan Agreement.

Principal, interest and all other amounts due with respect to the Term Loan, are payable in lawful money of the United States of America to Lender as set forth in the Loan Agreement and this Confessed Judgment Secured Promissory Note (this "Note").

The Loan Agreement, among other things, (a) provides for the making of a secured Term Loan by Lender to Borrower, and (b) contains provisions for acceleration of the maturity hereof upon the happening of certain stated events.

This Note may not be prepaid except as set forth in Section 2.2 (c) and Section 2.2(d) of the Loan Agreement.

This Note and the obligation of Borrower to repay the unpaid principal amount of the Term Loan, interest on the Term Loan and all other amounts due Lender under the Loan Agreement is secured as provided under the Loan Agreement.

Presentment for payment, demand, notice of protest and all other demands and notices of any kind in connection with the execution, delivery, performance and enforcement of this Note are hereby waived.

Borrower shall pay all reasonable fees and expenses, including, without limitation, reasonable attorneys' fees and costs, incurred by Lender in the enforcement or attempt to enforce any of Borrower's obligations hereunder not performed when due.

This Note shall be governed by, and construed and interpreted in accordance with, the internal laws of the Commonwealth of Virginia.

The ownership of an interest in this Note shall be registered on a record of ownership maintained by Lender or its agent. Notwithstanding anything else in this Note to the contrary, the right to the principal of, and stated interest on, this Note may be transferred only if the transfer is registered on such record of ownership and the transferee is identified as the owner of an interest in the obligation. Borrower shall be entitled to treat the registered holder of this Note (as recorded on such record of ownership) as the owner in fact thereof for all purposes and shall not be bound to recognize any equitable or other claim to or interest in this Note on the part of any other person or entity.

BORROWER HEREBY KNOWINGLY, VOLUNTARILY AND INTELLIGENTLY WAIVES ANY AND ALL RIGHTS THAT EACH PARTY TO THIS NOTE MAY NOW OR HEREAFTER HAVE UNDER THE LAWS OF THE UNITED STATES OF AMERICA OR THE COMMONWEALTH OF VIRGINIA, TO A TRIAL BY JURY OF ANY AND ALL ISSUES ARISING DIRECTLY OR INDIRECTLY IN ANY ACTION OR PROCEEDING RELATING TO THIS NOTE, THE LOAN DOCUMENTS OR ANY TRANSACTIONS CONTEMPLATED THEREBY OR RELATED THERETO. IT IS INTENDED THAT THIS WAIVER SHALL APPLY TO ANY AND ALL DEFENSES, RIGHTS, CLAIMS AND/OR COUNTERCLAIMS IN ANY SUCH ACTION OR PROCEEDING. BORROWER EXPRESSLY ACKNOWLEDGES THAT BORROWER IS SUBJECT TO PERSONAL JURISDICTION IN THE COMMONWEALTH OF THE VIRGINIA, THAT BORROWER INTENTIONALLY ENTERED INTO THE TRANSACTIONS THAT ARE THE SUBJECT OF THIS CONFESSED JUDGMENT PROMISSORY NOTE WITH LENDER, WHO IS LOCATED IN THE COMMONWEALTH OF VIRGINIA, AND THAT BORROWER WAIVES ANY AND ALL OBJECTIONS TO THE EXERCISE OF PERSONAL JURISDICTION OVER BORROWER OF THE COMMONWEALTH OF VIRGINIA AND TO VENUE IN THE CIRCUIT COURT FOR ARLINGTON COUNTY, VIRGINIA AND ANY OTHER COURT WITHIN THE COMMONWEALTH OF VIRGINIA

BORROWER UNDERSTANDS THAT THIS WAIVER IS A WAIVER OF A CONSTITUTIONAL SAFEGUARD, AND EACH PARTY INDIVIDUALLY BELIEVES THAT THERE ARE SUFFICIENT ALTERNATE PROCEDURAL AND SUBSTANTIVE SAFEGUARDS, INCLUDING, A TRIAL BY AN IMPARTIAL JUDGE, THAT ADEQUATELY OFFSET THE WAIVER CONTAINED HEREIN.

UPON THE OCCURRENCE OF AN EVENT OF DEFAULT HEREUNDER OR UNDER THE LOAN AGREEMENT, LEAD LENDER MAY CONFESS JUDGMENT AGAINST BORROWER AS PROVIDED HEREIN. UPON THE OCCURRENCE OF ANY EVENT OF DEFAULT HEREUNDER, BORROWER HEREBY AUTHORIZES AND EMPOWERS THE CLERK OF ANY COURT OF RECORD IN THE COMMONWEALTH OF VIRGINIA, INCLUDING BUT NOT LIMITED TO THE CLERK OF THE CIRCUIT COURT FOR THE COUNTY OF ARLINGTON TO ENTER JUDGMENT BY CONFESSION AGAINST BORROWER IN FAVOR OF LEAD LENDER FOR THE FULL AMOUNT DUE AND PAYABLE UNDER THE FINANCING AGREEMENTS AND SECURED BY THE LOAN AGREEMENT, TOGETHER WITH ALL PERMITTED FEES AND INTEREST, AS EVIDENCED BY AN AFFIDAVIT SIGNED BY AN OFFICER OF LEAD LENDER SETTING FORTH THE AMOUNT THEN DUE, TOGETHER WITH REASONABLE ATTORNEYS' FEES AND COLLECTION COSTS INCURRED BY LEAD LENDER AS PROVIDED IN THIS INSTRUMENT, TO THE EXTENT PERMITTED BY LAW, EXPRESSLY WAIVING SUMMONS AND OTHER PROCESS, AND DOES HEREBY CONSENT TO THE IMMEDIATE EXECUTION OF SUCH JUDGMENT, EXPRESSLY WAIVING THE BENEFIT OF ALL EXEMPTION OR HOMESTEAD LAWS.

BORROWER HEREBY CONSTITUTES AND APPOINTS JODIE E. BUCHMAN, ESQ., PIERCE C. MURPHY, ESQ., OF SILVERMAN, THOMPSON, SLUTKIN & WHITE, 400 E PRATT ST, SUITE 900, BALTIMORE, MD, 21202, OR A DULY APPOINTED SUBSTITUTE AS THE TRUE AND LAWFUL ATTORNEY-IN-FACT FOR BORROWER AND ALL PERSONS CLAIMING THROUGH OR UNDER BORROWER TO SIGN AN AGREEMENT FOR ENTERING IN ANY COMPETENT COURT AN AMICABLE ACTION IN EJECTMENT FOR POSSESSION OF THE COLLATERAL AND/OR TO APPEAR IN THE CLERK'S OFFICE OF THE CIRCUIT COURT OF ARLINGTON COUNTY, VIRGINIA, OR ANY COURT OF COMPETENT JURISDICTION AND TO CONFESS JUDGMENT AGAINST BORROWER, AND ALL PERSONS CLAIMING UNDER OR THROUGH BORROWER IN FAVOR OF LEAD LENDER, FOR WHICH THIS NOTE, OR A COPY THEREOF VERIFIED BY AFFIDAVIT, SHALL BE SUFFICIENT WARRANT; WHEREUPON A WRIT OF POSSESSION MAY IMMEDIATELY ISSUE FOR POSSESSION OF THE COLLATERAL, WITHOUT ANY PRIOR WRIT OR PROCEEDING WHATSOEVER AND WITHOUT ANY STAY OF EXECUTION. LEAD LENDER MAY BRING AN AMICABLE ACTION IN EJECTMENT AND/OR CONFESS JUDGMENT THEREIN EITHER BEFORE OR AFTER THE INSTITUTION OF PROCEEDINGS TO ENFORCE THIS NOTE AND/OR AFTER ENTRY OF JUDGMENT ON THIS NOTE, OR AFTER A PUBLIC SALE OF THE COLLATERAL IN WHICH LEAD LENDER IS THE SUCCESSFUL BIDDER.

BORROWER HEREBY RATIFIES AND CONFIRMS ALL THAT SAID ATTORNEY OR ATTORNEYS MAY DO PURSUANT TO THE FOREGOING POWER. PURSUANT TO SECTION 8.01-435 OF THE CODE OF VIRGINIA OF 1950, AS AMENDED, BORROWER IS HEREBY NOTIFIED THAT A SUBSTITUTE ATTORNEY-IN-FACT UNDER THIS PARAGRAPH MAY BE APPOINTED BY THE LEAD LENDER, OBLIGEE, OR PERSON OTHERWISE ENTITLED TO PAYMENT UNDER THIS AGREEMENT BY RECORDING AN INSTRUMENT NAMING SUCH SUBSTITUTE ATTORNEY-IN-FACT IN THE CLERK'S OFFICE WHERE JUDGMENT IS TO BE CONFESSED.

THE FOREGOING AUTHORIZATION TO PURSUE PROCEEDINGS FOR CONFESSING JUDGMENT AND ANY AND ALL JUDGMENT ENFORCEMENT MEASURES THAT LEAD LENDER OPTS TO PURSUE, INCLUDING BUT NOT LIMITED TO OBTAINING POSSESSION OF THE COLLATERAL, AND IS AN ESSENTIAL PART OF LEAD LENDER'S REMEDIES FOR ENFORCEMENT OF THIS NOTE AND THE LOAN AGREEMENT AND SHALL SURVIVE ANY ENFORCEMENT ACTIONS OR FORECLOSURE SALE BY OR TO LEAD LENDER.

[Signature Page to Follow]

IN WITNESS WHEREOF, Borrower caused this Note to be duly executed under seal by one of its officers thereunto duly authorized on the date hereof.
BORROWER:
By: CLAYTON JAMES HAYNES Date: February 07, 2025
STATE: COUNTY OF:
I hereby certify that on February 7, 2025, before me, the undersigned, Notary Public in and for the State of California, at large, personally appeare CLAYTON JAMES HAYNES, individually and as the CFO of SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation (*Parent*) and its subsidiaries, INPVP, LLC, A Domestic Delaware Corporation known to me or satisfactorily proven to be the person whose name is subscribed to the foregoing instrument and acknowledged that he executed the foregoing on behalf of himself individually, SUPER LEAGUE ENTERPRISE, INC., A Domestic Delaware Corporation ("Parent") and its subsidiaries, INPVP, LLC, A Domestic Delaware Corporation for the purposes set forth therein.
(Seal) Notary Public
My Commission Expires: Registration Number:

EXHIBIT E

PREPAYMENT AMENDMENT

Upon the prepayment of any principal amount, Borrower shall be obligated to pay a Prepayment Fee comprising make- whole premium payment on account of such principal so paid, which Prepayment Fee shall be equal to the aggregate and actual amount of interest (at the contract rate of interest) that would be paid through the Maturity Date

 Calendar Days After Funding
 Payoff Amount

 30 Days
 \$3,125,000.00

 45 Days
 \$3,250,000.00

 60 Days
 \$3,375,000.00

EQUITY PURCHASE AGREEMENT

This equity purchase agreement is entered into as of February 14, 2025 (this "Agreement"), by and between Super League Enterprise, Inc., a Delaware corporation (the "Company"), and Hudson Global Ventures, LLC, a Nevada limited liability company (the "Investor", and collectively with the Company, the "Parties").

WHEREAS, the Parties desire that, upon the terms and subject to the conditions contained herein, the Company shall issue and sell to the Investor, from time to time as provided herein, and the Investor shall purchase up to Three Million One Hundred Thousand Dollars (\$3,100,000.00) of the Company's Common Stock (as defined below);

NOW, THEREFORE, the Parties hereto agree as follows:

ARTICLE I CERTAIN DEFINITIONS

- Section 1.1 <u>DEFINED TERMS</u>. As used in this Agreement, the following terms shall have the following meanings specified or indicated (such meanings to be equally applicable to both the singular and plural forms of the terms defined):
 - "Agreement" shall have the meaning specified in the preamble hereof.
- "Average Daily Trading Value" shall mean the average trading volume of the Company's Common Stock on the Principal Market during the three (3) Trading Days immediately preceding the respective Put Date multiplied by the lowest closing price of the Company's Common Stock on the Principal Market during the three (3) Trading Days immediately preceding the respective Put Date.
 - "Bankruptcy Law" means Title 11, U.S. Code, or any similar federal or state law for the relief of debtors.
 - "Claim Notice" shall have the meaning specified in Section 9.3(a).
- "Clearing Costs" shall mean all fees incurred by the Investor with respect to the Put Shares, including but not limited to fees charged by or paid to any brokerage firm (including commissions), any clearing firm, and Transfer Agent fees, as well as attorney fees of \$1,000 per Put.
 - "Clearing Date" shall be the date on which the Investor receives the Put Shares in its brokerage account.
 - "Closing" shall mean one of the closings of a purchase and sale of shares of Common Stock pursuant to Section 2.3.

"Closing Certificate" shall mean the closing certificate of the Company in the form of Exhibit B hereto.

"Closing Date" shall mean the date of any Closing hereunder.

"Commitment Period" shall mean the period commencing on the Execution Date, and ending on the earlier of (i) the date on which the Investor shall have purchased Put Shares pursuant to this Agreement equal to the Maximum Commitment Amount, (ii) twenty-four (24) months after the date of this Agreement, (iii) written notice of termination by the Company to the Investor (which shall not occur at any time that the Investor holds any of the Put Shares), (iv) the Registration Statement is no longer effective after the initial effective date of the Registration Statement, or (v) the date that, pursuant to or within the meaning of any Bankruptcy Law, the Company commences a voluntary case or any Person commences a proceeding against the Company, a Custodian is appointed for the Company or for all or substantially all of its property or the Company makes a general assignment for the benefit of its creditors; provided, however, that the provisions of Articles III, IV, V, VI, IX and the agreements and covenants of the Company and the Investor set forth in Article X shall survive the termination of this Agreement.

"Commitment Shares" shall mean 300,000 shares of Common Stock.

"Common Stock" shall mean the Company's common shares, \$0.001 par value per share, and any shares of any other class of common stock whether now or hereafter authorized, having the right to participate in the distribution of dividends (as and when declared) and assets (upon liquidation of the Company).

"Common Stock Equivalents" means any securities of the Company or the Subsidiaries which would entitle the holder thereof to acquire at any time Common Stock, including, without limitation, any debt, preferred stock, right, option, warrant or other instrument that is at any time convertible into or exercisable or exchangeable for, or otherwise entitles the holder thereof to receive, Common Stock.

"Company" shall have the meaning specified in the preamble to this Agreement.

"Custodian" means any receiver, trustee, assignee, liquidator or similar official under any Bankruptcy Law.

"<u>Damages</u>" shall mean any loss, claim, damage, liability, cost and expense (including, without limitation, reasonable attorneys' fees and disbursements and costs and expenses of expert witnesses and investigation).

"Dispute Period" shall have the meaning specified in Section 9.3(a).

"DTC" shall mean The Depository Trust Company, or any successor performing substantially the same function for the Company.

- "DTC/FAST Program" shall mean the DTC's Fast Automated Securities Transfer Program.
- "DWAC" shall mean Deposit Withdrawal at Custodian as defined by the DTC.
- "DWAC Eligible" shall mean that (a) the Common Stock is eligible at DTC for full services pursuant to DTC's Operational Arrangements, including, without limitation, transfer through DTC's DWAC system, (b) the Company has been approved (without revocation) by the DTC's underwriting department, (c) the Transfer Agent is approved as an agent in the DTC/FAST Program, (d) the Put Shares are otherwise eligible for delivery via DWAC, and (e) the Transfer Agent does not have a policy prohibiting or limiting delivery of the Put Shares, as applicable, via DWAC.
- "DWAC Shares" means shares of Common Stock that are (i) issued in electronic form, (ii) freely tradable and transferable and without restriction on resale and (iii) timely credited by the Company to the Investor's or its designee's specified DWAC account with DTC under the DTC/FAST Program, or any similar program hereafter adopted by DTC performing substantially the same function.
 - "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
 - "Execution Date" shall mean the date of this Agreement.
 - "FINRA" shall mean the Financial Industry Regulatory Authority, Inc.
 - "Investment Amount" shall mean the Put Shares referenced in the Put Notice multiplied by the Purchase Price, minus the Clearing Costs.
 - "Indemnified Party" shall have the meaning specified in Section 9.2.
 - "Indemnifying Party" shall have the meaning specified in Section 9.2.
 - "Indemnity Notice" shall have the meaning specified in Section 9.3(e).
- "Initial Purchase Price" shall mean 92% of the closing price of the Company's Common Stock on the Principal Market on the Trading Day immediately preceding the respective Put Date.
 - "Investor" shall have the meaning specified in the preamble to this Agreement.
 - "Lien" means a lien, charge, pledge, security interest, encumbrance, right of first refusal, preemptive right or other restriction.
- "Market Price" shall mean 92% of the lowest closing price of the Company's Common Stock on the Principal Market on any Trading Day during the Valuation Period, as reported by Quotestream or other reputable source designated by the Investor, subject to adjustment as provided in this Agreement.

- "Material Adverse Effect" shall mean any effect on the business, operations, properties, or financial condition of the Company and the Subsidiaries that is material and adverse to the Company and the Subsidiaries and/or any condition, circumstance, or situation that would prohibit or otherwise materially interfere with the ability of the Company to enter into and perform its obligations under any Transaction Document.
 - "Maximum Commitment Amount" shall mean Three Million One Hundred Thousand Dollars (\$3,100,000.00), inclusive of the Commitment Shares.
- "Person" shall mean an individual, a corporation, a partnership, an association, a trust or other entity or organization, including a government or political subdivision or an agency or instrumentality thereof.
- "Principal Market" shall mean any of the national exchanges (i.e. NYSE, NYSE American, and Nasdaq) which is at the time the principal trading platform for the Common Stock (excluding all OTC marketplaces).
- "Purchase Price" shall mean the lesser of the (i) Initial Purchase Price or (ii) Market Price on such date on which the Purchase Price is calculated in accordance with the terms and conditions of this Agreement.
 - "Put" shall mean the right of the Company to require the Investor to purchase shares of Common Stock, subject to the terms and conditions of this Agreement.
 - "Put Date" shall mean any Trading Day during the Commitment Period that a Put Notice is deemed delivered pursuant to Section 2.2(b).
- "Put Notice" shall mean a written notice, substantially in the form of Exhibit A hereto, to Investor setting forth the Put Shares which the Company intends to require Investor to purchase pursuant to the terms of this Agreement.
- "Put Shares" shall mean all shares of Common Stock issued, or that the Company shall be entitled to issue, per any applicable Put Notice in accordance with the terms and conditions of this Agreement.
- "Registration Rights Agreement" shall mean that certain registration rights agreement entered into by the Company with the Investor on the date hereof in connection with this Agreement.
 - "Registration Statement" shall have the meaning specified in Section 6.4.
 - "Regulation D" shall mean Regulation D promulgated under the Securities Act.

- "Required Minimum" shall mean, as of any date, the maximum aggregate number of shares of Common Stock potentially issuable at such time pursuant to the Transaction Documents, which shall be calculated on each such date as follows: the then remaining Maximum Commitment Amount divided by the Initial Purchase Price on each such date, ignoring any beneficial ownership limitations set forth herein.
 - "Rule 144" shall mean Rule 144 under the Securities Act or any similar provision then in force under the Securities Act.
 - "SEC" shall mean the United States Securities and Exchange Commission.
 - "SEC Documents" shall have the meaning specified in Section 4.5.
 - "Securities" means, collectively, the Put Shares and Commitment Shares.
 - "Securities Act" shall mean the Securities Act of 1933, as amended.
- "Shareholder Approval" shall mean the approval of a sufficient amount of holders of the Company's Common Stock to satisfy the shareholder approval requirements for such action as provided in Nasdaq Rule 5635(d), to effectuate the transactions contemplated by this Agreement, including but not limited to the issuance of Common Stock under this Agreement, including but not limited to the Put Shares and Commitment Shares, in excess of 3,319,323 shares of Common Stock (the "Exchange Cap"), subject to appropriate adjustment for any stock dividend, stock split, stock combination, rights offerings, reclassification or similar transaction that proportionately decreases or increases the Common Stock.
- "Subsidiary" means any Person the Company wholly-owns or controls, or in which the Company, directly or indirectly, owns a majority of the voting stock or similar voting interest, in each case that would be disclosable pursuant to Item 601(b)(21) of Regulation S-K promulgated under the Securities Act.
 - "Third Party Claim" shall have the meaning specified in Section 9.3(a).
 - "Trading Day" shall mean a day on which the Principal Market shall be open for business.
 - "Transaction Documents" shall mean this Agreement, the Registration Rights Agreement, and all exhibits hereto and thereto.
- "Transfer Agent" shall mean Direct Transfer, LLC, the current transfer agent of the Company, with a mailing address of One Glenwood Avenue, Suite 1001, Raleigh, NC 27603, and any successor transfer agent of the Company.
- "Transfer Agent Instruction Letter" means the letter from the Company to the Transfer Agent entered into on or around the date of this Agreement that instructs the Transfer Agent to issue the Put Shares and Commitment Shares pursuant to the Transaction Documents.

"Valuation Period" shall mean the period beginning on the Put Date and continuing through the date that is three (3) Trading Days immediately following the Clearing Date associated with the applicable Put Notice.

ARTICLE II PURCHASE AND SALE OF COMMON STOCK

Section 2.1 <u>PUTS</u>. Subject to the terms and conditions set forth herein (including, without limitation, the provisions of Article VII), the Company shall have the right, but not the obligation, to direct the Investor, by its delivery to the Investor of a Put Notice from time to time, to purchase Put Shares (i) in a minimum amount not less than \$25,000.00 (calculated using the Initial Purchase Price) and (ii) in a maximum amount up to the lesser of (a) \$2,500,000.00 (calculated using the Initial Purchase Price) or (b) 200% of the Average Daily Trading Value.

Section 2.2 MECHANICS.

- (a) <u>PUT NOTICE</u>. At any time and from time to time during the Commitment Period, except as provided in this Agreement, the Company may deliver a Put Notice to Investor, subject to satisfaction of the conditions set forth in Section 7.2 and otherwise provided herein. The initial price per share identified in the respective Put Notice shall be equal to the Initial Purchase Price and shall be used for purposes of determining the number of shares of Common Stock that the Company can issue pursuant to a respective Put Notice in accordance with Section 2.1 of this Agreement. At the end of the Valuation Period, the Purchase Price for the respective Put Shares and Investment Amount shall be established as further provided in this Agreement. The Company shall deliver, or cause to be delivered, the Put Shares as DWAC Shares to the Investor on or before 4:30 p.m. Eastern time, on the Put Date.
- DATE OF DELIVERY OF PUT NOTICE. A Put Notice shall be deemed delivered on (i) the Trading Day it is received by email by the Investor if such notice is received on or prior to 2:30 p.m. Eastern time, or (ii) the immediately succeeding Trading Day if it is received by email after 2:30 p.m. Eastern time on a Trading Day or at any time on a day which is not a Trading Day. The Company shall not deliver a Put Notice to the Investor during the period beginning on the Put Date of the immediately prior Put Notice and continuing through the date that is three (3) Trading Days following the Clearing Date associated with the immediately prior Put Notice (the "Cooldown Period"), provided, however, that the respective Cooldown Period all not apply to the immediately prior Put Notice if (i) the Put Shares for the immediately prior Put Notice have been delivered to the Investor pursuant to the terms of this Agreement and (ii) the trading volume of the Common Shares on any Trading Day during the respective Cooldown Period exceeds 300% of the total Put Shares of the immediately prior Put Notice (the "Cooldown Waiver Trigger"). Notwithstanding anything herein to the contrary, all trading volume of the Common Shares on the respective Put Date that occurs prior to the specific time that the Put Notice is delivered to Investor shall not count towards the Cooldown Waiver Trigger.

- Section 2.3 <u>CLOSINGS</u>. At the end of the Valuation Period, the Purchase Price and Investment Amount for the respective Put Shares shall be established as provided in this Agreement. If the value of the Put Shares delivered to the Investor causes the Company to exceed the Maximum Commitment Amount, then immediately after the Valuation Period the Investor shall return to the Company the surplus amount of Put Shares associated with such Put and the Purchase Price with respect to such Put shall be reduced by any Clearing Costs related to the return of such Put Shares. The Closing of a Put shall occur within two (2) Trading Days following the end of the respective Valuation Period, whereby the Investor shall deliver the Investment Amount by wire transfer of immediately available funds to an account designated by the Company.
- Section 2.4 <u>PRINCIPAL MARKET REGULATION</u>. The Company shall not effect any issuances or sales of the Put Shares under this Agreement above the Exchange Cap and the Investor shall not have the obligation to purchase Put Shares under this Agreement above the Exchange Cap until the Shareholder Approval has been obtained by the Company and is in effect.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF INVESTOR

The Investor represents and warrants to the Company that:

- Section 3.1 <u>INTENT</u>. The Investor is entering into this Agreement for its own account and the Investor has no present arrangement (whether or not legally binding) at any time to sell the Securities to or through any Person in violation of the Securities Act or any applicable state securities laws; <u>provided</u>, <u>however</u>, that the Investor reserves the right to dispose of the Securities at any time in accordance with federal and state securities laws applicable to such disposition.
- Section 3.2 NO LEGAL ADVICE FROM THE COMPANY. The Investor acknowledges that it has had the opportunity to review this Agreement and the transactions contemplated by this Agreement with its own legal counsel and investment and tax advisors. The Investor is relying solely on such counsel and advisors and not on any statements or representations of the Company or any of its representatives or agents for legal, tax or investment advice with respect to this investment, the transactions contemplated by this Agreement or the securities laws of any jurisdiction.
- Section 3.3 <u>ACCREDITED INVESTOR.</u> The Investor is an accredited investor as defined in Rule 501(a)(3) of Regulation D, and the Investor has such experience in business and financial matters that it is capable of evaluating the merits and risks of an investment in the Securities. The Investor acknowledges that an investment in the Securities is speculative and involves a high degree of risk.

- Section 3.4 <u>AUTHORITY</u>. The Investor has the requisite power and authority to enter into and perform its obligations under this Agreement and the other Transaction Documents and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the other Transaction Documents and the consummation by it of the transactions contemplated hereby and thereby have been duly authorized by all necessary action and no further consent or authorization of the Investor is required. Each Transaction Document to which it is a party has been duly executed by the Investor, and when delivered by the Investor in accordance with the terms hereof, will constitute the valid and binding obligation of the Investor enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application.
- Section 3.5 NOT AN AFFILIATE. The Investor is not an officer, director or "affiliate" (as that term is defined in Rule 405 of the Securities Act) of the Company.
- Section 3.6 <u>ORGANIZATION AND STANDING.</u> The Investor is an entity duly incorporated or formed, validly existing and in good standing under the laws of the jurisdiction of its incorporation or formation with full right, corporate, partnership, limited liability company or similar power and authority to enter into and to consummate the transactions contemplated by this Agreement and the other Transaction Documents.
- Section 3.7 <u>ABSENCE OF CONFLICTS</u>. The execution and delivery of this Agreement and the other Transaction Documents, and the consummation of the transactions contemplated hereby and thereby and compliance with the requirements hereof and thereof, will not (a) violate any law, rule, regulation, order, writ, judgment, injunction, decree or award binding on the Investor, (b) violate any provision of any indenture, instrument or agreement to which the Investor is a party or is subject, or by which the Investor or any of its assets is bound, or conflict with or constitute a material default thereunder, (c) result in the creation or imposition of any lien pursuant to the terms of any such indenture, instrument or agreement, or constitute a breach of any fiduciary duty owed by the Investor to any third party, or (d) require the approval of any third-party (that has not been obtained) pursuant to any material contract, instrument, agreement, relationship or legal obligation to which the Investor is subject or to which any of its assets, operations or management may be subject.
- Section 3.8 <u>DISCLOSURE; ACCESS TO INFORMATION</u>. The Investor had an opportunity to review copies of the SEC Documents filed on behalf of the Company and has had access to all publicly available information with respect to the Company.
- Section 3.9 MANNER OF SALE. At no time was the Investor presented with or solicited by or through any leaflet, public promotional meeting, television advertisement or any other form of general solicitation or advertising.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to the Investor that:

- Section 4.1 <u>ORGANIZATION OF THE COMPANY</u>. The Company and each of the Subsidiaries is an entity duly incorporated or otherwise organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation or organization, with the requisite power and authority to own and use its properties and assets and to carry on its business as currently conducted. Neither the Company nor any Subsidiary is in violation nor default of any of the provisions of its respective certificate or articles of incorporation, bylaws or other organizational or charter documents. Each of the Company and the Subsidiaries is duly qualified to conduct business and is in good standing as a foreign corporation or other entity in each jurisdiction in which the nature of the business conducted or property owned by it makes such qualification necessary, except where the failure to be so qualified or in good standing, as the case may be, could not have or reasonably be expected to result in a Material Adverse Effect and no proceeding has been instituted in any such jurisdiction revoking, limiting or curtailing or seeking to revoke, limit or curtail such power and authority or qualification.
- Section 4.2 <u>AUTHORITY</u>. The Company has the requisite corporate power and authority to enter into and perform its obligations under this Agreement and the other Transaction Documents. The execution and delivery of this Agreement and the other Transaction Documents by the Company and the consummation by it of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action and no further consent or authorization of the Company or its Board of Directors or stockholders is required. Each of this Agreement and the other Transaction Documents has been duly executed and delivered by the Company and constitutes a valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, or similar laws relating to, or affecting generally the enforcement of, creditors' rights and remedies or by other equitable principles of general application.
- Section 4.3 <u>CAPITALIZATION</u>. Except as set forth in the SEC Documents, the Company has not issued any capital stock since its most recently filed periodic report under the Exchange Act, other than pursuant to the exercise of employee stock options under the Company's stock option plans, the issuance of shares of Common Stock to employees pursuant to the Company's employee stock purchase plans and pursuant to the conversion and/or exercise of Common Stock Equivalents outstanding as of the date of the most recently filed periodic report under the Exchange Act. No Person has any right of first refusal, preemptive right, right of participation, or any similar right to participate in the transactions contemplated by the Transaction Documents. Except as set forth in the SEC Documents and except as a result of the purchase and sale of the Securities, there are no outstanding options, warrants, scrip rights to subscribe to, calls or commitments of any character whatsoever relating to, or securities, rights or obligations convertible into or exercisable or exchangeable for, or giving any Person any right to subscribe for or acquire any shares of Common Stock, or contracts, commitments, understandings or arrangements by which the Company or any Subsidiary is or may become bound to issue additional shares of Common Stock or Common Stock Equivalents. Except as disclosed in the SEC Documents, the issuance and sale of the Securities will not obligate the Company to issue shares of Common Stock or other securities to any Person (other than the Investor) and will not result in a right of any holder of Company securities to adjust the exercise, conversion, exchange or reset price under any of such securities. There are no stockholders agreements, voting agreements or other similar agreements with respect to the Company's capital stock to which the Company is a party or, to the knowledge of the Company, between or among any of the Company's stockholders.

Section 4.4 <u>LISTING AND MAINTENANCE REQUIREMENTS</u>. The Common Stock is registered pursuant to Section 12(b) or 12(g) of the Exchange Act, and the Company has taken no action designed to, or which to its knowledge is likely to have the effect of, terminating the registration of the Common Stock under the Exchange Act nor has the Company received any notification that the SEC is contemplating terminating such registration. Except as set forth in the SEC Documents, the Company has not, in the twelve (12) months preceding the date hereof, received notice from the Principal Market on which the Common Stock is or has been listed or quoted to the effect that the Company is not in compliance with the listing or maintenance requirements of such Principal Market. The Company is, and has no reason to believe that it will not in the foreseeable future continue to be, in compliance with all such listing and maintenance requirements.

SEC DOCUMENTS; DISCLOSURE. Notwithstanding the late filing of the Company's annual report on Form 10-K for the fiscal year ended December 31, 2023, which report was filed within the fifteen (15) day extension period afforded under the Exchange Act, the Company has filed all reports, schedules, forms, statements and other documents required to be filed by the Company under the Securities Act and the Exchange Act, including pursuant to Section 13(a) or 15(d) thereof, for the one (1) year preceding the date hereof (or such shorter period as the Company was required by law or regulation to file such material) (the foregoing materials, including the exhibits thereto and documents incorporated by reference therein, being collectively referred to herein as the "SEC Documents") on a timely basis or has received a valid extension of such time of filing and has filed any such SEC Documents prior to the expiration of any such extension. As of their respective dates, the SEC Documents complied in all material respects with the requirements of the Securities Act and the Exchange Act, as applicable, and other federal laws, rules and regulations applicable to such SEC Documents, and none of the SEC Documents when filed contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The financial statements of the Company included in the SEC Documents comply as to form and substance in all material respects with applicable accounting requirements and the published rules and regulations of the SEC or other applicable rules and regulations with respect thereto. Such financial statements have been prepared in accordance with generally accepted accounting principles applied on a consistent basis during the periods involved (except (a) as may be otherwise indicated in such financial statements or the notes thereto or (b) in the case of unaudited interim statements, to the extent they may not include footnotes or may be condensed or summary statements) and fairly present in all material respects the financial position of the Company as of the dates thereof and the results of operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal, immaterial, year-end audit adjustments). Except with respect to the material terms and conditions of the transactions contemplated by the Transaction Documents, the Company confirms that neither it nor any other Person acting on its behalf has provided the Investor or its agents or counsel with any information that it believes constitutes or might constitute material, non-public information. The Company understands and confirms that the Investor will rely on the foregoing representation in effecting transactions in securities of the Company.

- Section 4.6 <u>VALID ISSUANCES</u>. The Securities are duly authorized and, when issued and paid for in accordance with the applicable Transaction Documents, will be duly and validly issued, fully paid, and non-assessable, free and clear of all Liens imposed by the Company other than restrictions on transfer provided for in the Transaction Documents.
- NO CONFLICTS. The execution, delivery and performance of this Agreement and the other Transaction Documents by the Company and the consummation by the Company of the transactions contemplated hereby and thereby, including, without limitation, the issuance of the Securities, do not and will not: (a) result in a violation of the Company's or any Subsidiary's certificate or articles of incorporation, by-laws or other organizational or charter documents, (b) conflict with, or constitute a material default (or an event that with notice or lapse of time or both would become a material default) under, result in the creation of any Lien upon any of the properties or assets of the Company or any Subsidiary, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture, instrument or any "lock-up" or similar provision of any underwriting or similar agreement to which the Company or any Subsidiary is a party, or (c) result in a violation of any federal, state or local law, rule, regulation, order, judgment or decree (including federal and state securities laws and regulations) applicable to the Company or any Subsidiary or by which any property or asset of the Company or any Subsidiary is bound or affected (except for such conflicts, defaults, terminations, amendments, accelerations, cancellations and violations as would not, individually or in the aggregate, have a Material Adverse Effect) nor is the Company otherwise in violation of, conflict with or in default under any of the foregoing. The business of the Company is not being conducted in violation of any law, ordinance or regulation of any governmental entity, except for possible violations that either singly or in the aggregate do not and will not have a Material Adverse Effect. The Company is not required under federal, state or local law, rule or regulation to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency in order for it to execute, deliver or perform any of its obligations under this Agreement or the other Transaction Documents (other than any SEC, FINRA or state securities filings that may be required to be made by the Company subsequent to any Closing or any registration statement that may be filed pursuant hereto); provided that, for purposes of the representation made in this sentence, the Company is assuming and relying upon the accuracy of the relevant representations and agreements of Investor herein.
- Section 4.8 NO MATERIAL ADVERSE CHANGE. No event has occurred that would have a Material Adverse Effect on the Company that has not been disclosed in subsequent SEC Documents.
- Section 4.9 <u>LITIGATION AND OTHER PROCEEDINGS</u>. Except as disclosed in the SEC Documents, there are no actions, suits, investigations, inquiries or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company, any Subsidiary or any of their respective properties, nor has the Company received any written or oral notice of any such action, suit, proceeding, inquiry or investigation, which would have a Material Adverse Effect. No judgment, order, writ, injunction or decree or award has been issued by or, to the knowledge of the Company, requested of any court, arbitrator or governmental agency which would have a Material Adverse Effect. There has not been, and to the knowledge of the Company, there is not pending or contemplated, any investigation by the SEC involving the Company, any Subsidiary or any current or former director or officer of the Company or any Subsidiary.

- Section 4.10 <u>REGISTRATION RIGHTS</u>. Except as set forth in the SEC Documents and as granted to Investor, no Person (other than the Investor) has any right to cause the Company to effect the registration under the Securities Act of any securities of the Company or any Subsidiary.
- Section 4.11 NO SOLICITATION. The Company represents and warrants that neither the Investor nor its employee(s), member(s), beneficial owner(s), or partner(s) solicited the Company to enter into this Agreement and consummate the transactions described in this Agreement. The Company represents and warrants that the Investor is not required to be registered as a broker-dealer under the Securities Exchange Act of 1934 in order to (i) enter into or consummate the transactions encompassed by the Transaction Documents, (ii) fulfill the Investor's obligations under the Transaction Documents, or (iii) exercise any of the Investor's rights under the Transaction Documents (including but not limited to the sale of the Securities).

ARTICLE V COVENANTS OF INVESTOR

Section 5.1 <u>COMPLIANCE WITH LAW; TRADING IN SECURITIES.</u> The Investor's trading activities with respect to shares of Common Stock will be in compliance with all applicable state and federal securities laws and regulations and the rules and regulations of FINRA and the Principal Market.

ARTICLE VI COVENANTS OF THE COMPANY

- Section 6.1 <u>RESERVATION OF COMMON STOCK.</u> The Company shall maintain a reserve from its duly authorized shares of Common Stock equal to the Required Minimum in accordance with the terms of this Agreement.
- Section 6.2 <u>LISTING OF COMMON STOCK</u>. The Company shall promptly secure the listing of all of the Securities to be issued to the Investor hereunder on the Principal Market (subject to official notice of issuance) and shall maintain the listing of all such Securities from time to time issuable hereunder. The Company shall maintain the listing and trading of the Common Stock on the Principal Market (including, without limitation, maintaining sufficient net tangible assets) and will comply in all respects with the Company's reporting, filing and other obligations under the bylaws or rules of FINRA and the Principal Market.

- Section 6.3 OTHER EQUITY LINES AND TRANSACTIONS. So long as this Agreement remains in effect, the Company covenants and agrees that it will not, without the prior written consent of the Investor, enter into any other Equity Line of Credit (as defined below) or Variable Rate Transaction (as defined below) with any other party. "Equity Line of Credit" shall mean any transaction involving a written agreement between the Company and an investor or underwriter whereby the Company has the right to "put" its securities to the investor or underwriter over an agreed period of time and at an agreed price or price formula. "Variable Rate Transaction" means a transaction in which the Company (i) issues or sells any debt or equity securities that are convertible into, exchangeable or exercisable for, or include the right to receive, additional shares of Common Stock either (A) at a conversion price, exercise price or exchange rate or other price that is based upon, and/or varies with, the trading prices of or quotations for the shares of Common Stock at any time after the initial issuance of such debt or equity securities or (B) with a conversion, exercise or exchange price that is subject to being reset at some future date after the initial issuance of such debt or equity security or upon the occurrence of specified or contingent events directly or indirectly related to the business of the Company or the market for the Common Stock or (ii) issues securities at a future determined price.
- Section 6.4 FILING OF CURRENT REPORT AND REGISTRATION STATEMENT. The Company agrees that it shall file a Current Report on Form 8-K, including the Transaction Documents as exhibits thereto, with the SEC within the time required by the Exchange Act, relating to the transactions contemplated by, and describing the material terms and conditions of, the Transaction Documents (the "Current Report"). The Company shall permit the Investor to review and comment upon the final pre-filing draft version of the Current Report at least one (1) Trading Day prior to its filing with the SEC, and the Company shall give reasonable consideration to all such comments. The Investor shall use its reasonable best efforts to comment upon the final pre-filing draft version of the Current Report within one (1) Trading Day from the date the Investor receives it from the Company. The Company shall also comply with the Registration Rights Agreement with respect to the filing and effectiveness deadlines of a new registration statement (the "Registration Statement") in accordance with the terms of such Registration Rights Agreement.
- Section 6.5 NO BROKER-DEALER ACKNOWLEDGEMENT. Absent a final adjudication from a court of competent jurisdiction stating otherwise, the Company shall not to any person, institution, governmental or other entity, state, claim, allege, or in any way assert, that Investor is currently, or ever has been, a broker-dealer under the Securities Exchange Act of 1934.

ARTICLE VII CONDITIONS TO DELIVERY OF PUT NOTICES AND CONDITIONS TO CLOSING

- Section 7.1 <u>CONDITIONS PRECEDENT TO THE RIGHT OF THE COMPANY TO ISSUE AND SELL PUT SHARES.</u> In addition to the other provisions of this Agreement, the right of the Company to issue and sell the Put Shares to the Investor is subject to the satisfaction of each of the conditions set forth below:
- (a) <u>ACCURACY OF INVESTOR'S REPRESENTATIONS AND WARRANTIES.</u> The representations and warranties of the Investor shall be true and correct in all material respects as of the date of this Agreement and as of the date of each Closing as though made at each such time.

- (b) <u>PERFORMANCE BY INVESTOR</u>. Investor shall have performed, satisfied and complied in all respects with all covenants, agreements and conditions required by this Agreement to be performed, satisfied or complied with by the Investor at or prior to such Closing.
- Section 7.2 <u>CONDITIONS PRECEDENT TO THE OBLIGATION OF INVESTOR TO PURCHASE PUT SHARES</u>. The obligation of the Investor hereunder to purchase Put Shares is subject to the satisfaction of each of the following conditions:
- (a) <u>EFFECTIVE REGISTRATION STATEMENT</u>. The Registration Statement, and any amendment or supplement thereto, shall remain effective for the resale by the Investor of the Put Shares and Commitment Shares at prevailing market prices (and not fixed prices) and (i) neither the Company nor the Investor shall have received notice that the SEC has issued or intends to issue a stop order with respect to such Registration Statement or that the SEC otherwise has suspended or withdrawn the effectiveness of such Registration Statement, either temporarily or permanently, or intends or has threatened to do so and (ii) no other suspension of the use of, or withdrawal of the effectiveness of, such Registration Statement or related prospectus shall exist.
- (b) <u>ACCURACY OF THE COMPANY'S REPRESENTATIONS AND WARRANTIES</u>. The representations and warranties of the Company shall be true and correct in all material respects as of the date of this Agreement and as of the date of each Closing (except for representations and warranties specifically made as of a particular date).
- (c) <u>PERFORMANCE BY THE COMPANY</u>. The Company shall have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by this Agreement to be performed, satisfied or complied with by the Company, including but not limited to the delivery of the Put Shares as provided in Section 2.2(a) of this Agreement.
- (d) <u>NO INJUNCTION.</u> No statute, rule, regulation, executive order, decree, ruling or injunction shall have been enacted, entered, promulgated or adopted by any court or governmental authority of competent jurisdiction that prohibits or directly and materially adversely affects any of the transactions contemplated by the Transaction Documents, and no proceeding shall have been commenced that may have the effect of prohibiting or materially adversely affecting any of the transactions contemplated by the Transaction Documents.
- (e) <u>ADVERSE CHANGES</u>. Since the date of filing of the Company's most recent SEC Document, no event that had or is reasonably likely to have a Material Adverse Effect has occurred.
- (f) NO SUSPENSION OF TRADING IN OR DELISTING OF COMMON STOCK. The trading of the Common Stock shall not have been suspended by the SEC, the Principal Market or FINRA, or otherwise halted for any reason, and the Common Stock shall have been approved for listing on and shall not have been delisted from the Principal Market. In the event of a suspension, delisting, or halting for any reason, of the trading of the Common Stock, as contemplated by this Section 7.2(f), the Investor shall have the right to return to the Company any remaining amount of Put Shares associated with such Put, and the Purchase Price with respect to such Put shall be reduced accordingly.

- BENEFICIAL OWNERSHIP LIMITATION. The number of Put Shares then to be purchased by the Investor shall not exceed the number of such shares that, when aggregated with all other shares of Common Stock then owned by the Investor beneficially or deemed beneficially owned by the Investor, would result in the Investor owning more than the Beneficial Ownership Limitation (as defined below), as determined in accordance with Section 16 of the Exchange Act and the regulations promulgated thereunder. For purposes of this Section 7.2(g), in the event that the amount of Common Stock outstanding, as determined in accordance with Section 16 of the Exchange Act and the regulations promulgated thereunder, is greater on a Closing Date than on the date upon which the Put Notice associated with such Closing Date is given, the amount of Common Stock outstanding on such Closing Date shall govern for purposes of determining whether the Investor, when aggregating all purchases of Common Stock made pursuant to this Agreement, would own more than the Beneficial Ownership Limitation following such Closing Date. The "Beneficial Ownership Limitation" shall be 4.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable pursuant to a Put Notice.
- (h) <u>PENNY STOCK</u>. The Common Stock shall not be deemed to be a "penny stock" as defined in SEC Rule 240.3a51-1 (17 CFR § 240.3a51-1). In the event that the Common Stock becomes a "penny stock" prior to the closing of a respective Put, the Investor shall have the right to return to the Company up to all of the Put Shares associated with such Put, and the Purchase Price with respect to such Put shall be reduced accordingly.
- (i) NO KNOWLEDGE. The Company shall have no knowledge of any event more likely than not to have the effect of causing the Registration Statement to be suspended or otherwise ineffective (which event is more likely than not to occur within the fifteen (15) Trading Days following the Trading Day on which such Put Notice is deemed delivered).
- (j) NO VIOLATION OF SHAREHOLDER APPROVAL REQUIREMENT. The issuance of the Put Shares shall not violate the shareholder approval requirements of the Principal Market, including but not limited to as contemplated by Section 2.4 of this Agreement.
- (k) OFFICER'S CERTIFICATE. On the date of delivery of each Put Notice, the Investor shall have received the Closing Certificate executed by an executive officer of the Company and to the effect that all the conditions to such Closing shall have been satisfied as of the date of each such certificate.
 - (l) <u>DWAC ELIGIBLE</u>. The Common Stock must be DWAC Eligible and not subject to a "DTC chill."
- (m) <u>SEC DOCUMENTS</u>. All reports, schedules, registrations, forms, statements, information and other documents required to have been filed by the Company with the SEC pursuant to the reporting requirements of the Exchange Act shall have been filed with the SEC within the applicable time periods prescribed for such filings under the Exchange Act.

- (n) <u>RESERVE</u>. The Company shall have reserved the Required Minimum for the Investor's benefit under this Agreement, the Company shall have satisfied the reserve requirements with respect to all other contracts between the Company and Investor, and the Transfer Agent Instruction Letter shall have been executed by the Company and the Transfer Agent as well as acknowledged and agreed to in writing by the Transfer Agent.
- (o) MINIMUM PRICING. The lowest traded price of the Common Stock in the ten (10) Trading Days immediately preceding the respective Put Date must exceed \$0.15 per share.
- (p) <u>BANKRUPTCY</u>. Bankruptcy, insolvency, reorganization or liquidation proceedings or other proceedings, voluntary or involuntary, for relief under any bankruptcy law or any law for the relief of debtors shall not be instituted by or against the Company or any subsidiary of the Company (the "Bankruptcy Proceedings"), and the Company shall have no knowledge of any event more likely than not to have the effect of causing Bankruptcy Proceedings to arise. In the event of Bankruptcy Proceedings as contemplated by this Section 7.2(p), the Investor shall have the right to return to the Company any remaining amount of Put Shares associated with such Put, and the Purchase Price with respect to such Put shall be reduced accordingly.

ARTICLE VIII LEGENDS

- Section 8.1 NO RESTRICTIVE STOCK LEGEND. No restrictive stock legend shall be placed on the share certificates representing the Put Shares.
- Section 8.2 <u>INVESTOR'S COMPLIANCE</u>. Nothing in this Article VIII shall affect in any way the Investor's obligations hereunder to comply with all applicable securities laws upon the sale of the Common Stock.

ARTICLE IX NOTICES; INDEMNIFICATION

Section 9.1 NOTICES. All notices, demands, requests, consents, approvals, and other communications required or permitted hereunder shall be in writing and, unless otherwise specified herein, shall be (a) personally served, (b) deposited in the mail, registered or certified, return receipt requested, postage prepaid, (c) delivered by reputable air courier service with charges prepaid, or (d) transmitted by hand delivery, telegram, or email as a PDF, addressed as set forth below or to such other address as such party shall have specified most recently by written notice given in accordance herewith. Any notice or other communication required or permitted to be given hereunder shall be deemed effective (i) upon hand delivery or delivery by email at the address designated below (if delivered on a business day during normal business hours where such notice is to be received), or the first business day following such delivery (if delivered other than on a business day during normal business hours where such notice is to be received) or (ii) on the second business day following the date of mailing by express courier service or on the fifth business day after deposited in the mail, in each case, fully prepaid, addressed to such address, or upon actual receipt of such mailing, whichever shall first occur.

The addresses for such communications shall be:

If to the Company:

Super League Enterprise, Inc. 2856 Colorado Avenue Santa Monica, CA 90404 Email: ann.hand@superleague.com Attention: Ann Hand

If to the Investor:

Hudson Global Ventures, LLC

Email: info@hudsonventuresllc.com

Either party hereto may from time to time change its address or email for notices under this Section 9.1 by giving at least ten (10) days' prior written notice of such changed address to the other party hereto.

INDEMNIFICATION. Each party (an "Indemnifying Party") agrees to indemnify and hold harmless the other party along with its officers, Section directors, employees, and authorized agents, and each Person or entity, if any, who controls such party within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act (an "Indemnified Party") from and against any Damages, joint or several, and any action in respect thereof to which the Indemnified Party becomes subject to, resulting from, arising out of or relating to (i) any misrepresentation, breach of warranty or nonfulfillment of or failure to perform any covenant or agreement on the part of the Indemnifying Party contained in this Agreement, (ii) any untrue statement or alleged untrue statement of a material fact contained in the Registration Statement or any post-effective amendment thereof or supplement thereto, or the omission or alleged omission therefrom of a material fact required to be stated therein or necessary to make the statements therein not misleading, (iii) any untrue statement or alleged untrue statement of a material fact contained in any preliminary prospectus or contained in the final prospectus (as amended or supplemented, if the Company files any amendment thereof or supplement thereto with the SEC) or the omission or alleged omission to state therein any material fact necessary to make the statements made therein, in the light of the circumstances under which the statements therein were made, not misleading, or (iv) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any state securities law or any rule or regulation under the Securities Act, the Exchange Act or any state securities law, as such Damages are incurred, except to the extent such Damages result primarily from the Indemnified Party's failure to perform any covenant or agreement contained in this Agreement or the Indemnified Party's negligence, recklessness or bad faith in performing its obligations under this Agreement; provided, however, that the foregoing indemnity agreement shall not apply to any Damages of an Indemnified Party to the extent, but only to the extent, arising out of or based upon any untrue statement or alleged untrue statement or omission or alleged omission made by an Indemnifying Party in reliance upon and in conformity with written information furnished to the Indemnifying Party by the Indemnified Party expressly for use in the Registration Statement, any post-effective amendment thereof or supplement thereto, or any preliminary prospectus or final prospectus (as amended or supplemented).

- Section 9.3 <u>METHOD OF ASSERTING INDEMNIFICATION CLAIMS</u>. All claims for indemnification by any Indemnified Party under Section 9.2 shall be asserted and resolved as follows:
- In the event any claim or demand in respect of which an Indemnified Party might seek indemnity under Section 9.2 is asserted against or sought to be collected from such Indemnified Party by a Person other than a party hereto or an affiliate thereof (a "Third Party Claim"), the Indemnified Party shall deliver a written notification, enclosing a copy of all papers served, if any, and specifying the nature of and basis for such Third Party Claim and for the Indemnified Party's claim for indemnification that is being asserted under any provision of Section 9.2 against an Indemnifying Party, together with the amount or, if not then reasonably ascertainable, the estimated amount, determined in good faith, of such Third Party Claim (a "Claim Notice") with reasonable promptness to the Indemnifying Party. If the Indemnified Party fails to provide the Claim Notice with reasonable promptness after the Indemnified Party receives notice of such Third Party Claim, the Indemnifying Party shall not be obligated to indemnify the Indemnified Party with respect to such Third Party Claim to the extent that the Indemnifying Party's ability to defend has been prejudiced by such failure of the Indemnified Party. The Indemnifying Party shall notify the Indemnified Party as soon as practicable within the period ending thirty (30) calendar days following receipt by the Indemnifying Party of either a Claim Notice or an Indemnity Notice (as defined below) (the "Dispute Period") whether the Indemnifying Party disputes its liability or the amount of its liability to the Indemnified Party under Section 9.2 and whether the Indemnifying Party desires, at its sole cost and expense, to defend the Indemnified Party against such Third Party Claim.
- (i) If the Indemnifying Party notifies the Indemnified Party within the Dispute Period that the Indemnifying Party desires to defend the Indemnified Party with respect to the Third Party Claim pursuant to this Section 9.3(a), then the Indemnifying Party shall have the right to defend, with counsel reasonably satisfactory to the Indemnified Party, at the sole cost and expense of the Indemnifying Party, such Third Party Claim by all appropriate proceedings, which proceedings shall be vigorously and diligently prosecuted by the Indemnifying Party to a final conclusion or will be settled at the discretion of the Indemnifying Party (but only with the consent of the Indemnified Party in the case of any settlement that provides for any relief other than the payment of monetary damages or that provides for the payment of monetary damages as to which the Indemnified Party shall not be indemnified in full pursuant to Section 9.2). The Indemnifying Party shall have full control of such defense and proceedings, including any compromise or settlement thereof; provided, however, that the Indemnified Party may, at the sole cost and expense of the Indemnified Party, at any time prior to the Indemnifying Party's delivery of the notice referred to in the first sentence of this clause (i), file any motion, answer or other pleadings or take any other action that the Indemnified Party reasonably believes to be necessary or appropriate to protect its interests; and provided, further, that if requested by the Indemnifying Party, the Indemnified Party will, at the sole cost and expense of the Indemnifying Party, provide reasonable cooperation to the Indemnifying Party in contesting any Third Party Claim that the Indemnifying Party elects to contest. The Indemnified Party may participate in, but not control, any defense or settlement of any Third Party Claim controlled by the Indemnifying Party pursuant to this clause (i), and except as provided in the preceding sentence, the Indemnified Party shall bear its own costs and expenses with

(ii) If the Indemnifying Party fails to notify the Indemnified Party within the Dispute Period that the Indemnifying Party desires to defend
Third Party Claim pursuant to Section 9.3(a), or if the Indemnifying Party gives such notice but fails to prosecute vigorously and diligently or settle the Third Party Cla
or if the Indemnifying Party fails to give any notice whatsoever within the Dispute Period, then the Indemnified Party shall have the right to defend, at the sole cost
expense of the Indemnifying Party, the Third Party Claim by all appropriate proceedings, which proceedings shall be prosecuted by the Indemnified Party in a reasona
manner and in good faith or will be settled at the discretion of the Indemnified Party(with the consent of the Indemnifying Party, which consent will not be unreasonal
withheld). The Indemnified Party will have full control of such defense and proceedings, including any compromise or settlement thereof; provided, however, that
requested by the Indemnified Party, the Indemnifying Party will, at the sole cost and expense of the Indemnifying Party, provide reasonable cooperation to the Indemnif
Party and its counsel in contesting any Third Party Claim which the Indemnified Party is contesting. Notwithstanding the foregoing provisions of this clause (ii), if
Indemnifying Party has notified the Indemnified Party within the Dispute Period that the Indemnifying Party disputes its liability or the amount of its liability hereunde
the Indemnified Party with respect to such Third Party Claim and if such dispute is resolved in favor of the Indemnifying Party in the manner provided in clause (iii) belong the Indemnified Party with respect to such Third Party Claim and if such dispute is resolved in favor of the Indemnifying Party in the manner provided in clause (iii) belong the Indemnified Party with respect to such Third Party Claim and if such dispute is resolved in favor of the Indemnifying Party in the manner provided in clause (iii) belong the Indemnified Party with respect to such Third Party Claim and if such dispute is resolved in favor of the Indemnifying Party in the manner provided in clause (iii) belong the Indemnified Party with respect to such Third Party Claim and if such dispute is resolved in favor of the Indemnifying Party in the manner provided in clause (iii) belong the Indemnified Party with respect to such Third Party Claim and Indemnified Party with respect to the Indemnifi
the Indemnifying Party will not be required to bear the costs and expenses of the Indemnified Party's defense pursuant to this clause (ii) or of the Indemnifying Part
participation therein at the Indemnified Party's request, and the Indemnified Party shall reimburse the Indemnifying Party in full for all reasonable costs and expension
incurred by the Indemnifying Party in connection with such litigation. The Indemnifying Party may participate in, but not control, any defense or settlement controlled by
Indemnified Party pursuant to this clause (ii), and the Indemnifying Party shall bear its own costs and expenses with respect to such participation.

(iii) If the Indemnifying Party notifies the Indemnified Party that it does not dispute its liability or the amount of its liability to the Indemnified Party with respect to the Third Party Claim under Section 9.2 or fails to notify the Indemnified Party within the Dispute Period whether the Indemnifying Party disputes its liability or the amount of its liability to the Indemnified Party with respect to such Third Party Claim, the amount of Damages specified in the Claim Notice shall be conclusively deemed a liability of the Indemnifying Party under Section 9.2 and the Indemnifying Party shall pay the amount of such Damages to the Indemnified Party on demand. If the Indemnifying Party has timely disputed its liability or the amount of its liability with respect to such claim, the Indemnifying Party and the Indemnified Party shall proceed in good faith to negotiate a resolution of such dispute; provided, however, that if the dispute is not resolved within thirty (30) days after the Claim Notice, the Indemnifying Party shall be entitled to institute such legal action as it deems appropriate.

- (b) In the event any Indemnified Party should have a claim under Section 9.2 against the Indemnifying Party that does not involve a Third Party Claim, the Indemnified Party shall deliver a written notification of a claim for indemnity under Section 9.2 specifying the nature of and basis for such claim, together with the amount or, if not then reasonably ascertainable, the estimated amount, determined in good faith, of such claim (an "Indemnity Notice") with reasonable promptness to the Indemnifying Party. The failure by any Indemnified Party to give the Indemnify Notice shall not impair such party's rights hereunder except to the extent that the Indemnifying Party demonstrates that it has been irreparably prejudiced thereby. If the Indemnifying Party notifies the Indemnified Party that it does not dispute the claim or the amount of the claim described in such Indemnity Notice or fails to notify the Indemnified Party within the Dispute Period whether the Indemnifying Party disputes the claim or the amount of the claim described in such Indemnity Notice, the amount of Damages specified in the Indemnity Notice will be conclusively deemed a liability of the Indemnifying Party under Section 9.2 and the Indemnifying Party shall pay the amount of such Damages to the Indemnified Party on demand. If the Indemnifying Party has timely disputed its liability or the amount of its liability with respect to such claim, the Indemnifying Party and the Indemnified Party shall proceed in good faith to negotiate a resolution of such dispute; provided, however, that if the dispute is not resolved within thirty (30) days after the Claim Notice, the Indemnifying Party shall be entitled to institute such legal action as it deems appropriate.
- (c) The Indemnifying Party agrees to pay the Indemnified Party, promptly as such expenses are incurred and are due and payable, for any reasonable legal fees or other reasonable expenses incurred by them in connection with investigating or defending any such Claim.
- (d) The indemnity provisions contained herein shall be in addition to (i) any cause of action or similar rights of the Indemnified Party against the Indemnifying Party or others, and (ii) any liabilities the Indemnifying Party may be subject to.

ARTICLE X MISCELLANEOUS

ARBITRATION OF CLAIMS; GOVERNING LAW; JURISDICTION. The Company and Investor shall submit all Claims (as defined in Exhibit C of this Agreement) (the "Claims") arising under this Agreement or any other agreement between the Company and Investor or their respective affiliates (including but not limited to the Transaction Documents) or any Claim relating to the relationship of the Company and Investor or their respective affiliates to binding arbitration pursuant to the arbitration provisions set forth in Exhibit C of the Agreement (the "Arbitration Provisions"). The Company and Investor hereby acknowledge and agree that the Arbitration Provisions are unconditionally binding on the Company and Investor hereto and are severable from all other provisions of this Agreement. By executing this Agreement, Company represents, warrants and covenants that Company has reviewed the Arbitration Provisions carefully, consulted with legal counsel about such provisions (or waived its right to do so), understands that the Arbitration Provisions are intended to allow for the expeditious and efficient resolution of any dispute hereunder, agrees to the terms and limitations set forth in the Arbitration Provisions, and that Company will not take a position contrary to the foregoing representations. Company acknowledges and agrees that Investor may rely upon the foregoing representations and covenants of Company regarding the Arbitration Provisions. This Agreement shall be construed and enforced in accordance with, and all questions concerning the construction, validity, interpretation and performance of this Agreement shall be governed by, the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. The Company and Investor consent to and expressly agree that the exclusive venue for arbitration of any Claims arising under this Agreement or any other agreement between the Company and Investor or their respective affiliates (including but not limited to the Transaction Documents) or any Claim relating to the relationship of the Company and Investor or their respective affiliates shall be in the State of Delaware. Without modifying the Company's and Investor's mandatory obligations to resolve disputes hereunder pursuant to the Arbitration Provisions, for any litigation arising in connection with any of the Transaction Documents (and notwithstanding the terms (specifically including any governing law and venue terms) of any transfer agent services agreement or other agreement between the Company's transfer agent and the Company, such litigation specifically includes, without limitation any action between or involving Company and the Company's transfer agent under the Transfer Agent Instruction Letter or otherwise related to Investor in any way (specifically including, without limitation, any action where Company seeks to obtain an injunction, temporary restraining order, or otherwise prohibit the Company's transfer agent from issuing shares of Common Stock to Investor for any reason)), each party hereto hereby (i) consents to and expressly submits to the exclusive personal jurisdiction of any state or federal court sitting in the State of Delaware, (ii) expressly submits to the exclusive venue of any such court for the purposes hereof, (iii) agrees to not bring any such action (specifically including, without limitation, any action where Company seeks to obtain an injunction, temporary restraining order, or otherwise prohibit the Company's transfer agent from issuing shares of Common Stock to Investor for any reason) outside of any state or federal court sitting in the State of Delaware, and (iv) waives any claim of improper venue and any claim or objection that such courts are an inconvenient forum or any other claim, defense or objection to the bringing of any such proceeding in such jurisdiction or to any claim that such venue of the suit, action or proceeding is improper. Notwithstanding anything in the foregoing to the contrary, nothing herein shall limit, or shall be deemed or construed to limit, the ability of the Investor to realize on any collateral or any other security, or to enforce a judgment or other court ruling in favor of the Investor, including through a legal action in any court of competent jurisdiction. The Company hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any objection to jurisdiction and venue of any action instituted hereunder, any claim that it is not personally subject to the jurisdiction of any such court, and any claim that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper (including but not limited to based upon forum non conveniens). THE COMPANY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTIONS CONTEMPLATED HEREBY . The Company irrevocably waives personal service of process and consents to process being served in any suit, action or proceeding in connection with this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to Company at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any other manner permitted by law. The prevailing party in any action or dispute brought in connection with this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby shall be entitled to recover from the other party its reasonable attorney's fees and costs. If any provision of this Agreement shall be invalid or unenforceable in any jurisdiction, such invalidity or unenforceability shall not affect the validity or enforceability of the remainder of this Agreement in that jurisdiction or the validity or enforceability of any provision of this Agreement in any other jurisdiction.

- Section 10.2 PAYMENT SET ASIDE. Further, to the extent that the (i) Company makes a payment or payments to the Investor pursuant to this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby, or (ii) the Investor enforces or exercises its rights pursuant to this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby (including but not limited to the sale of the Securities), and such payment or payments or the proceeds of such enforcement or exercise or any part thereof (including but not limited to the sale of the Securities) are for any reason (i) subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, or disgorged by the Investor, or (ii) are required to be refunded, repaid or otherwise restored to the Company, a trustee, receiver, government entity, or any other person or entity under any law (including, without limitation, any bankruptcy law, foreign, state or federal law, common law or equitable cause of action), then (i) to the extent of any such restoration the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred and (ii) the Company shall immediately pay to the Investor a dollar amount equal to the amount that was for any reason (i) subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, or disgorged by the Investor, or (ii) required to be refunded, repaid or otherwise restored to the Company, a trustee, receiver, government entity, or any other person or entity under any law (including, without limitation, any bankruptcy law, foreign, state or federal law, common law or equitable cause of action).
- Section 10.3 <u>ASSIGNMENT</u>. This Agreement shall be binding upon and inure to the benefit of the Company and the Investor and their respective successors. Neither this Agreement nor any rights of the Investor or the Company hereunder may be assigned by either party to any other Person.
- Section 10.4 <u>NO THIRD PARTY BENEFICIARIES</u>. This Agreement is intended for the benefit of the Company and the Investor and their respective successors, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person, except as set forth in Section 9.3.
- Section 10.5 <u>TERMINATION</u>. The Company may terminate this Agreement at any time by written notice to the Investor, except during any Valuation Period or during the seven (7) Trading Day period after any Valuation Period). In addition, this Agreement shall automatically terminate at the end of the Commitment Period. Notwithstanding anything in this Agreement to the contrary, (i) the provisions of Articles III, IV, VI, IX of this Agreement and the agreements and covenants of the Company and the Investor set forth in Article X of this Agreement shall survive the termination of this Agreement and (ii) the Investor shall retain all rights to the Commitment Shares even if this Agreement is terminated.
- Section 10.6 <u>ENTIRE AGREEMENT</u>. The Transaction Documents, together with the exhibits and schedules thereto, contain the entire understanding of the Company and the Investor with respect to the matters covered herein and therein and supersede all prior agreements and understandings, oral or written, with respect to such matters, which the Parties acknowledge have been merged into such documents, exhibits and schedules.
- Section 10.7 <u>FEES AND EXPENSES</u>. Except as expressly set forth in the Transaction Documents or any other writing to the contrary, each party shall pay the fees and expenses of its advisers, counsel, accountants and other experts, if any, and all other expenses incurred by such party incident to the negotiation, preparation, execution, delivery and performance of this Agreement. The Company shall pay all Transfer Agent fees (including, without limitation, any fees required for same-day processing of any instruction letter delivered by the Company), stamp taxes and other taxes and duties levied in connection with the delivery of any Securities to the Investor. On the date of this Agreement, the Company shall issue the Commitment Shares to Investor for its commitment to enter into this Agreement. The Commitment Shares shall be earned in full upon the execution of this Agreement, and the issuance of the Commitment Shares is not contingent upon any other event or condition, including but not limited to the effectiveness of the Registration Statement or the Company's submission of a Put Notice to the Investor. In addition, the Company shall pay \$15,000.00 to legal counsel of the Investor on the date of this Agreement for Investor's expenses relating to the preparation of this Agreement.

- Section 10.8 <u>COUNTERPARTS.</u> This Agreement may be executed in multiple counterparts, each of which may be executed by less than all of the Parties and shall be deemed to be an original instrument which shall be enforceable against the Parties actually executing such counterparts and all of which together shall constitute one and the same instrument. This Agreement may be delivered to the other Parties hereto by email of a copy of this Agreement bearing the signature of the Parties so delivering this Agreement.
- Section 10.9 <u>SEVERABILITY</u>. In the event that any provision of this Agreement becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, this Agreement shall continue in full force and effect without said provision; provided that such severability shall be ineffective if it materially changes the economic benefit of this Agreement to any party.
- Section 10.10 <u>FURTHER ASSURANCES</u>. Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as the other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- Section 10.11 NO STRICT CONSTRUCTION. The language used in this Agreement will be deemed to be the language chosen by the Parties to express their mutual intent, and no rules of strict construction will be applied against any party.
- Section 10.12 <u>EQUITABLE RELIEF</u>. The Company recognizes that in the event that it fails to perform, observe, or discharge any or all of its obligations under this Agreement, any remedy at law may prove to be inadequate relief to the Investor. The Company therefore agrees that the Investor shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages.
- Section 10.13 <u>TITLE AND SUBTITLES.</u> The titles and subtitles used in this Agreement are used for the convenience of reference and are not to be considered in construing or interpreting this Agreement.
- Section 10.14 <u>AMENDMENTS; WAIVERS.</u> No provision of this Agreement may be amended or waived by the Parties from and after the date that is one (1) Trading Day immediately preceding the initial filing of the Registration Statement with the SEC. Subject to the immediately preceding sentence, (i) no provision of this Agreement may be amended other than by a written instrument signed by both Parties hereto and (ii) no provision of this Agreement may be waived other than in a written instrument signed by the party against whom enforcement of such waiver is sought. No failure or delay in the exercise of any power, right or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such power, right or privilege preclude other or further exercise thereof or of any other right, power or privilege.
- Section 10.15 <u>PUBLICITY</u>. The Company and the Investor shall consult with each other in issuing any press releases or otherwise making public statements with respect to the transactions contemplated hereby and no party shall issue any such press release or otherwise make any such public statement, other than as required by law, without the prior written consent of the other Parties, which consent shall not be unreasonably withheld or delayed, except that no prior consent shall be required if such disclosure is required by law, in which such case the disclosing party shall provide the other party with prior notice of such public statement. Notwithstanding the foregoing, the Company shall not publicly disclose the name of the Investor without the prior written consent of the Investor, except to the extent required by law. The Investor acknowledges that this Agreement and all or part of the Transaction Documents may be deemed to be "material contracts," as that term is defined by Item 601(b) (10) of Regulation S-K, and that the Company may therefore be required to file such documents as exhibits to reports or registration statements filed under the Securities Act or the Exchange Act. The Investor further agrees that the status of such documents and materials as material contracts shall be determined solely by the Company, in consultation with its counsel.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their respective officers thereunto duly authorized as of the day and year first above written.

THE COMPANY:

SUPER LEAGUE ENTERPRISE, INC.

By:/s/ Ann Hand Name: Ann Hand

Title: Chief Executive Officer

INVESTOR:

HUDSON GLOBAL VENTURES, LLC

By: <u>/s/ Seth Ahdoot</u> Name: Seth Ahdoot Title: Member

[Signature Page to equity purchase agreement]

EXHIBIT A

FORM OF PUT NOTICE

TO: HUDSON GLOBAL VENTURES, LLC DATE:	
We refer to the equity purchase agreement, dated February 14, 2025 (the " $\frac{1}{2}$ 4 Capitalized terms defined in the Agreement shall, unless otherwise defined herein, have	greement"), entered into by and between Super League Enterprise, Inc. and you the same meaning when used herein.
We hereby:	
1) Give you notice that we require you to purchase Put Shares pursuant	to the Agreement; and
2) The Initial Purchase Price pursuant to the Agreement is; and	
2) Certify that, as of the date hereof, the conditions set forth in Section 7.2 of the Agree	ment are satisfied.
S	SUPER LEAGUE ENTERPRISE, INC.
1	By: Name: <u>Ann Hand</u> Title: <u>Chief Executive Officer</u>

EXHIBIT B

FORM OF OFFICER'S CERTIFICATE OF SUPER LEAGUE ENTERPRISE, INC.

Pursuant to Section 7.2(k) of that certain equity purchase agreement, dated February 14, 2025 (the <u>Agreement</u>"), by and between Super League Enterprise, Inc. (the "<u>Company</u>") and Hudson Global Ventures, LLC (the "<u>Investor</u>"), the undersigned, in his capacity as Chief Executive Officer of the Company, and not in his individual capacity, hereby certifies, as of the date hereof (such date, the "<u>Condition Satisfaction Date</u>"), the following:

	-
1. The representations and warranties of the Company are true and correct in all mathe Condition Satisfaction Date (except for representations and warranties specifically made as of circumstances occurring or existing to and including the Condition Satisfaction Date, except for a warranties of the Company set forth in the Agreement to be incorrect and which have been corrected warranties.	a particular date) with respect to all periods, and as to all events and my conditions which have temporarily caused any representations or
2. All of the conditions precedent to the obligation of the Investor to purchase Put 7.2 of the Agreement, have been satisfied as of the Condition Satisfaction Date.	Shares set forth in the Agreement, including but not limited to Section
Capitalized terms used herein shall have the meanings set forth in the Agreement unless other	rwise defined herein.
IN WITNESS WHEREOF, the undersigned has hereunto affixed his hand as of the	, 20
	By:Name: Ann Hand Title: Chief Executive Officer

EXHIBIT C

ARBITRATION PROVISIONS

- 1. <u>Dispute Resolution</u>. Each party consents to and expressly agrees that the exclusive venue for arbitration of any dispute arising out of or relating to any of the Transaction Documents or the relationship of the parties or their affiliates shall be in the State of Delaware. For purposes of this <u>Exhibit C</u>, the term "Claims" means any disputes, claims, demands, causes of action, requests for injunctive relief, requests for specific performance, questions regarding severability of any provisions of the Transaction Documents, liabilities, damages, losses, or controversies whatsoever arising from, related to, or connected with the transactions contemplated in the Transaction Documents and any communications between the parties related thereto, including without limitation any claims of mutual mistake, mistake, fraud, misrepresentation, failure of formation, failure of consideration, promissory estoppel, unconscionability, failure of condition precedent, rescission, and any statutory claims, tort claims, contract claims, or claims to void, invalidate or terminate the Agreement (or these Arbitration Provisions (defined below)) or any of the other Transaction Documents. The parties to this Agreement (the "parties") hereby agree that the Claims may be arbitrated in one or more Arbitrations pursuant to these Arbitration Provisions (one for an injunction or injunctions and a separate one for all other Claims). The parties hereby agree that the arbitration provisions set forth in this <u>Exhibit C</u> ("Arbitration Provisions") are binding on each of them. As a result, any attempt to rescind the Agreement (or these Arbitration Provisions) or any other Transaction Document) or declare the Agreement (or these Arbitration Provisions) or any other Transaction Document invalid or unenforceable pursuant to Section 29 of the 1934 Act or for any other reason is subject to these Arbitration Provisions. These Arbitration Provisions shall also survive any termination or expiration of the Agreement. Any capitalized term not defined in
- 2. <u>Arbitration</u>. Except as otherwise provided herein, all Claims must be submitted to arbitration ("Arbitration") to be conducted exclusively in the State of Delaware and pursuant to the terms set forth in these Arbitration Provisions. Subject to the arbitration appeal right provided for in Paragraph 5 below (the "Appeal Right"), the parties agree that the award of the arbitrator rendered pursuant to Paragraph 4 below (the "Arbitration Award") shall be (a) final and binding upon the parties, (b) the sole and exclusive remedy between them regarding any Claims, counterclaims, issues, or accountings presented or pleaded to the arbitrator, and (c) promptly payable in United States dollars free of any tax, deduction or offset (with respect to monetary awards). Subject to the Appeal Right, any costs or fees, including without limitation attorneys' fees, incurred in connection with or incident to enforcing the Arbitration Award shall, to the maximum extent permitted by law, be charged against the party resisting such enforcement. Judgment upon the Arbitration Award will be entered and enforced by any state or federal court sitting in the State of Delaware.
- 3. The Arbitration Act. The parties hereby incorporate herein the provisions and procedures set forth in the Delaware Uniform Arbitration Act, Chapter 38 (as amended or superseded from time to time, the "Arbitration Act"). Notwithstanding the foregoing, pursuant to, and to the maximum extent permitted by, the Arbitration Act, in the event of conflict or variation between the terms of these Arbitration Provisions and the provisions of the Arbitration Act, the terms of these Arbitration Provisions shall control and the parties hereby waive or otherwise agree to vary the effect of all requirements of the Arbitration Act that may conflict with or vary from these Arbitration Provisions.
- 4. <u>Arbitration Proceedings</u>. Arbitration between the parties will be subject to the following:
- 4 . 1 *Initiation of Arbitration*. Pursuant to the Arbitration Act, the parties agree that a party may initiate Arbitration by giving written notice to the other party ("Arbitration Notice") in the same manner that notice is permitted under Section 8(f) of the Agreement; provided, however, that the Arbitration Notice may not be given by email or fax. Arbitration will be deemed initiated as of the date that the Arbitration Notice is deemed physically delivered to such other party under Section 8(f) of the Agreement (the "Service Date"). After the Service Date, information may be delivered, and notices may be given, by email or fax pursuant to Section 8(f) of the Agreement or any other method permitted thereunder. The Arbitration Notice must describe the nature of the controversy, the remedies sought, and the election to commence Arbitration proceedings. All Claims in the Arbitration Notice must be pleaded consistent with the Delaware Rules of Civil Procedure.

4.2 Selection and Payment of Arbitrator.

- (a) Within ten (10) calendar days after the Service Date, Investor shall select and submit to Company the names of three (3) arbitrators that are designated as "neutrals" or qualified arbitrators by JAMS (https://www.jamsadr.com/) or other arbitration service provider agreed upon by the parties (such three (3) designated persons hereunder are referred to herein as the "Proposed Arbitrators"). For the avoidance of doubt, each Proposed Arbitrator must be qualified as a "neutral" with JAMS or other arbitration service provider agreed upon by the parties. Within five (5) calendar days after Investor has submitted to Company the names of the Proposed Arbitrators, Company must select, by written notice to Investor, one (1) of the Proposed Arbitrators to act as the arbitrator for the parties under these Arbitration Provisions. If Company fails to select one of the Proposed Arbitrators in writing within such 5-day period, then Investor may select the arbitrator from the Proposed Arbitrators by providing written notice of such selection to Company.
- (b) If Investor fails to submit to Company the Proposed Arbitrators within ten (10) calendar days after the Service Date pursuant to subparagraph (a) above, then Company may at any time prior to Investor so designating the Proposed Arbitrators, identify the names of three (3) arbitrators that are designated as "neutrals" or qualified arbitrators by JAMS or other arbitration service provider agreed upon by the parties by written notice to Investor. Investor may then, within five (5) calendar days after Company has submitted notice of its Proposed Arbitrators to Investor, select, by written notice to Company, one (1) of the Proposed Arbitrators to act as the arbitrator for the parties under these Arbitration Provisions. If Investor fails to select in writing and within such 5-day period one (1) of the three (3) Proposed Arbitrators selected by Company, then Company may select the arbitrator from its three (3) previously selected Proposed Arbitrators by providing written notice of such selection to Investor.
- (c) If a Proposed Arbitrator chosen to serve as arbitrator declines or is otherwise unable to serve as arbitrator, then the party that selected such Proposed Arbitrator may select one (1) of the other three (3) Proposed Arbitrators within three (3) calendar days of the date the chosen Proposed Arbitrator declines or notifies the parties he or she is unable to serve as arbitrator. If all three (3) Proposed Arbitrators decline or are otherwise unable to serve as arbitrator, then the arbitrator selection process shall begin again in accordance with this Paragraph 4.2.
- (d) The date that the Proposed Arbitrator selected pursuant to this Paragraph 4.2 agrees in writing (including via email) delivered to both parties to serve as the arbitrator hereunder is referred to herein as the "Arbitration Commencement Date". If an arbitrator resigns or is unable to act during the Arbitration, a replacement arbitrator shall be chosen in accordance with this Paragraph 4.2 to continue the Arbitration. If JAMS or other arbitration service provider agreed upon by the parties ceases to exist or to provide a list of neutrals and there is no successor thereto, then the arbitrator shall be selected under the then prevailing rules of the American Arbitration Association.
- (e) Subject to Paragraph 4.10 below, the cost of the arbitrator must be paid equally by both parties. Subject to Paragraph 4.10 below, if one party refuses or fails to pay its portion of the arbitrator fee, then the other party can advance such unpaid amount, with such amount being added to or subtracted from, as applicable, the Arbitration Award.
- 4.3 Applicability of Certain Delaware Rules. The parties agree that the Arbitration shall be conducted generally in accordance with the Delaware Rules of Civil Procedure and the Delaware Rules of Evidence. More specifically, the Delaware Rules of Civil Procedure shall apply, without limitation, to the filing of any pleadings, motions or memoranda, the conducting of discovery, and the taking of any depositions. The Delaware Rules of Evidence shall apply to any hearings, whether telephonic or in person, held by the arbitrator. Notwithstanding the foregoing, it is the parties' intent that the incorporation of such rules will in no event supersede these Arbitration Provisions. In the event of any conflict between the Delaware Rules of Civil Procedure or the Delaware Rules of Evidence and these Arbitration Provisions, these Arbitration Provisions shall control.
- 4.4 Answer and Default. An answer and any counterclaims to the Arbitration Notice shall be required to be delivered to the party initiating the Arbitration within twenty (20) calendar days after the Arbitration Commencement Date. If an answer is not delivered by the required deadline, the arbitrator must provide written notice to the defaulting party stating that the arbitrator will enter a default award against such party if such party does not file an answer within five (5) calendar days of receipt of such notice. If an answer is not filed within the five (5) day extension period, the arbitrator must render a default award, consistent with the relief requested in the Arbitration Notice, against a party that fails to submit an answer within such time period.

4.5 [Intentionally Omitted].

- 4.6 *Discovery*. The parties agree that discovery shall be conducted as follows:
- (a) Written discovery will only be allowed if the likely benefits of the proposed written discovery outweigh the burden or expense thereof, and the written discovery sought is likely to reveal information that will satisfy a specific element of a claim or defense already pleaded in the Arbitration. The party seeking written discovery shall always have the burden of showing that all of the standards and limitations set forth in these Arbitration Provisions are satisfied. The scope of discovery in the Arbitration proceedings shall also be limited as follows:
 - (i) To facts directly connected with the transactions contemplated by the Agreement.
 - (ii) To facts and information that cannot be obtained from another source or in another manner that is more convenient, less burdensome or less expensive than in the manner requested.
- (b) No party shall be allowed (i) more than fifteen (15) interrogatories (including discrete subparts), (ii) more than fifteen (15) requests for admission (including discrete subparts), (iii) more than ten (10) document requests (including discrete subparts), or (iv) more than three (3) depositions (excluding expert depositions) for a maximum of seven (7) hours per deposition. The costs associated with depositions will be borne by the party taking the deposition. The party defending the deposition will submit a notice to the party taking the deposition of the estimated attorneys' fees that such party expects to incur in connection with defending the deposition. If the party defending the deposition fails to submit an estimate of attorneys' fees within five (5) calendar days of its receipt of a deposition notice, then such party shall be deemed to have waived its right to the estimated attorneys' fees. The party taking the deposition must pay the party defending the deposition the estimated attorneys' fees prior to taking the deposition, unless such obligation is deemed to be waived as set forth in the immediately preceding sentence. If the party taking the deposition believes that the estimated attorneys' fees are unreasonable, such party may submit the issue to the arbitrator for a decision.
- (c) All discovery requests (including document production requests included in deposition notices) must be submitted in writing to the arbitrator and the other party. The party submitting the written discovery requests must include with such discovery requests a detailed explanation of how the proposed discovery requests satisfy the requirements of these Arbitration Provisions and the Delaware Rules of Civil Procedure. The receiving party will then be allowed, within five (5) calendar days of receiving the proposed discovery requests, to submit to the arbitrator an estimate of the attorneys' fees and costs associated with responding to such written discovery requests and a written challenge to each applicable discovery request. After receipt of an estimate of attorneys' fees and costs and/or challenge(s) to one or more discovery requests, consistent with subparagraph (c) above, the arbitrator will within three (3) calendar days make a finding as to the likely attorneys' fees and costs associated with responding to the discovery requests, and (ii) requires the responding party to requests as limited by the arbitrator within twenty-five (25) calendar days of the arbitrator's finding with respect to such discovery requests. If a party entitled to submit an estimate of attorneys' fees and costs and/or a challenge to discovery requests to do so within such 5-day period, the arbitrator will make a finding that (A) there are no attorneys' fees or costs associated with responding to such discovery requests, and (B) the responding party must respond to such discovery requests (as may be limited by the arbitrator) within twenty-five (25) calendar days of the arbitrator's finding with respect to such discovery requests. Any party submitting any written discovery requests, including without limitation interrogatories, requests for production subpoenas to a party or a third party, or requests for admissions, must prepay the estimated attorneys' fees and costs, before the responding party has any obligation to produce or res
- (d) In order to allow a written discovery request, the arbitrator must find that the discovery request satisfies the standards set forth in these Arbitration Provisions and the Delaware Rules of Civil Procedure. The arbitrator must strictly enforce these standards. If a discovery request does not satisfy any of the standards set forth in these Arbitration Provisions or the Delaware Rules of Civil Procedure, the arbitrator may modify such discovery request to satisfy the applicable standards, or strike such discovery request in whole or in part.
- (e) Each party may submit expert reports (and rebuttals thereto), provided that such reports must be submitted within sixty (60) days of the Arbitration Commencement Date. Each party will be allowed a maximum of two (2) experts. Expert reports must contain the following: (i) a complete statement of all opinions the expert will offer at trial and the basis and reasons for them; (ii) the expert's name and qualifications, including a list of all the expert's publications within the preceding ten (10) years, and a list of any other cases in which the expert has testified at trial or in a deposition or prepared a report within the preceding ten (10) years; and (iii) the compensation to be paid for the expert's report and testimony. The parties are entitled to depose any other party's expert witness one (1) time for no more than four (4) hours. An expert may not testify in a party's case-in-chief concerning any matter not fairly disclosed in the expert report.

- 4.6 Dispositive Motions. Each party shall have the right to submit dispositive motions pursuant to the Delaware Rules of Civil Procedure (a "Dispositive Motion"). The party submitting the Dispositive Motion may, but is not required to, deliver to the arbitrator and to the other party a memorandum in support (the "Memorandum in Support") of the Dispositive Motion. Within seven (7) calendar days of delivery of the Memorandum in Support, the other party shall deliver to the arbitrator and to the other party a memorandum in opposition to the Memorandum in Support (the "Memorandum in Opposition"). Within seven (7) calendar days of delivery of the Memorandum in Opposition, as applicable, the party that submitted the Memorandum in Support shall deliver to the arbitrator and to the other party a reply memorandum the Memorandum in Opposition ("Reply Memorandum"). If the applicable party shall fail to deliver the Memorandum in Opposition as required above, or if the other party fails to deliver the Reply Memorandum as required above, then the applicable party shall lose its right to so deliver the same, and the Dispositive Motion shall proceed regardless.
- 4 . 7 Confidentiality. All information disclosed by either party (or such party's agents) during the Arbitration process (including without limitation information disclosed during the discovery process or any Appeal (defined below)) shall be considered confidential in nature. Each party agrees not to disclose any confidential information received from the other party (or its agents) during the Arbitration process (including without limitation during the discovery process or any Appeal) unless (a) prior to or after the time of disclosure such information becomes public knowledge or part of the public domain, not as a result of any inaction or action of the receiving party or its agents, (b) such information is required by a court order, subpense or similar legal duress to be disclosed if such receiving party has notified the other party thereof in writing and given it a reasonable opportunity to obtain a protective order from a court of competent jurisdiction prior to disclosure, or (c) such information is disclosed to the receiving party's agents, representatives and legal counsel on a need to know basis who each agree in writing not to disclose such information to any third party. The arbitrator is hereby authorized and directed to issue a protective order to prevent the disclosure of privileged information and confidential information upon the written request of either party.
- 4.8 *Authorization; Timing; Scheduling Order.* Subject to all other portions of these Arbitration Provisions, the parties hereby authorize and direct the arbitrator to take such actions and make such rulings as may be necessary to carry out the parties' intent for the Arbitration proceedings to be efficient and expeditious. The parties hereby agree that an Arbitration Award must be made within one hundred twenty (120) calendar days after the Arbitration Commencement Date. The arbitrator is hereby authorized and directed to hold a scheduling conference within ten (10) calendar days after the Arbitration Commencement Date in order to establish a scheduling order with various binding deadlines for discovery, expert testimony, and the submission of documents by the parties to enable the arbitrator to render a decision prior to the end of such 120-day period.
- 4.9 Relief. The arbitrator shall have the right to award or include in the Arbitration Award (or in a preliminary ruling) any relief which the arbitrator deems proper under the circumstances, including, without limitation, specific performance and injunctive relief, provided that the arbitrator may not award exemplary or punitive damages.
- 4.10 Fees and Costs. As part of the Arbitration Award, the arbitrator is hereby directed to require the losing party (the party being awarded the least amount of money by the arbitrator, which, for the avoidance of doubt, shall be determined without regard to any statutory fines, penalties, fees, or other charges awarded to any party) to (a) pay the full amount of any unpaid costs and fees of the Arbitration, and (b) reimburse the prevailing party for all reasonable attorneys' fees, arbitrator costs and fees, deposition costs, other discovery costs, and other expenses, costs or fees paid or otherwise incurred by the prevailing party in connection with the Arbitration.

5. Arbitration Appeal.

5.1 Initiation of Appeal. Following the entry of the Arbitration Award, either party (the "Appellant") shall have a period of thirty (30) calendar days in which to notify the other party (the "Appellee"), in writing, that the Appellant elects to appeal (the "Appeal") the Arbitration Award (such notice, an "Appeal Notice") to a panel of arbitrators as provided in Paragraph 5.2 below. The date the Appellant delivers an Appeal Notice to the Appellee is referred to herein as the "Appeal Date". The Appeal Notice must be delivered to the Appellee in accordance with the provisions of Paragraph 4.1 above with respect to delivery of an Arbitration Notice. In addition, together with delivery of the Appeal Notice to the Appellee, the Appellant must also pay for (and provide proof of such payment to the Appellee together with delivery of the Appeal Notice) a bond in the amount of 110% of the sum the Appellant owes to the Appellee as a result of the Arbitration Award the Appellant is appealing. In the event an Appeal Notice to the Appellee (together with proof of payment of the applicable bond) in compliance with the provisions of this Paragraph 5.1, the Appeal will occur as a matter of right and, except as specifically set forth herein, will not be further conditioned. In the event a party does not deliver an Appeal Notice (along with proof of payment of the applicable bond) to the other party within the deadline prescribed in this Paragraph 5.1, such party shall lose its right to appeal the Arbitration Award. If no party delivers an Appeal Notice (along with proof of payment of the applicable bond) to the other party within the deadline described in this Paragraph 5.1, the Arbitration Award shall be final. The parties acknowledge and agree that any Appeal shall be deemed part of the parties' agreement to arbitrate for purposes of these Arbitration Provisions and the Arbitration Act.

- 5.2 Selection and Payment of Appeal Panel. In the event an Appellant delivers an Appeal Notice to the Appellee (together with proof of payment of the applicable bond) in compliance with the provisions of Paragraph 5.1 above, the Appeal will be heard by a three (3) person arbitration panel (the "Appeal Panel").
- (a) Within ten (10) calendar days after the Appeal Date, the Appellee shall select and submit to the Appellant the names of five (5) arbitrators that are designated as "neutrals" or qualified arbitrators by JAMS (https://www.jamsadr.com/) or other arbitration service provider agreed upon by the parties (such five (5) designated persons hereunder are referred to herein as the "Proposed Appeal Arbitrators"). For the avoidance of doubt, each Proposed Appeal Arbitrator must be qualified as a "neutral" with JAMS or other arbitration service provider agreed upon by the parties, and shall not be the arbitrator who rendered the Arbitration Award being appealed (the "Original Arbitrator"). Within five (5) calendar days after the Appellee has submitted to the Appellant the names of the Proposed Appeal Arbitrators, the Appellant must select, by written notice to the Appellee, three (3) of the Proposed Appeal Arbitrators to act as the members of the Appeal Panel. If the Appellant fails to select three (3) of the Proposed Appeal Arbitrators in writing within such 5-day period, then the Appellee may select such three (3) arbitrators from the Proposed Appeal Arbitrators by providing written notice of such selection to the Appellant.
- (b) If the Appellee fails to submit to the Appellant the names of the Proposed Appeal Arbitrators within ten (10) calendar days after the Appeal Date pursuant to subparagraph (a) above, then the Appellant may at any time prior to the Appellee so designating the Proposed Appeal Arbitrators, identify the names of five (5) arbitrators that are designated as "neutrals" or qualified arbitrators by JAMS or other arbitration service provider agreed upon by the parties (none of whom may be the Original Arbitrator) by written notice to the Appellee. The Appellee may then, within five (5) calendar days after the Appellant has submitted notice of its selected arbitrators to the Appellee, select, by written notice to the Appellant, three (3) of such selected arbitrators to serve on the Appeal Panel. If the Appellee fails to select in writing within such 5-day period three (3) of the arbitrators selected by the Appellant to serve as the members of the Appeal Panel, then the Appellant may select the three (3) members of the Appeal Panel from the Appellant's list of five (5) arbitrators by providing written notice of such selection to the Appellee.
- (c) If a selected Proposed Appeal Arbitrator declines or is otherwise unable to serve, then the party that selected such Proposed Appeal Arbitrator may select one (1) of the other five (5) designated Proposed Appeal Arbitrators within three (3) calendar days of the date a chosen Proposed Appeal Arbitrator declines or notifies the parties he or she is unable to serve as an arbitrator. If at least three (3) of the five (5) designated Proposed Appeal Arbitrators decline or are otherwise unable to serve, then the Proposed Appeal Arbitrator selection process shall begin again in accordance with this Paragraph 5.2; provided, however, that any Proposed Appeal Arbitrators who have already agreed to serve shall remain on the Appeal Panel.
- (d) The date that all three (3) Proposed Appeal Arbitrators selected pursuant to this Paragraph 5.2 agree in writing (including via email) delivered to both the Appellant and the Appellee to serve as members of the Appeal Panel hereunder is referred to herein as the "Appeal Commencement Date". No later than five (5) calendar days after the Appeal Commencement Date, the Appellee shall designate in writing (including via email) to the Appellant and the Appeal Panel the name of one (1) of the three (3) members of the Appeal Panel to serve as the lead arbitrator in the Appeal proceedings. Each member of the Appeal Panel shall be deemed an arbitrator for purposes of these Arbitration Provisions and the Arbitration Act, provided that, in conducting the Appeal, the Appeal Panel may only act or make determinations upon the approval or vote of no less than the majority vote of its members, as announced or communicated by the lead arbitrator on the Appeal Panel. If an arbitrator on the Appeal Panel ceases or is unable to act during the Appeal proceedings, a replacement arbitrator shall be chosen in accordance with Paragraph 5.2 above to continue the Appeal as a member of the Appeal Panel. If JAMS or other arbitration service provider agreed upon by the parties ceases to exist or to provide a list of neutrals, then the arbitrators for the Appeal Panel shall be selected under the then prevailing rules of the American Arbitration Association.
 - (d) Subject to Paragraph 5.7 below, the cost of the Appeal Panel must be paid entirely by the Appellant.
- 5.3 Appeal Procedure. The Appeal will be deemed an appeal of the entire Arbitration Award. In conducting the Appeal, the Appeal Panel shall conduct a de novo review of all Claims described or otherwise set forth in the Arbitration Notice. Subject to the foregoing and all other provisions of this Paragraph 5, the Appeal Panel shall conduct the Appeal in a manner the Appeal Panel considers appropriate for a fair and expeditious disposition of the Appeal, may hold one or more hearings and permit oral argument, and may review all previous evidence and discovery, together with all briefs, pleadings and other documents filed with the Original Arbitrator (as well as any documents filed with the Appeal Panel pursuant to Paragraph 5.4(a) below). Notwithstanding the foregoing, in connection with the Appeal, the Appeal Panel Pa

5.4 Timing.

- (a) Within seven (7) calendar days of the Appeal Commencement Date, the Appellant (i) shall deliver or cause to be delivered to the Appeal Panel copies of the Appeal Notice, all discovery conducted in connection with the Arbitration, and all briefs, pleadings and other documents filed with the Original Arbitrator (which material Appellee shall have the right to review and supplement if necessary), and (ii) may, but is not required to, deliver to the Appeal Panel and to the Appellee a Memorandum in Support of the Appellant's arguments concerning or position with respect to all Claims, counterclaims, issues, or accountings presented or pleaded in the Arbitration. Within seven (7) calendar days of the Appellant's delivery of the Memorandum in Support, as applicable, the Appellee shall deliver to the Appeal Panel and to the Appellant a Memorandum in Opposition to the Memorandum in Support. Within seven (7) calendar days of the Appellee's delivery of the Memorandum in Opposition, as applicable, the Appellant shall deliver to the Appeal Panel and to the Appellee a Reply Memorandum to the Memorandum in Opposition. If the Appellant shall fail to substantially comply with the requirements of clause (i) of this subparagraph (a), the Appellant shall lose its right to appeal the Arbitration Award, and the Arbitration Award shall be final. If the Appellee shall fail to deliver the Memorandum in Opposition as required above, or if the Appellant shall fail to deliver the Reply Memorandum as required above, then the Appellee or the Appellant, as the case may be, shall lose its right to so deliver the same, and the Appeal shall proceed regardless.
- (b) Subject to subparagraph (a) above, the parties hereby agree that the Appeal must be heard by the Appeal Panel within thirty (30) calendar days of the Appeal Commencement Date, and that the Appeal Panel must render its decision within thirty (30) calendar days after the Appeal is heard (and in no event later than sixty (60) calendar days after the Appeal Commencement Date).
- 5.5 Appeal Panel Award. The Appeal Panel shall issue its decision (the "Appeal Panel Award") through the lead arbitrator on the Appeal Panel. Notwithstanding any other provision contained herein, the Appeal Panel Award shall (a) supersede in its entirety and make of no further force or effect the Arbitration Award (provided that any protective orders issued by the Original Arbitrator shall remain in full force and effect), (b) be final and binding upon the parties, with no further rights of appeal, (c) be the sole and exclusive remedy between the parties regarding any Claims, counterclaims, issues, or accountings presented or pleaded in the Arbitration, and (d) be promptly payable in United States dollars free of any tax, deduction or offset (with respect to monetary awards). Any costs or fees, including without limitation attorneys' fees, incurred in connection with or incident to enforcing the Appeal Panel Award shall, to the maximum extent permitted by law, be charged against the party resisting such enforcement. Judgment upon the Appeal Panel Award will be entered and enforced by a state or federal court sitting in the State of Delaware.
- 5 . 6 Relief. The Appeal Panel shall have the right to award or include in the Appeal Panel Award any relief which the Appeal Panel deems proper under the circumstances, including, without limitation, specific performance and injunctive relief, provided that the Appeal Panel may not award exemplary or punitive damages.
- 5.7 Fees and Costs. As part of the Appeal Panel Award, the Appeal Panel is hereby directed to require the losing party (the party being awarded the least amount of money by the arbitrator, which, for the avoidance of doubt, shall be determined without regard to any statutory fines, penalties, fees, or other charges awarded to any party) to (a) pay the full amount of any unpaid costs and fees of the Arbitration and the Appeal Panel, and (b) reimburse the prevailing party (the party being awarded the most amount of money by the Appeal Panel, which, for the avoidance of doubt, shall be determined without regard to any statutory fines, penalties, fees, or other charges awarded to any part) the reasonable attorneys' fees, arbitrator and Appeal Panel costs and fees, deposition costs, other discovery costs, and other expenses, costs or fees paid or otherwise incurred by the prevailing party in connection with the Arbitration (including without limitation in connection with the Appeal).

Miscellaneous.

- 6.1 Severability. If any part of these Arbitration Provisions is found to violate or be illegal under applicable law, then such provision shall be modified to the minimum extent necessary to make such provision enforceable under applicable law, and the remainder of the Arbitration Provisions shall remain unaffected and in full force and effect.
 - 6.2 Governing Law. These Arbitration Provisions shall be governed by the laws of the State of Delaware without regard to the conflict of laws principles therein.
- 6.3 Interpretation. The headings of these Arbitration Provisions are for convenience of reference only and shall not form part of, or affect the interpretation of, these Arbitration Provisions.

6.4	Waiver. No waiver of any provision of these Arbitration Provisions shall be effective unless it is in the form of a writing signed by the party granting the waiver.		
6.5	Time is of the Essence. Time is expressly made of the essence with respect to each and every provision of these Arbitration Provisions.		
[Remainder of page intentionally left blank]			

REGISTRATION RIGHTS AGREEMENT

REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of February 14, 2025, by and between **SUPER LEAGUE ENTERPRISE**, **INC.**, a Delaware corporation (the "Company"), and **HUDSON GLOBAL VENTURES**, **LLC**, a Nevada limited liability company (together with it permitted assigns, the "Investor"). Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the equity purchase agreement by and between the parties hereto, dated as of the date hereof (as amended, restated, supplemented or otherwise modified from time to time, the "Purchase Agreement").

WHEREAS:

The Company has agreed, upon the terms and subject to the conditions of the Purchase Agreement, to sell to the Investor up to Three Million One Hundred Thousand Dollars (\$3,100,000.00) of Put Shares and Commitment Shares (each as defined in the Purchase Agreement) and to induce the Investor to enter into the Purchase Agreement, the Company has agreed to provide certain registration rights under the Securities Act of 1933, as amended, and the rules and regulations thereunder, or any similar successor statute (collectively, the "Securities Act"), and applicable state securities laws.

NOW, THEREFORE, in consideration of the promises and the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Company and the Investor hereby agree as follows:

1. <u>DEFINITIONS</u>.

As used in this Agreement, the following terms shall have the following meanings:

- a. "Investor" shall have the meaning set forth above.
- b. "Person" means any individual or entity including but not limited to any corporation, a limited liability company, an association, a partnership, an organization, a business, an individual, a governmental or political subdivision thereof or a governmental agency.
- c. "Register," "registered," and "registration" refer to a registration effected by preparing and filing one or more registration statements of the Company in compliance with the Securities Act and/or pursuant to Rule 415 under the Securities Act or any successor rule providing for offering securities on a continuous basis ("Rule 415"), and the declaration or ordering of effectiveness of such registration statement(s) by the United States Securities and Exchange Commission (the <u>SEC</u>").
- d. "Registrable Securities" means all of the Put Shares which have been, or which may, from time to time be issued, including without limitation all of the shares of Common Stock (as defined in the Purchase Agreement) (the "Common Stock") which have been issued or will be issued to the Investor under the Purchase Agreement (without regard to any beneficial ownership or restriction on purchases therein), as well as the Commitment Shares (as defined in the Purchase Agreement) (the "Commitment Shares"), without regard to any limitation on beneficial ownership or restriction on purchases therein, and shares of Common Stock issued to the Investor as a result of any stock split, stock dividend, recapitalization, exchange or similar event or otherwise, without regard to any limitation on beneficial ownership in the Purchase Agreement or rights and designations of the Commitment Shares.
 - e. "Registration Statement" means one or more registration statements of the Company.

2. REGISTRATION.

- Mandatory Registration. The Company shall, within seven (7) calendar days from the date of this Agreement, file with the SEC an initial Registration Statement covering the maximum number of Registrable Securities as shall be permitted to be included thereon in accordance with applicable SEC rules, regulations and interpretations so as to permit the resale of such Registrable Securities by the Investor for Investor's resale of the Registrable Securities), including but not limited to under Rule 415 under the Securities Act at then prevailing market prices (and not fixed prices), subject to the aggregate number of authorized shares of the Company's Common Stock then available for issuance in its Certificate of Incorporation. The initial Registration Statement shall register only the Registrable Securities. The Investor and its counsel shall have a reasonable opportunity to review and comment upon such Registration Statement and any amendment or supplement to such Registration Statement and any related prospectus prior to its filing with the SEC, and the Company shall give due consideration to all reasonable comments. The Investor shall furnish all information reasonably requested by the Company for inclusion therein. The Company shall have the Registration Statement declared effective by the SEC within sixty (60) calendar days from the date hereof (or at the earliest possible date if prior to sixty (60) calendar days from the date hereof), and any amendment to the Registration Statement thereafter declared effective by the SEC at the earliest possible date. The Company shall keep the Registration Statement effective, including but not limited to pursuant to Rule 415 promulgated under the Securities Act and available for the resale by the Investor of all of the Registrable Securities covered thereby at all times until the date on which the Investor shall have sold all the Registrable Securities and the Maximum Commitment Amount (as defined in the Purchase Agreement) under the Purchase Agreement has been drawn down by the Company pursuant to a Registration Statement (the "Registration Period"). The Registration Statement (including any amendments or supplements thereto and prospectuses contained therein) shall not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein, or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading. In the event that (i) the Registration Statement or New Registration Statement (as defined below) becomes stale after the initial effectiveness of such Registration Statement or New Registration Statement and (ii) the Investor still has ownership of any of the Registrable Securities, the Company shall immediately file one or more post-effective amendments to facilitate the SEC's declaration of effectiveness with respect to such Registration Statement or New Registration Statement.
- b . Rule 424 Prospectus. The Company shall, as required by applicable securities regulations, from time to time file (in each case, at the earliest possible date) with the SEC, pursuant to Rule 424 promulgated under the Securities Act, the prospectus and prospectus supplements, if any, to be used in connection with sales of the Registrable Securities under the Registration Statement. The Company shall file such initial prospectus covering the Investor's sale of the Registrable Securities on the same date that the Registration Statement is declared effective by the SEC. The Investor and its counsel shall have a reasonable opportunity to review and comment upon such prospectus prior to its filing with the SEC, and the Company shall give due consideration to all such comments. The Investor shall use its reasonable best efforts to comment upon such prospectus within one (1) Business Day from the date the Investor receives the final pre-filing version of such prospectus.
- c. <u>Sufficient Number of Shares Registered.</u> In the event the number of shares available under the Registration Statement is insufficient to cover all of the Registrable Securities, the Company shall amend the Registration Statement or file a new Registration Statement (a "<u>New Registration Statement</u>"), so as to cover all of such Registrable Securities (subject to the limitations set forth in Section 2(a)) as soon as practicable, but in any event not later than ten (10) Business Days after the necessity therefor arises, subject to any limits that may be imposed by the SEC pursuant to Rule 415 under the Securities Act. The Company shall use it reasonable best efforts to cause such amendment and/or New Registration Statement to become effective as soon as practicable following the filing thereof. In the event that any of the Registrable Securities are not included in the Registration Statement, and the Company files any other registration statement under the Securities Act (other than on Form S-4, Form S-8, or with respect to other employee related plans or rights offerings) ("<u>Other Registration Statement</u>") then the Company shall include such remaining Registrable Securities in such Other Registration Statement or otherwise have been registered for resale as described above.

d. Offering. If the staff of the SEC (the "Staff") or the SEC seeks to characterize any offering pursuant to a Registration Statement filed pursuant to this Agreement as constituting an offering of securities that does not permit such Registration Statement to become effective and be used for resales by the Investor under Rule 415 at then prevailing market prices (and not fixed prices), or if after the filing of the initial Registration Statement with the SEC pursuant to Section 2(a), the Company is otherwise required by the Staff or the SEC to reduce the number of Registrable Securities to be included in such initial Registration Statement, then the Company shall reduce the number of Registrable Securities to be included in such initial Registration Statement (with the prior consent, which shall not be unreasonably withheld, of the Investor and its legal counsel as to the specific Registrable Securities to be removed therefrom) until such time as the Staff and the SEC shall so permit such Registration Statement of the SEC shall so permit such Registration Statement (with the prior consent, which shall not be unreasonably withheld, of the Investor and its legal counsel as to the specific Registrable Securities to be removed therefrom) until such time as the Staff and the SEC shall so permit such Registration Statement of the SEC shall so permit such Registration Statement in accordance with Section 2(c) until such time as all Registrable Securities pursuant to this paragraph, the Company shall file one or more New Registration Statements in accordance with Section 2(c) until such time as all Registrable Securities have been included in Registration Statements that have been declared effective and the prospectus contained therein is available for use by the Investor. Notwithstanding any provision herein or in the Purchase Agreement to the contrary, the Company's obligations to register Registrable Securities (and any related conditions to the Investor's obligations) shall be qualified as necessary to comport wit

3. RELATED OBLIGATIONS.

With respect to the Registration Statement and whenever any Registrable Securities are to be registered pursuant to Section 2 including on any New Registration Statement, the Company shall use its reasonable best efforts to effect the registration of the Registrable Securities in accordance with the intended method of disposition thereof and, pursuant thereto, the Company shall have the following obligations:

- a. The Company shall prepare and file with the SEC such amendments (including post-effective amendments) and supplements to any registration statement and the prospectus used in connection with such registration statement, which prospectus is to be filed pursuant to Rule 424 promulgated under the Securities Act, as may be necessary to keep the Registration Statement or any New Registration Statement effective at all times during the Registration Period, and, during such period, comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities of the Company covered by the Registration Statement or any New Registration Statement until such time as all of such Registrable Securities shall have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof as set forth in such registration statement.
- b. The Company shall permit the Investor to review and comment upon the Registration Statement or any New Registration Statement and all amendments and supplements thereto at least two (2) Business Days prior to their filing with the SEC, and not file any document in a form to which Investor reasonably objects. The Investor shall use its reasonable best efforts to comment upon the Registration Statement or any New Registration Statement and any amendments or supplements thereto within two (2) Business Days from the date the Investor receives the final version thereof. The Company shall furnish to the Investor, without charge any correspondence from the SEC or the staff of the SEC to the Company or its representatives relating to the Registration Statement or any New Registration Statement.

- c. Upon request of the Investor, the Company shall furnish to the Investor, (i) promptly after the same is prepared and filed with the SEC, at least one copy of such registration statement and any amendment(s) thereto, including financial statements and schedules, all documents incorporated therein by reference and all exhibits, (ii) upon the effectiveness of any registration statement, a copy of the prospectus included in such registration statement and all amendments and supplements thereto (or such other number of copies as the Investor may reasonably request) and (iii) such other documents, including copies of any preliminary or final prospectus, as the Investor may reasonably request from time to time in order to facilitate the disposition of the Registrable Securities owned by the Investor. For the avoidance of doubt, any filing available to the Investor via the SEC's live EDGAR system shall be deemed "furnished to the Investor" hereunder.
- d. The Company shall use reasonable best efforts to (i) register and qualify the Registrable Securities covered by a registration statement under such other securities or "blue sky" laws of such jurisdictions in the United States as the Investor reasonably requests, (ii) prepare and file in those jurisdictions, such amendments (including post-effective amendments) and supplements to such registrations and qualifications as may be necessary to maintain the effectiveness thereof during the Registration Period, (iii) take such other actions as may be necessary to maintain such registrations and qualifications in effect at all times during the Registration Period, and (iv) take all other actions reasonably necessary or advisable to qualify the Registrable Securities for sale in such jurisdictions; provided, however, that the Company shall not be required in connection therewith or as a condition thereto to (x) qualify to do business in any jurisdiction where it would not otherwise be required to qualify but for this Section 3(d), (y) subject itself to general taxation in any such jurisdiction, or (z) file a general consent to service of process in any such jurisdiction. The Company shall promptly notify the Investor who holds Registrable Securities of the receipt by the Company of any notification with respect to the suspension of the registration or qualification of any of the Registrable Securities for sale under the securities or "blue sky" laws of any jurisdiction in the United States or its receipt of actual notice of the initiation or threatening of any proceeding for such purpose.
- e. As promptly as practicable after becoming aware of such event or facts, the Company shall notify the Investor in writing of the happening of any event or existence of such facts as a result of which the prospectus included in any registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, and promptly prepare a supplement or amendment to such registration statement and/or take any other necessary steps (which, if in accordance with applicable SEC rules and regulations, may consist of a document to be filed by the Company with the SEC pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act and to be incorporated by reference in the prospectus) to correct such untrue statement or or omission, and deliver a copy of such supplement or amendment to the Investor (or such other number of copies as the Investor may reasonably request). The Company shall also promptly notify the Investor in writing (i) when a prospectus or any prospectus supplement or post-effective amendment has been filed, and when a registration statement or any post-effective amendment has become effective (notification of such effectiveness shall be delivered to the Investor by email on the same day of such effectiveness and by overnight mail), (ii) of any request by the SEC for amendment to a registration statement or related prospectus or related information, and (iii) of the Company's reasonable determination that a post-effective amendment to a registration statement would be appropriate.

- f. The Company shall use its reasonable best efforts to prevent the issuance of any stop order or other suspension of effectiveness of any registration statement, or the suspension of the qualification of any Registrable Securities for sale in any jurisdiction and, if such an order or suspension is issued, to obtain the withdrawal of such order or suspension at the earliest possible moment and to notify the Investor of the issuance of such order and the resolution thereof or its receipt of actual notice of the initiation or threat of any proceeding for such purpose.
- g. The Company shall (i) cause all the Registrable Securities to be listed on each securities exchange on which securities of the same class or series issued by the Company are then listed, if any, if the listing of such Registrable Securities is then permitted under the rules of such exchange, or (ii) secure designation and quotation of all the Registrable Securities on the Principal Market (as defined in the Purchase Agreement). The Company shall pay all fees and expenses in connection with satisfying its obligation under this Section.
- h. The Company shall cooperate with the Investor to facilitate the timely preparation and delivery of the Registrable Securities (not bearing any restrictive legend) either by DWAC, DRS, or in certificated form if DWAC or DRS is unavailable, to be offered pursuant to any registration statement and enable such Registrable Securities to be in such denominations or amounts as the Investor may reasonably request and registered in such names as the Investor may request.
 - i. The Company shall at all times provide a transfer agent and registrar with respect to its Common Stock.
- j. If reasonably requested by the Investor, the Company shall (i) immediately incorporate in a prospectus supplement or post-effective amendment such information as the Investor believes should be included therein relating to the sale and distribution of Registrable Securities, including, without limitation, information with respect to the number of Registrable Securities being sold, the purchase price being paid therefor and any other terms of the offering of the Registrable Securities; (ii) make all required filings of such prospectus supplement or post-effective amendment as soon as practicable upon notification of the matters to be incorporated in such prospectus supplement or post-effective amendment; and (iii) supplement or make amendments to any registration statement.
- k. The Company shall use its reasonable best efforts to cause the Registrable Securities covered by any registration statement to be registered with or approved by such other governmental agencies or authorities as may be necessary to consummate the disposition of such Registrable Securities.
- l. Within one (1) Business Day after any registration statement which includes the Registrable Securities is ordered effective by the SEC, the Company shall deliver, and shall cause legal counsel for the Company to deliver, to the transfer agent for such Registrable Securities (with copies to the Investor) confirmation that such registration statement has been declared effective by the SEC in the form attached hereto as Exhibit A. Thereafter, if requested by the Investor at any time, the Company shall require its counsel to deliver to the Investor a written confirmation whether or not the effectiveness of such registration statement has lapsed at any time for any reason (including, without limitation, the issuance of a stop order) and whether or not the registration statement is current and available to the Investor for sale of all of the Registrable Securities.
- m. The Company shall take all other reasonable actions necessary to expedite and facilitate disposition by the Investor of Registrable Securities pursuant to any registration statement.

4. OBLIGATIONS OF THE INVESTOR

- a. The Company shall notify the Investor in writing of the information the Company reasonably requires from the Investor in connection with any registration statement hereunder. The Investor shall furnish to the Company such information regarding itself, the Registrable Securities held by it and the intended method of disposition of the Registrable Securities held by it as shall be reasonably required to effect the registration of such Registrable Securities and shall execute such documents in connection with such registration as the Company may reasonably request.
- b. The Investor agrees to cooperate with the Company as reasonably requested by the Company in connection with the preparation and filing of any registration statement hereunder.
- c. The Investor agrees that, upon receipt of any notice from the Company of the happening of any event or existence of facts of the kind described in Section 3(f) or the first sentence of 3(e), the Investor will immediately discontinue disposition of Registrable Securities pursuant to any registration statement(s) covering such Registrable Securities until the Investor's receipt of the copies of the supplemented or amended prospectus contemplated by Section 3(f) or the first sentence of 3(e). Notwithstanding anything to the contrary, the Company shall cause its transfer agent to promptly deliver shares of Common Stock without any restrictive legend in accordance with the terms of the Purchase Agreement in connection with any sale of Registrable Securities with respect to which an Investor has entered into a contract for sale prior to the Investor's receipt of a notice from the Company of the happening of any event of the kind described in Section 3(f) or the first sentence of Section 3(e) and for which the Investor has not yet settled.

5. EXPENSES OF REGISTRATION.

All reasonable expenses, other than sales or brokerage commissions, incurred in connection with registrations, filings or qualifications pursuant to Sections 2 and 3, including, without limitation, all registration, listing and qualifications fees, printers and accounting fees, and fees and disbursements of counsel for the Company, shall be paid by the Company.

6. INDEMNIFICATION.

To the fullest extent permitted by law, the Company will, and hereby does, indemnify, hold harmless and defend the Investor, each Person, if any, who controls the Investor, the members, the directors, officers, partners, employees, agents, representatives of the Investor and each Person, if any, who controls the Investor within the meaning of the Securities Act or the Securities Exchange Act of 1934, as amended (the "Exchange Act") (each, an "Indemnified Person"), against any losses, claims, damages, liabilities, judgments, fines, penalties, charges, costs, attorneys' fees, amounts paid in settlement or expenses, joint or several, (collectively, "Claims") incurred in investigating, preparing or defending any action, claim, suit, inquiry, proceeding, investigation or appeal taken from the foregoing by or before any court or governmental, administrative or other regulatory agency, body or the SEC, whether pending or threatened, whether or not an indemnified party is or may be a party thereto ("Indemnified Damages"), to which any of them may become subject insofar as such Claims (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon: (i) any untrue statement or alleged untrue statement of a material fact in the Registration Statement, any New Registration Statement or any post-effective amendment thereto or in any filing made in connection with the qualification of the offering under the securities or other "blue sky" laws of any jurisdiction in which Registrable Securities are offered ("Blue Sky Filing"), or the omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, (ii) any untrue statement or alleged untrue statement of a material fact contained in the final prospectus (as amended or supplemented, if the Company files any amendment thereof or supplement thereto with the SEC) or the omission or alleged omission to state therein any material fact necessary to make the statements made therein, in light of the circumstances under which the statements therein were made, not misleading, (iii) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any other law, including, without limitation, any state securities law, or any rule or regulation thereunder relating to the offer or sale of the Registrable Securities pursuant to the Registration Statement or any New Registration Statement or (iv) any material violation by the Company of this Agreement (the matters in the foregoing clauses (i) through (iv) being, collectively, "Violations"). The Company shall reimburse each Indemnified Person promptly as such expenses are incurred and are due and payable, for any reasonable legal fees or other reasonable expenses incurred by them in connection with investigating or defending any such Claim. Notwithstanding anything to the contrary contained herein, the indemnification agreement contained in this Section 6(a): (i) shall not apply to a Claim by an Indemnified Person arising out of or based upon a Violation which occurs in reliance upon and in conformity with information about the Investor furnished in writing to the Company by such Indemnified Person expressly for use in connection with the preparation of the Registration Statement, any New Registration Statement or any such amendment thereof or supplement thereto, if such prospectus was timely made available by the Company pursuant to Section 3(c) or Section 3(e); (ii) with respect to any superseded prospectus, shall not inure to the benefit of any such person from whom the person asserting any such Claim purchased the Registrable Securities that are the subject thereof (or to the benefit of any person controlling such person) if the untrue statement or omission of material fact contained in the superseded prospectus was corrected in the revised prospectus, as then amended or supplemented, if such revised prospectus was timely made available by the Company pursuant to Section 3(c) or Section 3(e), and the Indemnified Person was promptly advised in writing not to use the incorrect prospectus prior to the use giving rise to a violation and such Indemnified Person, notwithstanding such advice, used it; (iii) shall not be available to the extent such Claim is based on a failure of the Investor to deliver or to cause to be delivered the prospectus made available by the Company, if such prospectus was timely made available by the Company pursuant to Section 3(c) or Section 3(e); and (iv) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Company, which consent shall not be unreasonably withheld. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Indemnified Person and shall survive the transfer of the Registrable Securities by the Investor pursuant to Section 9.

- Promptly after receipt by an Indemnified Person or Indemnified Party under this Section 6 of notice of the commencement of any action or proceeding (including any governmental action or proceeding) involving a Claim, such Indemnified Person or Indemnified Party shall, if a Claim in respect thereof is to be made against any indemnifying party under this Section 6, deliver to the indemnifying party a written notice of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume control of the defense thereof with counsel mutually satisfactory to the indemnifying party and the Indemnified Person or the Indemnified Party, as the case may be; provided, however, that an Indemnified Person or Indemnified Party shall have the right to retain its own counsel with the fees and expenses to be paid by the indemnifying party, if, in the reasonable opinion of counsel retained by the indemnifying party, the representation by such counsel of the Indemnified Person or Indemnified Party and the indemnifying party would be inappropriate due to actual or potential differing interests between such Indemnified Person or Indemnified Party and any other party represented by such counsel in such proceeding. The Indemnified Party or Indemnified Person shall cooperate fully with the indemnifying party in connection with any negotiation or defense of any such action or claim by the indemnifying party and shall furnish to the indemnifying party all information reasonably available to the Indemnified Party or Indemnified Person which relates to such action or claim. The indemnifying party shall keep the Indemnified Party or Indemnified Person fully apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. No indemnifying party shall be liable for any settlement of any action, claim or proceeding effected without its written consent, provided, however, that the indemnifying party shall not unreasonably withhold, delay or condition its consent. No indemnifying party shall, without the consent of the Indemnified Party or Indemnified Person, consent to entry of any judgment or enter into any settlement or other compromise which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Party or Indemnified Person of a release from all liability in respect to such claim or litigation. Following indemnification as provided for hereunder, the indemnifying party shall be subrogated to all rights of the Indemnified Party or Indemnified Person with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the Indemnified Person or Indemnified Party under this Section 6, except to the extent that the indemnifying party is prejudiced in its ability to defend such action.
- c. The indemnification required by this Section 6 shall be made by periodic payments of the amount thereof during the course of the investigation or defense, as and when bills are received or Indemnified Damages are incurred.
- d. The indemnity agreements contained herein shall be in addition to (i) any cause of action or similar right of the Indemnified Party or Indemnified Person against the indemnifying party or others, and (ii) any liabilities the indemnifying party may be subject to pursuant to the law.

7. <u>CONTRIBUTION</u>.

To the extent any indemnification by an indemnifying party is prohibited or limited by law, the indemnifying party agrees to make the maximum contribution with respect to any amounts for which it would otherwise be liable under Section 6 to the fullest extent permitted by law; provided, however, that: (i) no seller of Registrable Securities guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any seller of Registrable Securities who was not guilty of fraudulent misrepresentation; and (ii) contribution by any seller of Registrable Securities shall be limited in amount to the net amount of proceeds received by such seller from the sale of such Registrable Securities.

8. REPORTS AND DISCLOSURE UNDER THE SECURITIES ACTS.

With a view to making available to the Investor the benefits of Rule 144 promulgated under the Securities Act or any other similar rule or regulation of the SEC that may at any time permit the Investor to sell securities of the Company to the public without registration ("Rule 144"), the Company agrees, at the Company's sole expense, to:

- a. make and keep public information available, as those terms are understood and defined in Rule 144;
- b. file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act so long as the Company remains subject to such requirements and the filing of such reports and other documents is required for the applicable provisions of Rule 144;

- c. furnish to the Investor so long as the Investor owns Registrable Securities, promptly upon request, (i) a written statement by the Company that it has complied with the reporting and or disclosure provisions of Rule 144, the Securities Act and the Exchange Act, (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company, and (iii) such other information as may be reasonably requested to permit the Investor to sell such securities pursuant to Rule 144 without registration; and
- d. take such additional action as is requested by the Investor to enable the Investor to sell the Registrable Securities pursuant to Rule 144, including, without limitation, delivering all such legal opinions, consents, certificates, resolutions and instructions to the Company's transfer agent as may be requested from time to time by the Investor and otherwise fully cooperate with Investor and Investor's broker to effect such sale of securities pursuant to Rule 144.

The Company agrees that damages may be an inadequate remedy for any breach of the terms and provisions of this Section 8 and that Investor shall, whether or not it is pursuing any remedies at law, be entitled to equitable relief in the form of a preliminary or permanent injunctions, without having to post any bond or other security, upon any breach or threatened breach of any such terms or provisions.

9. <u>ASSIGNMENT OF REGISTRATION RIGHTS</u>.

The Company shall not assign this Agreement or any rights or obligations hereunder without the prior written consent of the Investor.

10. <u>AMENDMENT OF REGISTRATION RIGHTS.</u>

No provision of this Agreement may be amended or waived by the parties from and after the date that is one Business Day immediately preceding the initial filing of the Registration Statement with the SEC. Subject to the immediately preceding sentence, no provision of this Agreement may be (i) amended other than by a written instrument signed by both parties hereto or (ii) waived other than in a written instrument signed by the party against whom enforcement of such waiver is sought. Failure of any party to exercise any right or remedy under this Agreement or otherwise, or delay by a party in exercising such right or remedy, shall not operate as a waiver thereof.

11. MISCELLANEOUS.

- a. A Person is deemed to be a holder of Registrable Securities whenever such Person owns or is deemed to own of record such Registrable Securities. If the Company receives conflicting instructions, notices or elections from two or more Persons with respect to the same Registrable Securities, the Company shall act upon the basis of instructions, notice or election received from the registered owner of such Registrable Securities.
- b. Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by email (provided confirmation of transmission is mechanically or electronically generated and kept on file by the sending party); or (iii) one (1) Business Day after deposit with a nationally recognized overnight delivery service, in each case properly addressed to the party to receive the same. The addresses for such communications shall be:

If to the Company, to:

SUPER LEAGUE ENTERPRISE, INC.

2856 Colorado Avenue Santa Monica, CA 90404 Email: ann.hand@superleague.com

Attention: Ann Hand

If to the Investor:

HUDSON GLOBAL VENTURES, LLC

e-mail: info@hudsonventuresllc.com

or at such other address, email address, and/or to the attention of such other person as the recipient party has specified by written notice given to each other party three (3) Business Days prior to the effectiveness of such change. Written confirmation of receipt (A) given by the recipient of such notice, consent, waiver or other communication, (B) mechanically or electronically generated by the sender's email account containing the time, date, recipient email address, as applicable, and an image of the first page of such transmission or (C) provided by a nationally recognized overnight delivery service, shall be rebuttable evidence of personal service, receipt from a nationally recognized overnight delivery service in accordance with clause (i), (ii) or (iii) above, respectively.

The Company and Investor shall submit all Claims (as defined in Exhibit C of the Purchase Agreement) (the "Claims") arising under this Agreement or any other agreement between the parties and their affiliates or any Claim relating to the relationship of the parties to binding arbitration pursuant to the arbitration provisions set forth in Exhibit C of the Purchase Agreement (the "Arbitration Provisions"). The Company and Investor hereby acknowledge and agree that the Arbitration Provisions are unconditionally binding on the Company and Investor hereto and are severable from all other provisions of this Agreement. By executing this Agreement, Company represents, warrants and covenants that Company has reviewed the Arbitration Provisions carefully, consulted with legal counsel about such provisions (or waived its right to do so), understands that the Arbitration Provisions are intended to allow for the expeditious and efficient resolution of any dispute hereunder, agrees to the terms and limitations set forth in the Arbitration Provisions, and that Company will not take a position contrary to the foregoing representations. Company acknowledges and agrees that Investor may rely upon the foregoing representations and covenants of Company regarding the Arbitration Provisions. This Agreement shall be construed and enforced in accordance with, and all questions concerning the construction, validity, interpretation and performance of this Agreement shall be governed by, the internal laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of Delaware. The Company and Investor consent to and expressly agree that the exclusive venue for arbitration of any Claims arising under this Agreement or any other agreement between the Company and Investor or their respective affiliates (including but not limited to the Transaction Documents (as defined in the Purchase Agreement)) or any Claim relating to the relationship of the Company and Investor or their respective affiliates shall be in the State of Delaware. Without modifying the Company's and Investor's obligations to resolve disputes hereunder pursuant to the Arbitration Provisions, for any litigation arising in connection with any of the Transaction Documents (and notwithstanding the terms (specifically including any governing law and venue terms) of any transfer agent services agreement or other agreement between the Company's transfer agent and the Company, such litigation specifically includes, without limitation any action between or involving Company and the Company's transfer agent or otherwise related to Investor in any way (specifically including, without limitation, any action where Company seeks to obtain an injunction, temporary restraining order, or otherwise prohibit the Company's transfer agent from issuing shares of Common Stock to Investor for any reason)), each party hereto hereby (i) consents to and expressly submits to the exclusive personal jurisdiction of any state or federal court sitting in the State of Delaware, (ii) expressly submits to the exclusive venue of any such court for the purposes hereof, (iii) agrees to not bring any such action (specifically including, without limitation, any action where Company seeks to obtain an injunction, temporary restraining order, or otherwise prohibit the Company's transfer agent from issuing shares of Common Stock to Investor for any reason) outside of any state or federal court sitting in the State of Delaware, and (iv) waives any claim of improper venue and any claim or objection that such courts are an inconvenient forum or any other claim, defense or objection to the bringing of any such proceeding in such jurisdiction or to any claim that such venue of the suit, action or proceeding is improper. Notwithstanding anything in the foregoing to the contrary, nothing herein shall limit, or shall be deemed or construed to limit, the ability of the Investor to realize on any collateral or any other security, or to enforce a judgment or other court ruling in favor of the Investor, including through a legal action in any court of competent jurisdiction. The Company hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any objection to jurisdiction and venue of any action instituted hereunder, any claim that it is not personally subject to the jurisdiction of any such court, and any claim that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper (including but not limited to based upon forum non conveniens). THE COMPANY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTIONS CONTEMPLATED HEREBY . The Company irrevocably waives personal service of process and consents to process being served in any suit, action or proceeding in connection with this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby by mailing a copy thereof via registered or certified mail or overnight delivery (with evidence of delivery) to Company at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any other manner permitted by law. The prevailing party in any action or dispute brought in connection with this Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby shall be entitled to recover from the other party its reasonable attorney's fees and costs. If any provision of this Agreement shall be invalid or unenforceable in any jurisdiction, such invalidity or unenforceability shall not affect the validity or enforceability of the remainder of this Agreement in that jurisdiction or the validity or enforceability of any provision of this Agreement in any other jurisdiction.

d.	This Agreement, the Purchase Agreement, and all other ancillary documentation entered into between the Company and Investor therewith
constitute the entire agree	ement among the parties hereto with respect to the subject matter hereof and thereof. There are no restrictions, promises, warranties or undertakings
other than those set fort	th or referred to herein and therein. This Agreement, the Purchase Agreement, and all other ancillary documentation entered into between the
Company and Investor th	nerewith supersede all prior agreements and understandings among the parties hereto with respect to the subject matter hereof and thereof.

- e. Subject to the requirements of Section 9, this Agreement shall inure to the benefit of and be binding upon the successors and permitted assigns of each of the parties hereto.
 - f. The headings in this Agreement are for convenience of reference only and shall not limit or otherwise affect the meaning hereof.
- g. This Agreement may be executed in identical counterparts, each of which shall be deemed an original but all of which shall constitute one and the same agreement. This Agreement, once executed by a party, may be delivered to the other party hereto by e-mail in a ".pdf" format data file of a copy of this Agreement bearing the signature of the party so delivering this Agreement.
- h. Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as the other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- i. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent and no rules of strict construction will be applied against any party.
- j. This Agreement is intended for the benefit of the parties hereto and their respective successors and permitted assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person.

* * * * * *

IN WITNESS WHEREOF, the parties have caused this Agreement to be duly executed as of day and year first above written.

THE COMPANY:

SUPER LEAGUE ENTERPRISE, INC.

By: /s/ Ann Hand

Name: ANN HAND Title: CHIEF EXECUTIVE OFFICER

INVESTOR:

HUDSON GLOBAL VENTURES, LLC

By: <u>/s/ Seth Ahdoot</u> Name: SETH AHDOOT Title: MEMBER

[Signature Page to registration rights agreement]

EXHIBIT A

TO REGISTRATION RIGHTS AGREEMENT

FORM OF NOTICE OF EFFECTIVENESS OF REGISTRATION STATEMENT

, 2025	
	
Re: Effectiveness of Registration Statement	
Ladies and Gentlemen:	
We are counsel to SUPER LEAGUE ENTERPRISE, INC., a Delaware corporate certain Purchase Agreement, dated as of February 14, 2025 (the "Purchase Agreement"), exclude a Nevada limited liability company (the "Investor") pursuant to which the Company value per share (the "Common Stock"), in an amount up to Three Million One Hundred Telefined in the Purchase Agreement) in accordance with the terms of the Purchase Agreementhe Company has registered with the U.S. Securities & Exchange Commission the following	ntered into by and between the Company and HUDSON GLOBAL VENTURES y has agreed to issue to the Investor common stock of the Company, \$0.001 par Thousand Dollars (\$3,100,000.00) of https://documents.org/least-actions Commitment Shares (each as nt. In connection with the transactions contemplated by the Purchase Agreement.
(1) Put Shares to be issued to the Investor by the Company in accord	dance with the Purchase Agreement; and
(2) Commitment Shares.	
Pursuant to the Purchase Agreement, the Company also has entered into a Registration Rights Agreement") pursuant to which the Company agreed, among other the of 1933, as amended (the "Securities Act"). In connection with the Company's obligation of the Company filed a Registration Statement (File No. 333) (the SEC") relating to the resale of the Put Shares and Commitment Shares.	ings, to register the Put Shares and Commitment Shares under the Securities Actons under the Purchase Agreement and the Registration Rights Agreement, on
In connection with the foregoing, we advise you that a member of the SEC's sta Registration Statement effective under the Securities Act at [] [A.M./P.M.] on [] of the SEC's staff, that any stop order suspending its effectiveness has been issued or that and the Put Shares and Commitment Shares are available for resale under the Securities Act egend.], 2025 and we have no knowledge, after telephonic inquiry of a member any proceedings for that purpose are pending before, or threatened by, the SEC
	y truly yours, mpany Counsel]
Ву:	

HUDSON GLOBAL VENTURES, LLC

cc: